

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

RECEPTOR MECHANISMS COMMON TO

PLATELETS AND PLATELET PROGENITOR CELLS

A Thesis submitted in candidature for the degree of Doctor of Philosophy to the Faculty of Science of the University of Glasgow

by

LINDA J.M. MacMILLAN

Department of Pharmacology

1986

ProQuest Number: 10991889

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10991889

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

Acknowledgements

I would like to thank Professor Gillespie for allowing me to study in his department. My time spent there was both illuminating and interesting, thank you. A deeply felt thank you is also extended to my supervisor, Dr Euan MacIntyre. His ability to see the light at the end of the tunnel, when all around you was in darkness, in many different situations, was much appreciated. Additionally his renditions of a seemingly endless list of events at past conference meetings was likewise much appreciated. Thanks a lot, Euan. I would also like to thank my peers, Drs Kenny Pollock, Gus Shaw and Simon Guild and my contemporaries, Mark, Archie and Adriano for their help and useful discussion and just for being there, thank you. My deepest gratitude and thanks is to my parents, whose constant love, support and encouragement helped me throughout the course of this study, thank you. I would like to thank those other members of staff, the technical staff and Mr G.H.D. Melville, who in their own way, helped me during the completion of this thesis. Finally I would like to thank Mrs Peedle for the expedient typing of this thesis.

- iii -

•

LIST	OF CONTENTS	Page No.
Ackno	owledgements	ii
List	of Contents	iii
List	of Figures and Tables	x
Abbr	eviations	$\mathbf{x}\mathbf{v}$
Summa	ary	vvi i
1.	Introduction	1
1.1	Blood Platelets	1
	1.1.1 Platelet Morphology	2
	1.1.2 Platelet Energy Metabolism	8
	1.1.3 Platelet Functional Responses	9
	1.1.3.1 Shape change	9
	1.1.3.2 Adhesion	10
	1.1.3.3 Aggregation	11
	1.1.3.4 Icosanoid synthesis	12
	1.1.3.5 Secretion	16
	1.1.4 Physiological Functions of Platelets	17
	1.1.5 Pathological Functions of Platelets	19
1.2	Megakaryocytes	22
	1.2.1 Megakaryocyte formation	24
	1.2.2 Megakaryocyte Morphology	26
	1.2.3 Megakaryocyte Functional Responses	29
	1.2.4 Physiological Functions of Megakaryocytes	33
	1.2.5 Pathological Functions of Megakaryocytes	35
1.3	Molecular Mechanisms of Cellular Reactivity	37
	1.3.1 The Second Messengers	38

			D 17
	DDM		Page No
6•6•6	RPM		98
	2.2.2.1	Methods of cell culture	99
		2.2.2.1.1 Maintenance of aseptic conditions	101
		2.2.2.1.2 Passaging of RPM	101
	2.2.2.2	Preparation of RPM	101
	2.2.2.3	Radioisotope labelling of RPM with [³² P] P _i	102
	2.2.2.4	Radioisotope labelling of RPM with [3H] inositol	102
	2.2.2.5	Measurement of $[Ca^{2+}]_{i}$:- loading with Quin 2.	104
2.2.3	Phospho	lipid Studies	104
	2.2.3.1	Metabolism of inositol phospholipids in cells prelabelled with [32P] P _i	104
	2.2.3.2	Phospholipid extraction	105
		2.2.3.2.1 Extraction of phospholipids from rat platelets	s 105
		2.2.3.2.2 Extraction of phospholipids from RPM	3 106
	2.2.3.3	Separation of phospholipids by thin layer chromatography	106
		2.2.3.3.1 Separation by thin-layer chromatography of PtdA and PtdIns	107
		2.2.3.3.2 Separation by thin-layer chromatography of poly- phosphoinositides	108
	2•2•3•4	Metabolism of inositol phospholipids in cells prelabelled with [⁹ H] inositol	110
	2.2.3.5	Extraction of inositol phosphates	112
	2.2.3.6	Separation by anion-exchange chromatography of inositol phosphates	s 112

- v -

Page No.

2.2.4 M	Measurement of cytosolic free calcium [Ca ²⁺] _i	113
2.2.5 H	Platelet functional tests	117
2.2.6	Quantification of radioactivity	118
2.2.7 0	Calculation of results	118
2.2.8 0	Other assays: Trypan Blue exclusion	119
	: Protein determination	119
2.2.9 # e	Aim of study and rationale behind experimental approach	120
2	2.2.9.1 Radioactive markers and inositol phospholipid metabolism	124
	2.2.9.1.1 Incorporation of $\begin{bmatrix} 3^2 p & p \end{bmatrix}$	125
	2.2.9.1.2 Incorporation of $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol	127
	2.2.9.1.3 Incorporation of [3 _H] glycerol and fatty acid labelling	128
	2.2.9.1.4 Direct measurement of inositol phospholipid metabolism	129
2	2.2.9.2 Measurement of cytosolic free calcium concentration, [Ca ²⁺] _i	130 [.]
	2.2.9.2.1 Aequorin	130
	2.2.9.2.2 Chlortetracycline	131
	2.2.9.2.3 Quin 2	132
	2.2.9.2.4 Fura 2	133
3. <u>Results</u>		135
3.1 <u>Rat Pla</u>	atelets	144
3.1.1 A F	Agonist-induced metabolism of inositol phospholipids	144
3	3.1.1.1 ADP-induced platelet aggregation	144
3	3.1.1.2 Effect of ADP on platelet inositol phospholipid metabolism:- Time course	146

– vi –

Page No.

	3.1.1.3	Effect of ADP on platelet inositol phospholipid metabolism:-	1 1,0
		concentration-response relationship	140
	3.1.1.4	Inhibition of ADP-induced changes in [32P] Phosphatidate	150
		3.1.1.4.1 Effect of PGE ₁	150
	3.1.1.5	Effect of ADP on polyphosphoinositides	152
		3.1.1.5.1 Effect of ADP on rat platelet polyphosphoinositide metabolis time course	sm 153
3.1.2	ADP ₂ indu [Ca ²⁺] _i	1ced changes in cytosolic free calcium	155
	3.1.2.1	ADP-induced platelet aggregation	155
	3.1.2.2	Effect of ADP on rat platelet cytosolic free calcium [Ca ²⁺] _i	157
	3.1.2.3	Inhibition of ADP-induced changes in [Ca ²⁺] _i	160
3.1.3	Characte ADP in 1	erisation of the receptors recognizing rat platelets	163
	3.1.3.1	Effect of ATP, ADP, AMP and Adenosine on rat platelet inositol phospholipid metabolism:- concentration-response relationship	164
	3.1.3.2	Effect of ATP, ADP, AMP and Adenosine on rat platelet cytosolic free calcium [Ca ²⁺] _i	167
3.1.4	Thrombir phosphol	n-induced metabolism of inositol Lipids	175
	3.1.4.1	Thrombin-induced platelet aggregation	175
	3.1.4.2	Effect of Thrombin on platelet inositol phospholipid metabolism:- Time course	175
	3.1.4.3	Effect of Thrombin on platelet inositol phospholipid metabolism:- Concentration-response relationship	- 178
	3.1.4.4	Inhibition of Thrombin-induced changes in $[3^{2}P]$ - Phosphatidate	180

- viii -

Page No.

3.1.4.4.1 Effect of PGE	182
3.1.4.5 Effect of Thrombin on polyphospho- inositide metabolism:- Time-course	182
3.1.5 Thrombin-induced changes in cytosolic free calcium [Ca ²⁺] _i	185
3.1.5.1 Thrombin-induced platelet aggregation	185
3.1.5.2 Effect of Thrombin on rat platelet cytosolic free calcium [Ca ⁻⁺] _i	185
3.1.5.3 Inhibition of Thrombin-induced changes in [Ca ²⁺] _i	187
3.1.5.3.1 Effect of Hirudin	191
$3.1.5.3.2$ Effect of PGE $_1$	191
3.2 Rat promegakaryoblasts, RPM	191
3.2.1 Agonist-induced metabolism of inositol phospholipids	191
3.2.1.1 Functional responsiveness in RPM	195
3.2.1.2 Effect of ADP on RPM inositol phospholipid metabolism: Time course	196
3.2.1.3 Effect of ADP on RPM inositol phospholipid metabolism: Concentration- response relationship	196
3.2.1.4 Inhibition of ADP-induced changes in [⁹² P] Phosphatidate	199
3.2.1.5 Effect of ADP on polyphosphoinositide metabolism: Time-course	199
3.2.1.6 Effect of ADP on inositol phosphate formation in RPM	203
3.2.2 ADP-induced changes in cytosolic free calcium [Ca ²⁺] _i	207
3.2.2.1 Effect of ADP on RPM cytosolic free calcium [Ca ²⁺] _i	207
3.2.2.2 Inhibition of ADP-induced changes in $\left[Ca^{2+}\right]_{i}$	207

•

	3.2.3	Characte ADP in H	erisation of receptors recognizing RPM	212
	·	3.2.3.1	Effect of ATP, ADP, AMP and Adenosine on RPM inositol phospholipid metabolism: Concentration-response relationship	212
		3.2.3.2	Effect of ATP, ADP, AMP and Adenosine on RPM cytosolic free calcium [Ca ²⁺] _i	214
	3.2.4	Thrombin phosphol	n-induced metabolism of inositol Lipids	217
		3.2.4.1	Effect of Thrombin on RPM inositol phospholipid metabolism: Time-course	222
		3.2.4.2	Effect of Thrombin on RPM inositol phospholipid metabolism: Concentration -response relationship	222
		3.2.4.3	Effect of Thrombin on polyphospho- inositide metabolism: Time-course	225
		3.2.4.4	Effect of Thrombin on inositol phos- phate formation in RPM.	227
	3.2.5	Thrombin calcium,	n-induced changes in cytosolic free [Ca ²⁺] _i	230
		3.2.5.1	Effect of Thrombin on RPM cytosolic free calcium, [Ca ² +] _i	230
4. <u>1</u>	Discuss	sion		232
4.1	Pharma	acology		233
	4.1.1	Metaboli	ism of inositol phospholipids	233
	4.1.2	Changes	in cytosolic free calcium [Ca ²⁺]	236
	4.1.3	Specific	city of agonist effects	240
4.2	<u>Classi</u> and RI	ficatior M	n of ADP receptor on rat platelets	241
4.3	<u>Correl</u> lipids	ation wi and ele	th metabolism of inositol phospho- evation of (Ca ²⁺) _i	248
4.4	Alterr	natives t	to stimulation of inositol phospho-	
	lipid	metaboli	ism in rat platelets/RPM activation	256
4•5	Conclu	iding Ren	narks	257
Refe	erences	3		261

.

List of Figures

Page	No.

1.	Diagrammatic representation of an unactivated platelet .	3
2a.	Metabolic pathways for synthesis and degradation $\dot{\mathfrak{o}}\mathfrak{f}$ inositol phospholipids .	62
2b.	Proposed metabolism and interconversion of inositol phospholipids and inositol phosphates .	63
2c.	Structures of the inositol phospholipids \cdot	72
3.	Incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix} P$ into rat platelets phospholipids with time.	97
4.	Light microscopy of the RPM cell line in various stages of maturation.	100
5.	Incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ P into RPM phospholipids with time.	103
6.	Schematic representation of phospholipid separation by two-dimensional thin layer chromatography.	109
7.	Schematic representation of phospholipids separated by one dimensional thin layer chromatography.	111
8a.	Spectral characteristics of rat platelets incubated with Quin 2 AM.	115
8b.	Spectral characteristics of RPM incubated with Quin 2 AM.	116
9•	Aggregation of washed rat platelets in response to ADP.	145
10.	Time-course of ADP-induced changes in $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdA, in rat platelets.	147
11.	Concentration-response relationship for ADP-induced stimulation of $[3^2P]$ PtdA formation in rat platelets.	149
12.	Effect of PGE on ADP-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation in rat platelets.	151
13.	Time-course of ADP-induced changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns $(4,5)P_2$ in rat platelets.	154
14.	Aggregation of washed rat platelets in response to ADP.	156

- x -

- xi -

Page	No	•
------	----	---

15a.	ADP-induced elevation of cytosolic free calcium, [Ca ²⁺] _i in rat platelets.	158
15b.	Concentration-response relationship for ADP- induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in rat platelets.	159
16.	ADP-induced elevation of cytosolic free calcium in the presence and absence of extracellular free calcium in rat platelets.	161
17.	Effect of PGE on ADP-induced elevation of cytosolic free calcium, [Ca ²⁺], in rat platelets.	162
18.	Concentration-response relationship for agonist- induced stimulation of $[3^2P]$ PtdA formation in rat platelets.	165
19.	Concentration-response relationship for ATP and ATP S-induced stimulation of $[3^2P]$ PtdA formation in rat platelets.	168
20.	Concentration-response relationship for agonist- induced changes in cytosolic free calcium, [Ca ²⁺] _i , in rat platelets.	169
21.	ATP-induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in rat platelets.	170
22.	Concentration-response relationship for ATP- and ATPAS-induced elevation of cytosolic free calcium, $\begin{bmatrix} Ca^2 + \\ \end{bmatrix}_i$, in rat platelets.	172
23.	ATP- and ATPOS-induced elevation of cytosolic free calcium, [Ca ²] _i , in rat platelets.	173
24.	AMP- and Adenosing-induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in rat platelets.	174
25.	Aggregation of washed rat platelets in response to Thrombin.	176
26.	Time-course of Thrombin-induced changes in ^{32}P PtdA in rat platelets.	177
27.	Concentration-response relationship for Thrombin- induced stimulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation in rat platelets.	179
28.	Effect of Hirudin on Thrombin-induced $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdA formation in rat platelets.	181

		Page N
29.	Effect of PGE ₁ on Thrombin-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation in Fat platelets.	183
30.	Time-course of Thrombin-induced changes in [³² P] PtdIns(4,5)P ₂ in rat platelets.	184
31.	Aggregation of washed rat platelets in response to Thrombin.	186
32a.	Thrombin-induced elevation of cytosolic free calcium, [Ca ⁻⁺] _i , in rat platelets.	188
32b.	Concentration-response relationship for Thrombin- induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in rat platelets.	189
33.	Thrombin-induced elevation of cytosolic free calcium in the presence and absence of extra- cellular free calcium in rat platelets.	190
34.	Effect of Hirudin on Thrombin induced elevation of cytosolic free calcium, $\begin{bmatrix} Ca^2 + J_i \end{bmatrix}_i$, in rat platelets	• 192
35.	Effect of PGE ₁ on Thrombin-induced elevation of cytosolic free calcium, $[Ca^{2+}]_i$, in rat platelets.	193
36.	Time-course of ADP-induced changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA in RPM.	197
37.	Concentration-response relationship for ADP- induced stimulation of $\lfloor \frac{32}{2} P \rfloor$ PtdA formation in RPM.	198·
38.	Effect of PGE ₁ on ADP-induced $[3^2P]$ PtdA formation in RPM.	200
39.	Time-course of ADP-induced formation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol polyphosphates in RPM.	205
40a.	ADP_induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in RPM.	208
40ъ.	Concentration-response relationship for ADP- induced elevation in cytosolic free calcium, [Ca ⁻⁺] _i , in RPM.	209
41.	ADP-induced elevation of cytosolic free calcium in the presence and absence of extracellular free calcium in RPM.	210
42.	Effect of PGE, on ADP-induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in RPM.	211

No.

Page No.

43.	Concentration-response relationship for agonist-induced stimulation of [] PldA formation in RPM.	213
44.	Concentration-response relationship for ATP- and ATP δ S-induced formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA in RPM.	215
45.	Concentration-response relationship for agonist- induced changes in cytosolic free calcium, Ca ²⁺ J _i , in RPM.	216
46.	ATP_induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in RPM.	218
47.	Concentration-response relationship for ATP- and ATPYS-induced elevation in cytosolic free calcium, [Ca ²] _i , in RPM.	219
48.	ATP- and ATP/S-induced elevation of cytosolic free calcium, $\begin{bmatrix} Ca^2 \\ j_i \end{bmatrix}$, in RPM.	220
49•	AMP- and Adenosine-induced elevation of cytosolic free calcium, $\left[\operatorname{Ca}^{2+}\right]_{i}$, in RPM.	221
50.	Time-course of Thrombin-induced changes in $\begin{bmatrix} 2 & 2 \\ 2 & P \end{bmatrix}$ PtdA in RPM.	223
51	Concentration-response relationship for Thrombin- induced stimulation of $[3^2P]$ PtdA formation in RPM.	224
52	Time-course of Thrombin-induced formation of [3H] inositol polyphosphates in RPM.	228
53	Thrombin-induced elevation of cytosolic free calcium, [Ca ²⁺] _i , in RPM.	231
54a.	Concentration response relationship for agonist- induced stimulation of PtdA formation and elevation of [Ca ²¹].	250
54ъ.	Concentration-response relationship for agonist- induced stimulation of PtdA formation and elevation of [Ca ²⁺] _i .	251

- xiv -

List of Tables		
1.	Icosanoid-induced cyclic AMP formation by RPM.	122
2.	Icosanoid production by RPM.	123
3.	Time-course of ADP-induced changes in the polyphosphoinositides in RPM.	202
4.	Time-course of Thrombin-induced changes in the polyphosphoinositides in RPM.	226
5•	Comparison of potency (EC ₅₀) values for agonist- induced $[3^2P]$ PtdA formation and elevation of $[Ca^{2+}]_i$ in rat platelets.	243
6.	Comparison of potency (EC ₅₀) values for agonist- induced $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation and elevation of $\begin{bmatrix} Ca^2 \end{pmatrix}_i$ in RPM.	244

an the second states of the second states in the second states of the second states of the second states of the

Abbreviations

A list of the more commonly used abbreviations used in this thesis is presented below:

PtdIns	- phosphatidylinositol
PtdIns(4)P	- phosphatidylinositol-4-phosphate
PtdIns(4,5)P ₂	- phosphatidylinositol-4,5-bisphosphate
PtdA	- phosphatidic acid
Ins(1,4,5)P ₃	- inositol-1,4,5-trisphosphate
Ins(1,4)P ₂	- inositol-1,4-bisphosphate
Ins(1)P	- inositol-l-monophosphate
Ins(1,2)-Cyc- (4,5)P ₃	<pre>- inositol-(1,2-cyclic, 4,5)trisphosphate</pre>
OAG	- 1-oleoyl, 2-acety1-SN3-glycerol
PG	- prostaglandin
HETE	- hydroxyeicosatetraenoic acid
HPETE	- hydroperoxyeicosatetraenoic acid
Tx	- thromboxane
Pi	- orthophosphate
HEPES	- 4-(2-hydroxyethyl)-1-piperazine-ethanesulphoni acid
HBSS	- Hanks balanced salt solution
cyclic AMP	- 3'5'cyclic adenosine monophosphate
cyclic GMP	- 3'5' cyclic guanosine monophosphate
RPM	- rat promegakaryoblast
$\left[c_{a}^{2+} \right]_{i}$	- cytosolic free calcium
DMSO	- Dimethyl sulphoxide
Quin 2	- Quin 2 tetracarboxylic acid
Quin 2 AM	- Quin 2 acetoxymethyl ester
АТР	- adenosine 5'-triphosphate
ADP	- adenosine 5'-diphosphate

C

- xv -

.

- xvi -

AMP	-	adenosine 3'-monophosphate
ATPY S	-	adenosine-5'-0-[3-thiotriphosphate]
5HT	-	5-hydroxytryptamine
GTP	-	guanosine 5'-triphosphate
GDP	-	guanosine 5'-diphosphate
EGTA	-	ethyleneglycol bis(β -aminoethyl ether)N,N,N', N' tetraacetic acid
EGF		epidermal growth factor
PDGF	-	platelet derived growth factor
TPA	-	12-0-tetradecanoyl phorbol-13-acetate
РМА	-	phorbol-12-myristate-13-acetate

Summary

Blood platelets are small, anucleate cells which circulate freely in the cardiovascular system. One of their major physiological functions is concerned with haemostasis - the cessation of bleeding. However, platelets are also involved in the pathological counterpart of haemostasis - thrombosis. Platelets are derived from megakaryocytes which are large nucleated cells located in the bone marrow.

In view of the contribution of platelets towards thrombosis (predominantly arterial thrombosis) for many years the search for novel antithrombotic agents has centred largely on drugs that impair platelet function. However, evidence now exists suggesting, that besides acting at the level of the platelet, some potential antithrombotic drugs, such as aspirin, act elsewhere in the body since the duration of action of aspirin long outlasts the time after this drug would be removed from In addition the duration of aspirins inhibitory the body. effect on platelet icosanoid biosynthesis exceeds the platelet lifespan. This further implies that aspirin can influence platelet reactivity by effects other than on the platelet per se, presumably by acting at the platelet progenitor cell, the megakaryocyte.

In order to exert an effect, drugs, <u>per se</u>, must interact with some receptor or influence the component processes triggered by receptor occupancy which lead to cellular activation. If antithrombotic drugs are in fact able to act at the level of the megakaryocyte as well as the blood platelet, then similar receptors or component processes must exist in both these cell types. This thesis attempted to address this possibility by investigating the nature of the transduction processes in both rat platelets and RPM, a cell line resembling normal rat promegakaryoblasts.

Much evidence has accumulated in recent years suggesting that the transduction process operating in a number of different cell types, including human platelets, involves the agonistinduced metabolism of inositol phospholipids - especially PtdIns $(4,5)P_2$ - with the consequent formation of 1,2-diacylglycerol and $Ins(1,4,5)P_3$. Both 1,2-diacylglycerol and $Ins(1,4,5)P_3$ are believed to act as synergistic second messengers - the former by activating protein kinase (resulting in the phosphorylation of a 40K protein, the latter by mobilizing Ca^{2+} from intracellular stores.

The transduction processes or receptor mechanisms existing in both rat platelets and RPM were investigated using Thrombin and a range of purines. Metabolism of inositol phospholipids was assessed principally as $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation or disappearance of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdIns $(4,5)P_2$, in cells prelabelled with $\begin{bmatrix} 3^2P \end{bmatrix} P_i$ or as the accumulation of $\begin{bmatrix} 3H \end{bmatrix}$ inositol phosphates in cells prelabelled with $\begin{bmatrix} 3H \end{bmatrix}$ inositol. Agonist-induced changes in cytosolic free calcium, $\begin{bmatrix} Ca^{2+} \end{bmatrix}_i$, were monitored using Quin 2. Using this approach, the major findings and conclusions reached were:-

(1) In rat platelets, Thrombin >>ADP elicited, in a concentration dependent manner, the metabolism of inositol phospholipids,

as monitored by $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation and elevation in $\begin{bmatrix} Ca^{2+} \end{bmatrix}_i$, as monitored by Quin 2. As the extent of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation and elevation of $\begin{bmatrix} Ca^{2+} \end{bmatrix}_i$ induced by Thrombin exceeded that evoked by ADP, it can be concluded that Thrombin was a more potent and more efficacious agonist than ADP. (2) In RPM, ADP > Thrombin elicited, in a concentration-dependent manner, the metabolism of inositol phospholipids as monitored by $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA and elevation of $\begin{bmatrix} Ca^{2+} \end{bmatrix}_i$, as monitored by Quin 2. As the extent of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation and elevation of $\begin{bmatrix} Ca^{2+} \end{bmatrix}_i$ induced by ADP exceeded that elicited by Thrombin, in this

system ADP is the more efficacious agonist. (3) In rat platelets, but not RPM, both Thrombin and ADP elicited the rapid disappearance of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdIns(4,5)P₂. In

RPM, but not rat platelets, both Thrombin and ADP induced a significant accumulation of inositol phosphates.

(4) While attempting to characterise the nature of the "purinoreceptor" on rat platelets and RPM it was found that the rank order of potency (EC_{50}) of a range of purines at eliciting inositol phospholipid metabolism and elevation in $[Ca^{2+}]_i$, in both rat platelets and RPM was ADP > ATP >> AMP > Adenosine = 0. In rat platelets ATP was a partial agonist when agonistinduced $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation or elevation of $[Ca^{2+}]_i$ were monitored, whereas both ADP and ATP are full agonists in RPM. Consequently the possibility exists that a distinct and separate receptor for purines exists on rat platelets and RPM. These results would indicate that receptors for both Thrombin and ADP (ergo purines) are expressed on both rat platelets and RPM. When occupied these receptors are coupled to the metabolism of inositol phospholipids and elevation of $\left[\operatorname{Ca}^{2+}\right]_i$. The rank order of potency for purines acting on the "purinoreceptor" appears to be ADP > ATP >> AMP > Adenosine = 0. Since ATP is an antagonist at the ADP receptor on human platelets, the possibility exists that the rat platelet and RPM "purinoreceptors" is distinct from the receptor on human platelets.

 $\mathbf{x}\mathbf{x}$

TO MY MOTHER AND FATHER

1. INTRODUCTION

1.1 Blood Platelets

Blood platelets are small, anucleate cells which circulate freely within the cardiovascular system. Many scientists, throughout the ages, were intrigued by blood coagulation and thrombosis and the nature of cells purported to be involved in such processes. However, it was not until 1881 that Bizzozero demonstrated, that the corpuscular element involved in thrombosis was related neither to the red cell nor the white cell but was of independent origin. He designated this cell a platelet. Almost a hundred years later Born (1962) described a turbidometric technique for measuring platelet reactivity in the laboratory which heralded a flood of investigation into platelet biochemistry, physiology and pharmacology.

Platelets from most species circulate in blood as biconvex discs approximately 2-3 μ in diameter. They are derived from large, nucleated cells named megakaryocytes which are located in bone marrow. Their derivation from megakaryocytes was demonstrated by Wright in 1910 and is the topic of the following section (Section 1.2). Platelet count, rate of production and life span vary with animal species. For example, the platelet count in human whole blood is 2-4 x $10^8/ml$, while in rat blood it is approximately 1-2 x $10^9/ml$. The normal turnover rate of human platelets is about 3.5 x $10^7/ml/$ day (Harker and Finch, 1969). Smaller mammals such as mice and rats usually have a faster turnover. Finally the life

- 1 -

span of platelets in man is about 10 days, while in rats it is approximately 4-5 days. Platelets are removed from the circulation either by involvement in haemostatic plug formation or when senescent by sequestration in the liver or reticuloendothelial system in the spleen. The spleen also contains a pool of viable platelets that can be released into the circulation.

1.1.1 Platelet Morphology

Although conventional optical microscopy revealed that platelets underwent a morphological transformation in their conversion from the quiescent to activated state, the advent of electron microscopic techniques allowed the identification of the "machinery" involved in the platelet activation process. Figure 1 illustrates the major structural features of the platelet. The platelet plasma membrane is rich in glycoproteins, which when stained with lanthanum, gives a characteristic fuzzy appearance to the platelet surface coat or "glycocalyx". The glycocalyx is associated with the outer surface of the plasma membrane, the latter having a typical trilaminar lipid structure.

Subjacent to the plasma membrane, in the equatorial plane of the resting cell is a bundle of microtubules which forms a peripheral ring round the platelet. This microtubule band appears to be a flexible cytoskeleton exerting tension outward, producing the typical 'discoid' shape of the platelet (White and Gerrard, 1979). Consistent with this, selective

- 2 -



glycogen particles can also be seen. A microtubule band encircles the cell just beneath sectioned through the sagittal and equatorial planes, granules including dense granules, lpha granules, lysosomes are randomly distributed throughout the cell. Mitochondria and the plasma membrane. The surface-connected canalicular system and the dense tubular system are prominent in the region near the microtubule band. Figure 1:

After Nichols et al. (1981).

depolymerization of microtubules converts the cell to a more spherical form. When platelets are activated, the peripheral microtubule band becomes constricted, and microtubules can also be visualized within pseudopodia (White and Gerrard, 1979) (vide infra). Other elements of the platelet structural and contractile apparatus include actin and myosin (see Crawford, 1985) which are important for pseudopod formation, internal contraction, granule labilization and secretion, and clot retraction (Cohen et al., 1979).

Within the cytoplasm of the platelet various granular organelles exist. The main types include the dense granules! and the ' α -granules'. Dense granules from human platelets contain calcium, pyrophosphate, ADP, ATP, serotonin and possibly antiplasmin (Holmsen, 1975; Joist et al., 1976). About 65% of human platelet ADP and ATP is stored in dense granules. This is known as the 'storage pool' of adenine nucleotides and may be secreted during platelet activation, but does not exchange readily with the remaining 35% of adenine nucleotides which form the cytoplasmic "metabolic pool" (Holmsen and Weiss, 1979). Based on the observation that enzyme systems removing ADP, such as pyruvate kinase/phosphoenolpyruvate can inhibit aggregation induced by certain agents (Haslam, 1964) a role for released ADP in mediating platelet activation induced by other platelet agonists has been proposed. However, while release of ADP clearly can be demonstrated during platelet activation induced by Thrombin or Adrenaline (Mills et al., 1968; Grette, 1962) the observation that ADP is not detected

- 4 -

during the first phase of aggregation induced by the same agents, but is detected during the second phase (Mills <u>et al</u>., 1968; Harde**gt**y <u>et al</u>., 1970) is incompatible with a universal role for ADP in platelet aggregation. Nevertheless, agonistinduced release of ADP probably is important in recruiting additional platelets to an aggregate thus perpetuating platelet aggregation (See Holmsen, 1977). The other main constituents of dense granules (Ca²⁺ and 5HT) probably subserve a similar role in perpetuating platelet activation. Secreted calcium, for instance, may provide a high local concentration immediately extracellularly to facilitate platelet-platelet cohesion (see Gerrard <u>et al</u>., 1981). Platelets do not synthesise 5HT, but actively scavenge it from plasma. Secreted 5HT could possibly recruit the activity of additional platelets during haemostasis by acting on cell surface receptors.

The other main type of storage organelle whose constituents are secreted when platelets are stimulated are known as the α -granules. The contents of these granules are largely protein in nature and it appears that they may be heterogenous with respect to their capabilities to store different proteins (Holmsen and Weiss, 1979). Two of the constituents of α granules, namely platelet factor four (PF4) and to a lesser extent beta-thromboglobulin (β TG) have antiheparin activity. The role of these secretable proteins in hemostasis is largely unknown, but Walsh <u>et al</u>. (1974) have suggested they might promote plasmatic coagulation around platelet hemostatic plugs by neutralising heparin locally. Platelet derived growth factor (PDGF), another constituent of α -granules, is a low

- 5 -

molecular weight basic protein and has been implicated as causative in promoting atherosclerosis (Packham and Mustard, 1986; see Section 1:1:5) (Loberal, 1974)

Other constituents of the α -granules include certain coagulation factors including factor V and fibrinogen. (Pifer et al., 1977; James et al., 1977; Kaplan et al., 1979a). Factor V is believed to be an essential cofactor in the generation of thrombin from prothrombin in the presence of factor Xa (Nesheim et al., 1980) and thus plays an integral role in blood coagulation. Besides generating fibrin strands, which help consolidate haemostatic plug formation (see Section 1:1:4), fibrinogen is thought to play an important role in blood coagulation by promoting platelet-platelet attachment during platelet aggregation (Bennett and Vilaire, 1979; Mustard et al., 1978; Marguerie et al., 1985) an event crucial to the development of haemostatic plug formation. Addition of platelet agonists is thought to induce a single class of fibrinogen receptor on the surface of platelets (Marguerie et al., 1985). Although direct evidence of the nature of the fibrinogen receptor has yet to be demonstrated, there is reason to believe that glycoproteins GP11b and GP11la constitute the binding site for fibrinogen (Marguerie et al., 1985). Thrombospondin (TSP), a large glycoprotein, is also found in α -granules (McLaren, 1983; Hagen, 1975). Much evidence exists suggesting that this compound serves to stabilize fibrinogen binding to the activated platelet surface and reinforces the strength of interplatelet interactions (Leung, 1984; Silverstein et al., 1986).

- 6 -

Other types of granules located in the cytoplasm are lysosomes and glycogen granules. The former contain a number of degradative enzymes including cathepsins and other proteases and hydrolases (Ehrlich and Gordon, 1976). These degradative enzymes may be of importance in platelet phagocytic processes, and perhaps also in platelet interactions with subendothelial surfaces (Ehrlich and Gordon, 1976). The smallest and most numerous granules in platelets are the glycogen granules; their abundance contrasts with the paucity of platelet mitochondria, and this distribution is consistent with the platelets' metabolic energy being derived mainly by glycolysis rather than oxidative phosphorylation. However, platelet mitochondria can help in providing synthesis of ATP which is essential for most platelet functions (Holmsen, 1977a).

Two other important structures are the surface-connected canalicular system (SCCS) and the network of dense tubules. The SCCS is an invagation of the plasma membrane and in transverse sections may give the appearance of vesicles rather than invagations. It appears to serve as a conduit through which secreted granule contents pass to the cell exterior (White, 1974). The dense tubular system, a smooth endoplasmic reticulum, is strongly implicated as a major site of sequestration of calcium within platelets and may be analogous to the sarcoplasmic reticulum of muscle (White, 1972; Dean and Sullivan, 1982; Cutler <u>et al</u>., 1978; Menashi <u>et al</u>., 1982). The dense tubular system is also the major site of platelet prostaglandin and thromboxane synthesis (Gerrard et al., 1976; Hammarstrom and Falareteau, 1977). Since the dense tubular

- 7 -

system probably is the major source of intracellular calcium used during platelet activation, it is conceivable that these icosanoids could be involved in mobilizing this store of calcium. Such a role was ascribed to thromboxane A_2 by Gerrard and coworkers (1978). However, this claim has been disputed (see Menashi et al., 1984).

1.1.2 Platelet energy metabolism

Platelets are anucleate cells that do not perform many energy requiring anabolic processes, but utilize large amounts of energy during execution of their agonist-induced responses (Akkerman, 1979; Holmsen, 1982). These cells contain a well developed glycolytic system for ATP production (Holmsen, 1972) and although glucose is the preferred substrate, other sugars such as manose, fructose and galactose can be utilized, but ten times less effectively than glucose. That ATP was mandatory for platelet reactivity was elegantly demonstrated in studies where platelets were starved of glucose (Akkermann et al., 1983). These experiments revealed that starved platelets have low levels of available ATP and do not respond to stimuli: refeeding of the starved cells markedly elevates the ATP availability with a concerted reappearance of responsiveness. Platelets can not synthesize adenine nucleotides de novo, but they can synthesize them from adenine, adenosine and hypoxanthine. The store of metabolic adenine nucleotides used in platelet activation is that located in the cytoplasm.

- 8 -

1.1.3 Platelet functional responses

In 1962 Born described an optical technique to monitor platelet aggregation under experimental conditions. Since then numerous techniques have been developed to enable study of a variety of platelet functional responses occurring after exposure to stimuli. These responses extend to (i) shape change; (ii) adhesion; (iii) aggregation; (iv) icosanoid synthesis and (v) secretion. Regardless of the initiating stimulus, platelet activation appears to follow a similar, sequential pattern of response similar to that noted above. However, this is not because each depends on the former one but because of the order of their dependence on second messenger concentration/ATP requirement. Such a pattern of response led Holmsen (1974) to propose the hypothesis of "the basic platelet reaction". Simply stated, depending on the initiating stimuli, different amounts of second messenger will be formed which will independently trigger the different platelet functional increasing amounts of ATP will be utilized as responses: platelet reactivity continues.

A brief account of the platelet functional responses is given below. The nature of the second messenger involved in platelet activation is discussed in Sections 1.3.1.3 - 1.3.1.8. 1.1.3.1 <u>Shape change</u>

Most stimuli, including Thrombin, ADP, calcium ionophore A23187, cause a change in shape. This change involves first the formation of very fine (0.1 µm diameter) pseudopodia (consisting mostly of actin) from the rim of the disc, followed

- 9 -

by a general "rounding up" of the platelet so that it becomes a spiny sphere, often with much broader pseudopodia. In human platelets adrenaline fails to initiate shape change (Mills, 1973). Instead small protrusions qualitatively dissimilar to those evoked by other platelet agonists is The second step in shape change, the change from a observed. disc into an irregular sphere, is correlated with the phosphorylation of platelet myosin, specifically the 20K light chain (Daniel et al., 1984). Shape change, with or without secretion, causes the microtubule bundle that lies beneath the rim of the disk to become centralised and surround the platelet granules, which consequently are concentrated toward the center of the platelet (White, 1974). Unlike aggregation, shape change can occur in the absence of Ca^{2+} in the bathing medium (McLean and Veloso, 1967) suggesting that the internal contractile proteins must utilize Ca²⁺ from internal stores, such as the dense tubular system.

1.1.3.2 Adhesion

Under normal conditions the discoid platelets circulate freely within the cardiovascular system and do not readily adhere to any surface they may come in contact with such as other platelets, other blood components (e.g. red blood cells or leucocytes) or vascular endothelial cells. If however a blood vessel wall is damaged exposing the subendothelium, blood platelets will rapidly accumulate on this surface forming a platelet aggregate and initiate the development of a haemostatic plug. The component of the subendothelium which

- 10 -

induces platelets to aggregate has been shown to be fibrillar collagen (Zucker and Borelli, 1962; Hovig, 1963). At high shear rates, von Willebrand factor - a high molecular weight portion of coagulation factor VIII - is necessary for platelet adhesion to subendothelium (Weiss <u>et al.</u>, 1978).

1.1.3.3 Aggregation

The term 'platelet aggregation' describes the phenomenon when platelets cohere to other platelets so that platelet clumps are formed. Unlike shape change, aggregation has a requirement for fluid phase calcium (or magnesium) (Zucker and Grant, 1978) and fibrinogen (Bennett and Vilaire, 1979; Mustard et al., 1978; Marguerie et al., 1985). Certain platelet agonists, such as ADP or adrenaline, require the presence of this protein in the suspension medium, whereas with Thrombin or collagen, it is secreted from the α granules (Kaplan et al., 1979a). It is believed that Ca²⁺ ions and fibrinogen form links between adjacent platelets thereby promoting platelet clumping. The nature of these links was investigated by Gerrard et al. (1981) (see also Marguerie et al., 1985). An additional requirement for aggregation is that the platelets are able to come into contact with each other; experimentally this can be achieved by continuously stirring a sample of platelets before and after agonist addition. Aggregation, like shape change, is usually measured using the turbidometric method discovered by Born (1962). With this technique, aggregation is measured as the increase in the transmission of light (as a result of platelet clumping) passed through a suspension of platelets. Two types of aggregatory response

- 11 -

can be observed: "primary" aggregation, where platelets stick together and then dissociate, or "secondary" aggregation where platelets stick together irreversibly. The secondary wave of platelet aggregation usually is associated with and mediated by secretory products which further stimulate the cell beyond primary aggregation. Such positive feedback mediators may include released ADP, 5HT or metabolites of arachidonic acid.

1.1.3.4 Icosanoid synthesis

Platelets are capable of synthesizing an array of compounds collectively known as icosanoids (vide infra) which are of great importance in platelet reactivity. These compounds, which include thromboxanes e.g. thromboxane A_2 (TXA₂); prostaglandins e.g. PGG₂ and PGH₂ and hydroxy acids e.g. 12-HPETE and 12 HETE, are synthesized from arachidonic acid. The phospholipids of platelets can act as a repository for arachidonic acid, in particular the Sn2-position of phosphoinositides are especially rich in this fatty acid (Marcus <u>et al</u>., 1969). As such these compounds may act as a potential source of arachidonic acid for use in icosanoid synthesis. There is evidence to suggest that in fact the phosphoinositide species with the highest arachidonic acid content are selectively hydrolysed during platelet activation (Mahadevappa and Holub, 1983).

Release of arachidonic acid from phospholipids, in particular phosphoinositides, may occur by two separate routes. Firstly agonist-induced activation of platelets may cause the

- 12 -

formation of 1,2-diacylglycerol, via the action of a specific phospholipase type enzyme (see Section 1.3.1.5). This entity constitutes the fatty acid-containing backbone of the phospholipids. Subsequent and sequential activation of di- and monoglyceride lipases release the arachidonic acid (Mauco <u>et al.</u>, 1984; Bell <u>et al.</u>, 1979; Prescott and Majenis, 1983). The second route by which arachidonic acid may be released from phospholipids involves the action of a phospholipase \mathbf{f}_2 enzyme (Bills <u>et al.</u>, 1977; Broekman <u>et al.</u>, 1980; McKean <u>et al.</u>, 1981; Lapetina, 1982). In passing it is prudent to mention that although phosphoinositides can provide a source of arachidonic acid, it appears that this source represents but a small fraction of released arachidonic acid; the majority being derived from phosphatidylcholine (Bills <u>et al.</u>, 1977a).

Following the release of arachidonic acid it is converted by a lipoxygenase enzyme to 12-HPETE and 12-HETE or by a cyclo-oxygenase enzyme to two labile endoperoxides, PGG_2 and PGH_2 (Hamberg and Samuelsson, 1974). These compounds are not stored in the platelet and are synthesized only after platelet activation has been instigated. The physiological role of the lipoxygenase products is not clear, although a number of physiologic effects of these compounds have been described. Both 12-HPETE and 12-HETE have been shown to inhibit <u>in vitro</u> (human) platelet aggregation (Aharony <u>et al</u>., 1982), and 12-HETE inhibits PGH_2 -induced platelet aggregation (Croset and Lagarde, 1983; Lagarde <u>et al</u>., 1985). However, to date, no single clear-cut function for the 12-lipoxygenase

products has been noted. The products of cyclo-oxygenase the endoperoxides PGG, and PGH,- are converted primarily to thromboxane A₂ by a thromboxane synthetase (Hamberg <u>et al</u>., 1975), or spontaneously to the degradation products 12Lhydroxy-5,8,10-hepatadecatrienoic acid (HHT) and malondialdehyde (MDA), with small amounts of PGD₂, PGE₂ and PGF_{2 α} being formed by specific synthetases. Thromboxane A_2 , a very potent aggregating agent, has a short half life and is degraded nonenzymatically to thromboxane B_2 (Hamberg <u>et al</u>., 1975). However, in plasma TxA₂ is more stable and has a half-life of several minutes (Smith et al., 1976). Addition of TxA₂ to platelets causes shape change, aggregation and secretion (Gerrard and White, 1978). Receptors exist for TxA, on platelets (see Jones et al., 1985 for a review) however Gerrard et al. (1978a) proposed that TxA₂ may act as a calcium ionophore in platelets and hence may not act at specific receptors. When added to platelets both PGG, and PGH, are capable of evoking platelet aggregation (Hamberg et al., 1974). However, controversy exists as to whether PGG₂ and PGH₂ must be converted to TxA₂ prior to activating platelets. Some researchers suggest that conversion is necessary (Hamberg et al., 1975; Gorman et al., 1977) whereas others have stated facts to the contrary (Raz et al., 1977). The available evidence to date suggests that conversion of endoperoxides to TxA₂ is not a prerequisite for platelet activation. Based on structure-activity-relationship studies, and the use of selective antagonists, current evidence suggests that the endoperoxides and TxA₂ share a common type of receptor, namely a TP-receptor (MacIntyre, 1985; Coleman

- 14 -
<u>et al</u>., 1984). In the absence of specific inhibitors, the endoperoxides are rapidly metabolised to TxA_2 so that in the usual situation TxA_2 is the major activating agent formed in platelets.

Other tissues, besides platelets, are capable of synthesizing icosanoids. Of particular importance is the production of icosanoids by endothelial cells (Weksler, 1984). Unlike the situation observed in platelets where TxA2 is the major active agent, endothelial cells preferentially synthesize prostacyclin (PGI2). This compound is one of the most active agents at inhibiting platelet reactivity by virtue of elevating levels of cyclic AMP in platelets (Moncada <u>et al</u>., 1976; Tateson <u>et al</u>., 1977; Moncada and Vane, 1979; Moncada and Whittle, 1985). $\mathrm{PGD}_{\mathrm{p}}$ acts similarly but is less potent and since only trace amounts are produced it is probably less significant (Oelz et al., 1977). The production of icosanoids with activating and inhibitory actions has led to the concept that a balance in production of these compounds may be critical in overall hemostatic function (Bunting et al., 1983; Moncada and Vane, 1979a). Imbalances in the production of icosanoids - in particular TxA₂ and PGI₂ - could potentially lead to bleeding or thrombotic tendencies. In this context, great use has been made of such agents as aspirin and indomethacin. These compounds are known as potent antiplatelet (antithrombotic) agents by virtue of their ability to inhibit the activity of cyclo-oxygenase by acetylating it - the former irreversibly (Roth and Majerus, 1975). As a consequence the production of TxA2- and its consequent effects on platelet reactivity are terminated.

- 15 -

Agonist-induced formation of TxA, and its subsequent release into the suspending medium (or plasma) may act to perpetuate platelet aggregation by activating additional platelets.(Holmsen, 1977). However release of icosanoids, including TxA₂, are not a prerequisite for platelet activation induced by all agonists. Detwiler and Huang (1980) proposed that two types of agonists exist, weak stimulators (ADP and Adrenaline) and strong stimulators (Thrombin) based on their susceptibility to inhibition by agents inhibiting cyclo-oxygenase. Thus agonists such as Thrombin are capable of eliciting full platelet activation in the presence of these compounds, ergo in the absence of endoperoxides and TxA2, whilst other agonists such as ADP and Adrenaline require their presence (Detwiler and Huang, 1980). Nevertheless, by virtue of inhibiting the generation of endoperoxides and TxA2, inhibitors of cyclooxygenase still remain an important group of drugs in antiplatelet (antithromboric) therapy (Packham, 1983; Weksler, 1984). 1.1.3.5 Secretion

Secretion of the constituents of dense granules, α granules and lysosomes represents the last characteristic feature of activated platelets to be discussed. There is extreme selectivity in the secretion of constituents from different granules. Holmsen and Day (1970) demonstrated that, in the secretory response evoked by Thrombin release from dense granules preceded release of α granule constituents. Consequently it was proposed that two phases of degranulation exist, namely release I, which involves release from dense granules and release II, which

- 16 -

It can be demonstrated that weak agonists such as ADP and Adrenaline or low concentrations of Thrombin do not induce release of α granule contents whereas high concentrations of Thrombin can release constituents of both granules (Day and Holmsen, 1971). Release of lysosomal granule contents requires powerful stimuli in that ADP, Adrenaline and TxA, fail to evoke release of lysosomal hydrolases, whereas Thrombin, collagen and A23187 are effective secretagogues. During platelet activation granule centralisation occurs (Section 1.1.3.1). However, when platelets aggregate, granules move to the centre of the aggregate more often than to the centre of individual cells (Gerrard et al., 1979). Actual secretion of granule contents occurs at the SCCS which serves as a conduit through which secreted granule contents pass to the cell exterior (White, 1974). Secretion often accompanies aggregation, however aggregation is not a prerequisite as strong stimuli, such as Thrombin can induce platelet secretion in the absence of aggregate formation (Detwiler and Huang, 1980).

1.1.4 Physiological Functions of Platelets

Blood platelets usually circulate freely within the cardiovascular system as inert cells. However, in response to tissue damage, or wounding, they become highly active and participate in haemostasis. This process, of stemming the flow of blood, is the major physiological role of blood platelets (Gordon and Milner, 1976).

Platelets contribute to haemostasis in two major ways. Firstly they form haemostatic plugs at the site of vascular

- 17 -

injury and secondly they help promote coagulation which consolidates the plug with fibrin strands.

Formation of the haemostatic plug incorporates the previously mentioned platelet functions, namely adhesion, aggregation and secretion. During damage to the blood vessel wall the subendothelium is exposed and platelets rapidly form an adherent layer over the exposed collagen. Such platelets then undergo secretion, releasing ADP, Ca²⁺, 5HT and icosanoids which can then stimulate and recruit additional platelets forming a haemostatic plug at the site of injury. Coincident with the initiation of haemostasis is the initiation of coagulation. Platelets help promote this process in three main ways. Firstly, the procoagulant phospholipid complex known as platelet factor 3 is exposed at the surface of activated platelets, secondly, platelets secrete procoagulant proteins (such as fibrinogen and platelet factor 4) from their a-granules when activated, and finally, they carry coagulation Factor XI on their plasma membrane, and this factor is activated when platelets come into contact with collagen. Factor V, released from platelet α -granules, is an essential cofactor in the generation of Thrombin from prothrombin during the coagulation Besides being a powerful activator of platelets, cascade. Thrombin is also able to catalyse the polymerization of fibrinogen to fibrin thus helping to consolidate the haemostatic plug with fibrin strands. After stemming the flow of blood, the next process to occur is 'repair' of vascular damage. Certain secreted products, such as PDGF, have mitogenic properties

- 18 -

and can cause the proliferation of fibroblasts and arterial smooth muscle cells maintained in culture (Kaplan <u>et al</u>., 1979a; Ross <u>et al</u>., 1974; see also Ross <u>et al</u>., 1984). This property of PGDF, in addition to its ability to stimulate DNA synthesis in a variety of confluent cells in culture (Heldin <u>et al</u>., 1976; Antoniades <u>et al</u>., 1979) has led some researchers to suggest that it may play an integral role in the development of atherosclerosis (Packham and Mustard, 1986; Section 1.1.5).

Other important properties of platelets include maintaining endothelial cell integrity (Johnson, 1971) and clearing particles from the blood using pinocytotic and phagocytotic mechanisms.

1.1.5 Pathological functions of platelets

Besides playing an essential role in haemostasis, platelets also have an integral role in its pathological perversion thrombosis. Other pathological processes in which platelets play a fundamental role are atherosclerosis and inflammation.

Stemming the flow of blood after vascular damage, by forming a haemostatic plug, is a normal physiological function of platelets. However, in certain conditions, haemostatic plugs or thrombi can form inside blood vessels in the absence of damage severe enough to cause blood loss. Thrombi can form in veins, arteries or chambers of the heart. In venous thrombosis, packets of platelets aggregate together and adhere to the vessel wall, particularly in the neighbourhood of valves. Localised coagulation follows and the intercalating strands of fibrin so produced form a dense network - a so-called red

- 19 -

Arterial thrombi consist of a larger platelet mass thrombus. with fewer fibrin strands - a so-called white thrombus. The difference between the two types of thrombi appears to arise partly from the greater blood flow in arteries compared to veins. However, common to both types is the possibility of a part of the thrombus (embolus) breaking off and being swept into the general circulation, later to lodge in the small blood vessels of the lung or brain. Such occurrences represent perhaps one of the more serious medical complications. The causative factor(s) in the development of thrombi are unknown, but certain conditions such as immobility, or pregnancy and certain social habits such as smoking may predispose an individual to developing thrombosis.

Atherosclerosis is a process characterised by focal lesions of the arterial intima that consist of plaques containing varying amounts of fibrous tissue and lipid. The result of the plaque formation is localised thickening of the arterial wall with consequent narrowing of the vascular lumen. Consequences of this narrowing include cardiac, cerebral or peripheral ischemia which in advanced stages, can result in The actiology of atherosclerosis is unknown, however death. dietary factors, smoking, hypertension and diabetes have all been implicated (Packham and Mustard, 1986). The exact role of blood platelets in the genesis of atherosclerosis is They may however participate in one of several ways. unknown. Firstly, mural thrombi may become incorporated into the endothelium and develop intimal thickenings similar to atherosclerotic plaques. However, studies have shown that

- 20 -

these plaques differ from atherosclerotic plaques which occur spontaneously (Craig et al., 1973), hence the contribution, made by platelets, to atherosclerosis, through this mechanism is ill-defined. A second, more likely, contribution made by platelets to the atherosclerotic process lies in their ability to secrete products which may inflict changes on the endothelium. Atherosclerotic plaques often occur around bifurcations in the vasculature (Murphy et al., 1962). Such disturbances in blood flow may result in small aggregates of platelets with consequent release of their granule contents. Amongst the constituents of α -granules are mitogens such as platelet derived growth factor and β -transforming growth factor. These compounds may be of importance in the development of atherosclerosis (Harker et al., 1978; Packham and Mustard, 1986). For a recent review of the actiology of atherosclerosis and the possible role(s) of platelets in the development of the disease, see Packham and Mustard (1986).

Inflammation has been defined as a local reaction to injury of the microcirculation and its contents (Spector and Willoughby, 1968). Acute inflammation is characterised by warmth, redness, pain and swelling. Physiologically such effects are mediated respectively by vasodilation and increased blood flow, activation of sensory nerve endings and increased vascular permeability. As already stated, activated platelets, as a result of injury, are capable of forming icosanoids from released arachidonic acid (Section 1.1.3.4). In addition to the endoperoxides (PGG₂ and PGH₂) prostaglandin E_2 is also formed and this compound increases vascular permeability

- 21 -

(Wallis <u>et al</u>., 1972; Silver et al., 1974). HETE, a substance formed by platelet 12-lipooxygenase, has been demonstrated to be chemotactic for polymorphonuclear leukocytes (Turner <u>et al</u>., 1975) further emphasizing the importance of platelet icosanoid formation in inflammation processes. Chronic inflammation is associated with tissue degradation. Platelets also contain protinases (Ehrlich and Gordon, 1976) and the possibility exists that platelet protinases may participate in such pathological conditions.

1.2 <u>Megakaryocytes</u>

Although the megakaryocyte was known to Bizzozero, it was not identified as the cell of origin of platelets until the studies of Wright at the beginning of the century. Using histological techniques and observing megakaryocytes <u>in vitro</u>, Wright (1910) established the megakaryocyte as the source of blood platelets. Since then, various more elaborate techniques have been used in the study of these cells and in 1969 Behnke, using electron micrographical techniques, confirmed and extended the observations of Wright.

Megakaryocytes are large, nucleated cells that originate in the bone marrow. However, the number of these cells in human bone marrow is very low, about 0.037% of all cells. This value is only 10% of that recorded in bone marrow from the guinea pig or rat megakaryocyte (Levine, 1980). As technology has advanced there has been a corresponding upsurge of interest in these cells. Specific criteria now exist for the identification of megakaryocytes (Levine, 1981). They can

- 22 -

be identified on the basis of (1) size. Typically these large cells range from between 10 µm to 50 µm, (2) nuclear size or ploidy level. Unlike most other cells in the body, megakaryocytes are capable not only of cytoplasmic division but of the company of the second s with a resultant increase in nuclear material. Hence ploidy levels can range from 2N - 64N, where N is the diploid number; (3) staining for acetylcholinesterase. Staining for this enzyme was once thought to be a specific marker for megakaryocytes, especially of the rodent species (Jackson, 1973). However the discovery that other, non-megakaryocytic cells could also stain positive for this enzyme lends the fidelity of this as a marker for rodent megakaryocytes doubtful (Lepore et al., 1984). Megakaryocytes have a distinctive morphology and this can also be used in their identification (Levine, 1981; Williams and Levine, 1982). The paucity of this cell in bone marrow in no way reflects its biological importance. As the precursor of platelets, its relevance to haemostasis is fundamental. Besides elaborating blood platelets, studies have also demonstrated their capacity to synthesize coagulation proteins such as fibrinogen, factor VIII antigen, platelet factor 4 and factor V. They can also take up serotonin (5HT) from the blood and store it in granules and impart the majority of functional properties to the platelet (vide infra; see also Schick and Schick (1980) for review on megakaryocyte biochemistry). Most studies on megakaryocytes have been performed on animal tissue primarily due to the greater numbers

- 23 -

of cells obtainable from such sources (see Levine (1986) for a comparison of cell harvest amongst different species). However, there is no good evidence available from the present literature, to suggest any major difference may exist when In order to study these cells, isolation using human tissue. from the bone marrow is required. Several techniques now exist for isolating and enriching megakaryocytes obtainable from both human and non-human sources. (Schick and Schick, 1986). Megakaryocytes also appear in blood (vide infra) and Morgan and Brodsky (1985) recently demonstrated the ability of a megakaryoblast-like cell, obtained from blood, to be maintained in culture, although it did not differentiate into a more mature megakaryocyte. Given the scarcity of megakaryocytes in human bone marrow, the blood as a convenient and ready source of megakaryocytes seems unlikely.

1.2.1 Megakaryocyte Formation

Megakaryocytes are thought to be derived from pluripotential stem cells (Ebbe, 1981). These stem cells do not seem to be in a proliferative mode, nor does it appear that homeostatic mechanisms regulating megakaryocytopoiesis act at this level (Ebbe, 1981).

Thus, a more differentiatied cell must therefore respond to changes in platelet demand. The earliest detectable progeny of these pluripotential stem cells capable of elaborating morphologically identifiable megakaryocytes appears to be the colony-forming unit megakaryocyte (CFU-M) (Nakeff and Daniels-McQueen, 1976). This cell is neither cytochemically

- 24 -

nor morphologically recognizable, however it is capable of forming colonies of morphologically recognizable megakaryocytes. When appropriately stimulated, CFU-M begin to proliferate (Hoffman et al., 1975). They give rise to progeny which have less proliferative qualities but more characteristics of the mature cell they will eventually become, commonly known as transitional cells. Zajicek (1954) demonstrated these cells to stain positive for acetylcholinesterase in murine tissue and proposed that these cells were precursors for megakaryocytes. These observations were confirmed by Jackson (1973) (but see Lepore et al. (1984)). Definitive proof that these cells were in fact immature megakaryocytes was provided by Long and Williams (1981). They found that the small acetylcholinesterasepositive cells did not give rise to colonies of megakaryocytes, but rather matured into single megakaryocytes. The human counterpart of these immature megakaryocytes have proved more elusive to identify, however their existence is not disputed. These immature megakaryocytes eventually become recognizable at the level of the light microscope by increasing their ploidy level and state of cytoplasmic maturation (Long et al., 1982). Such cells then routinely pass through various stages of maturation distinguishable by difference in nuclear/ cytoplasmic ratio, nuclear configuration and size (Williams and Levine, 1982) (vide infra).

Odell and Jackson (1968) using rat tissue, demonstrated that only the most immature megakaryocytes can synthesize DNA and that development of polyploidy occurs only in young megakaryocytes. These young megakaryocytes were shown to

- 25 -

exhibit the full range in ploidy levels namely 4N; 8N; 16N; 64N. Odell and Jackson postulated that the development of polyploidy occurs before cytoplasmic maturation begins and that when cytoplasmic maturation commences no more DNA is synthesized. In a later study, Levine <u>et al</u>. (1982) demonstrated that maturation and polyploidy appeared to be linked in that low ploidy (4N - 8N) megakaryocytes were immature whereas platelet-shedding megakaryocytes were (gN - 32N).

Regulation of megakaryocytopoiesis is believed to be under the control of two humoral stimuli, namely, Thrombopoientin and megakaryocytic colony stimulating factor. (Williams and Levine, 1982; Gewirtz, 1986). Not much is known about the exact mechanisms of action of these substances, however, with respect to regulating platelet formation, it is clear that a deficiency of circulating platelets is associated with stimulation of megakaryocytopoiesis (Craddock <u>et al.</u>, 1955) and an excess with its suppression (Cronkite, 1957).

1.2.2 Megakaryocyte Morphology

Megakaryocytes at all stages of their development have a distinct morphology (Williams and Levine, 1982; Levine, 1981) and depending upon the author and techniques used their development has been postulated to cover 3 (Ebbe and Stohlman, 1965) to 4 (Levine <u>et al</u>., 1982; Williams and Levine, 1982) separate stages (see also Breton-Gorius and Vainchenker, 1986).

According to Ebbe and Stohlman (1965) megakaryocytopoiesis in the rat proceeds through three stages, Stage I or megakaryoblast; Stage II or basophilic and Stage III or granular megakaryocyte.

- 26 -

Stage I represents the earliest identifiable cell by light microscope as belonging to the megakaryocyte series. Salient features at this stage are a compact nucleus occupying most of the cytoplasm; the cytoplasm has a clear, basophilic appearance; mitochondria are also present at this stage. In their study Rabellino et al. (1981) demonstrated the existence of specific platelet glycoproteins, platelet factor 4 and factor VIII-antigen occurring in what they termed "early megakaryocytes". The appearance of specific organelles, including granules have also been detected early in megakaryocyte development (Paulus, 1970). Various studies have confirmed both the synthesis and localisation of various α -granule constituents, including platelet factor 4; factor VIII antigen; factor V and fibrinogen in megakaryocytes (Rabellino et al., 1979; Ryo et al., 1983; Nachman et al., 1977; Chiu et al., 1985). The formation of the dense membrane system (DMS) was shown by MacPherson (1972) to be an event, occurring in earliest recognizable megakaryocytes. This membraneous system is thought to be derived from the megakaryocyte plasma membrane and to form, at least in part, the surface coat of future platelets (Behnke, 1968). However, elegant freeze-fracture studies performed by Zucker-Franklin and Petursson (1984) have shown differences in the plasma membrane constituents of platelets and megakaryocytes, suggesting that the surface membranes of these two cells differ in origin and structure. The cells in this stage are present in a range of ploidy classes (Odell and Jackson, 1968) suggesting that a number of successive endomitosis must occur during the transit of cells through this stage.

- 27 -

Stage II cells have a characteristically different These cells have a lower nuclear-cytoplasmic ratio; morphology. the nucleus is usually lobulated and the cytoplasm has a foamy basophilic appearance. The mitochondria are more numerous than at the previous stage and scattered throughout the cytoplasm. On electron microscopy they exhibit all the cell specific organelles present in stage I, but in greater numbers. The smooth endoplasmic-reticulum in these cells has been shown by histochemical techniques to be identical with the dense tubular systems of the circulating platelet (Breton-Gorious and Reyes, 1976). Before this stage is completed, endomitosis has ceased and nuclear segmentation increases (Ebbe and Stohlman, 1965); it is likely that the development of cytoplasmic maturation is itself a potent inhibitor of further synthesis of DNA, however the exact nature of this inhibitor is unknown. At this stage of maturation, the full ploidy distribution has been attained and cells belonging to each class of ploidy mature in parallel (Odel and Jackson, 1968) (Contrast this with Levine et al. (1982) where they state that a relationship exists between ploidy level and state of maturation). It is probable therefore that platelet production can take place in megakaryocytes at several ploidy stages.

Stage III cells possess nuclei which have become condensed and centralised in the cell. The cells however continue to grow in size. The penultimate phase of megakaryocyte maturation (Stage IV by other authors) includes cells which are capable of releasing platelets (<u>vide infra</u>). In rats, the distribution between the various stages is 18% in Stage I; 25% in Stage II and 56% in Stage III; the average total maturation time of

- 28 -

rat megakaryocytes was estimated to be approximately 43-75 hours (Ebbe and Stohlman, 1965).

It is clear that throughout megakaryocyte maturation, most of the platelet specific proteins and organelles are synthesized within the megakaryocyte and it is likely that subsequent to platelet release, granule contents are packaged within the megakaryocyte.

1.2.3 Megakaryocyte functional responses

Megakaryocytes are thought to be designed specifically for elaborating platelets and hence are indirectly involved in Nevertheless, under certain conditions, megahaemostasis. karyocytes have been shown to exert platelet-like behaviour often to a striking degree. These responses include cellular shape change or cell spreading (Leven et al., 1983; Leven and Nachmas, 1981; Fedorko, 1977; Leven and Nachmas, 1982); metabolism of arachidonic acid (Walenga et al., 1984; Miller et al., 1982; Levine et al., 1985); uptake, storage and release of serotonin (5HT) (Fedorko, 1977; Schick and Weinstein, 1981; Tranzer et al., 1972) and release of ATP (Miller, 1983). These responses in megakaryocytes can be elicited by a range of platelet agonists including Thrombin; ADP; Adrenaline and calcium ionophore A23187 (Leven and Nachmias, 1982; Walenga et al., 1984; Fedorko, 1977; Miller, 1983; Leven et al., 1983; Leven and Nachmias, 1981). Because of the difficulty in obtaining human tissue, most of the above studies were performed using animal tissue. However there is no reason to expect any major differences in response were human tissue to be The significance of these responses to normal megakaryocyte used.

- 29 -

function is unknown but serves to demonstrate the similarities between them and their progeny.

In response to ADP $(1-100 \mu M)$ and Thrombin $(1 \mu/m1)$ Levine and Nachmias (1982) demonstrated the spreading of cultured guinea pig megakaryocytes. Such spreading was blocked by dibutyryl cyclic AMP or isobutylmethylxanthine but not by colchicine indicating the action of actomyosin system rather than the microtubular system - similar to the situation observed in platelet activation. Mechanistic studies suggested that intracellular alkalinization accompanied by an increase in intracellular free calcium was required for spreading induced by ADP or Thrombin (Nachmias, 1983). Whether this holds true for other agonists or represents only one mechanism for megakaryocyte spreading is unknown. Fedorko (1977), also using guinea pig megakaryocytes, could also observe marked changes in megakaryocyte shape with ADP (10 μ M - 1 mM); $(1 - 12 \mu M)$. The requirement for a high agonist concentration to observe any spreading in the latter study may reflect species variations of megakaryocytes and hence an ability to respond to stimuli; additionally during preparation necessary cofactors may be washed from the megakaryocyte preparation. One major difference between shape change in megakaryocytes and their projenitors is, that in the former, shape change occurs over a period of minutes rather than seconds as occurs in platelets. Additionally, instead of producing filopodia, megakaryocytes appear to form ruffles at the leading edge (Nachmias, 1983).

- 30 -

Metabolism of arachidonic acid has been demonstrated in megakaryocytes from a variety of species. In rat megakaryocytes Demers et al. (1980) demonstrated the synthesis of significant amounts of prostaglandin E. Also in rat megakaryocytes, Worthington and Nakeff (1981) demonstrated the synthesis of thromboxane B_2^{\bullet} . Using guinea pig megakaryocytes Miller <u>et al</u>. (1982) demonstrated the synthesis of thromboxane B_2 , 12-hydroxyheptadecatrienac acid (HHT) and 12-hydroxy elcosatetraenoic acid (12-HETE). These results suggested that megakaryocytes possess the full complement of cyclo-oxygenase and lipoxygenase activities found in circulating platelets. The metabolism of arachidonic acid in platelets can be affected by inhibitors of cyclo-oxygenase such as aspirin. Studies have demonstrated that in addition to inhibiting platelet cyclo-oxygenase, aspirin may also inhibit cyclo-oxygenase in megakaryocytes. Using rat megakaryocytes Demers et al. (1980) demonstrated that aspirin could abolish the ability of the megakaryocytes to synthesize prostaglandin E. Worthington and Nakeff (1982) also demonstrated the ability of aspirin to inhibit rat megakaryocyte thromboxane synthesis. The latter authors demonstrated a lag period occurring between aspirin addition and full recovery of cyclo-oxygenase activity in platelets. This is thought to represent inhibition of cyclooxygenase in megakaryocytes - normal cyclooxygenase activity returning after synthesis of new cyclooxygenase in the megakaryocyte. Some studies have reported a difference in sensitivity of cyclooxygenase in platelets and megakaryocytes in response to inhibition by aspirin. Worthington and Nakeff

- 31 -

(1982) reported that platelets were approximately 25 times more sensitive to the effects of aspirin on cyclooxygenase as were megakaryocytes. Contrary findings were reported by Walenga <u>et al</u>. (1984). The difference may be explained by species differences in that rat tissue was used in the former study and guinea pig tissue in the latter.

Like platelets, megakaryocytes have been demonstrated to take up, store and then release 5HT (Fedorko, 1977). Uptake was temperature dependent and thus in part energy dependent; uptake was also inhibited by reserpine and imipramine - similar to findings using platelets. Release of 5HT was induced by a range of known platelet agonists such as ADP, Thrombin and Adrenaline. Schick and Weinstein (1981) demonstrated by autoradiography, at the level of the light microscope, that megakaryocytes appeared to be the only cells in marrow taking However, regardless of stage of maturation, similar up 5HT. amounts of labelled 5HT had been accumulated into the cells. They proposed that 5HT accumulation may be a marker for the immature megakaryocyte. 5HT cannot be synthesised in megakaryocytes and like platelets these cells must accumulate it from the blood. Uptake of 5HT into specific subcellular organelles has been demonstrated (Tranzer et al., 1972), and it is possible that these organelles may represent the precursors of 5HT storage organelles in platelets.

In response to Thrombin and A23187, Miller (1983) demonstrated the release of ATP from guinea pig megakaryocytes. However, although one megakaryocyte contains about 200-300 times as much ATP as does one platelet, the extent of ATP

- 32 -

release induced from each megakaryocyte is only about 34 times as great as that per platelet when Thrombin is used, or about 63 times as great when using A23187. This discrepancy in amount present in each cell and amount released may be resolved by postulating either that the releasable nucleotide pool develops late in megakaryocyte life and due to heterogeneous population of megakaryocytes, one may get less ATP released or, equally likely, ATP may enter a storage pool compartment relatively early in megakaryocyte development, but the ability of the cell to undergo degranulation - and hence possess the machinery to enable it to perform this function - may occur late in maturation and hence are incapable of eliciting a full response to platelet stimuli.

These studies serve to demonstrate that in addition to imparting structural and functional aspects to platelets, megakaryocytes are also capable of undergoing several "plateletlike" responses subsequent to addition of platelet stimuli, thus serving to strengthen further the teleological relationship between these two cells.

1.2.4 Physiological Functions

The physiological function of megakaryocytes can be seen as maintaining the status quo in terms of platelet production. The manner in which the body regulates the production of platelets has already been alluded to. This section deals with the manner in which platelets are elaborated from the mature megakaryocyte and the nature of the released platelets.

- 33 -

Platelet release from the mature megakaryocyte is poorly understood and at least three mechanisms have been postulated. (i) Megakaryocytes located in the bone marrow have been demonstrated to extend pseudopodia from the cell, seeking out These pseudopodia have been shown to the marrow sinusoids. contain microtubules orientated along their axis thus affording rigidity (see Penington, 1981). Once they have passed through (Wright, 1910) the walls of the marrow sinusoids the pseudopods fracture thus releasing platelets. (ii) Some megakaryocytes may actually pass through the sinusoids intact and be carried via the venous circulation to the lung. Once lodged in the microcirculation of the lung they could fracture and release platelets (Kaufman et al., 1965; Trowbridge et al., 1982); (iii) Recent studies of Zucker-Franklin and Petursson (1984) (See abo Paulus, 1975) have suggested that the entire megakaryocyte cytoplasm disintegrates in situ and release platelets in this fashion. It must be emphasized that regardless of the method of release, the "platelets" released from the megakaryocyte are not fully The transition from these "platelets" to the developed. typical discoid platelet probably occurs in the general Additionally, those "platelets" which may be circulation. formed in the bone marrow are non-adhesive as they pass through the marrow sinusoids. They probably develop their adhesive capacity by "taking-up" plasma components on to their surface (Behnke, 1969).

Platelets are known to be composed to a heterogenous population with respect to their size. Paulus (1975) argued

- 34 -

that as platelet formation is characterised by a random fragmentation of megakaryocyte cytoplasm, each megakaryocyte must therefore produce a log normal distribution of platelets, including large and small platelets. This postulate was later confirmed, on purely mathematical grounds, by Trowbridge <u>et al</u>. (1982). The number of platelets reportedly produced from one megakaryocyte varies but based on ultrastructural studies, it appears that perhaps as many as 3000 platelets may be elaborated from each megakaryocyte (Chernoff <u>et al</u>., 1980).

1.2.5 Pathological Functions

Megakaryocytes are the progenitor cells of platelets and have been shown to contain most of the constituents found in platelets. Any derangement in elaboration of platelets from the megakaryocyte, or premature release of certain platelet constituents from the magekaryocyte may result in the development of pathological situations.

Trowbridge <u>et al</u>. (1982) have suggested that all platelets are produced by physical fragmentation of megakaryocytes lodged in the lung. The same authors (Martin <u>et al</u>., 1983a; 1983b) have suggested that abnormal platelet production (due either to abnormalities in the megakaryocyte itself or in the pulmonary vasculature) may be involved in the pathogenesis of vascular disease. A correlation between abnormal platelet size and reactivity previously has been made (Karpatkin, 1978; see Eldor <u>et al</u>., 1982).

During normal platelet production it is possible for platelets to be derived from each of the principal ploidy classes (8N, 16N and 32N) of megakaryocytes (Odell and Jackson, 1968). Persington <u>et al</u>. (1976) have suggested that platelets produced from each individual class of ploidy may elaborate platelets with differences in reactivity. It is quite possible that a propensity of any one of these types may result in initiation of pathological diseases. Such a possibility awaits further investigation into the nature of each type of platelet produced by the different megakaryocytes.

Studies have shown that human megakaryocytes contain growth factor(s) with similar biological properties to platelet derived growth factor (Castro-Malaspina <u>et al</u>., 1981).

It has been suggested that megakaryocytes may play a role in the pathogenesis of the marrow fibrosis observed in myeloproliferative disorders by stimulating fibroblast proliferation and collagen secretion (Castro-Malaspina <u>et al</u>., 1981). In myelofibrosis, abnormalities in megakaryocyte morphology and ineffective megakaryocytopoiesis are common. The above authors suggest that this combination may lead to excessive concentrations of megakaryocyte derived growth factor accumulating with the consequent initiation of myeloproliferative disorders.

- 36 -

Receipt and transduction of external signals initially depends upon surface phenomena and in particular upon the existence of membrane associated 'cognitive' elements or receptors. In many cases, these have been directly identified and partially or completely purified; in other systems their existence is entirely hypothetical. Receptor proteins, which may be in the form of monomeric or oligomeric subunits are assumed, per se, to span the phospholipid bilayer of the plasma membrane and contain ligand binding sites, recognizing specific ligands, on the outer surface. The mechanism whereby ligand-receptor interaction on the cell surface evokes an intracellular response involves the action of a'transductionprocess'. Activation of these transduction processes elaborate intracellular effector molecules or second messengers which can then elicit activation (or inhibition) of the cell (Berridge, 1980). In a variety of different cell types, including human platelets, the more common effector molecules or second messengers are cyclic 3',5' adenosine monophosphate (cyclic AMP) and calcium (Berridge, 1980, 1981). These second messengers are the keys to controlling many of the enzymes and protein complexes responsible for cellular activation.

The most notable exception to the second messenger theory for controlling cellular reactivity is the cholinergic nicotinic receptor. This receptor is coupled to a sodium channel whose opening leads to a rapid change in membrane potential, the extracellular signal being transmitted intracellularly by an electrical impulse rather than a chemical messenger. The transduction process controlling generation of cyclic AMP has been well characterised in a number of different cell types however, by comparison the mechanisms controlling the generation of Ca²⁺ signals is, at present, highly speculative.

1.3.1 The Second Messengers

1.3.1.1 Cyclic AMP (cyclic adenosine-3',5'-monophosphate)

Cyclic AMP was first proposed as an intracellular mediator of the effects of glucagon and adrenaline in the liver by Sutherland and Rall (1960). Since this discovery innumerable studies have addressed the generation of this nucleotide, following agonist exposure, and in a plethora of different cell types cyclic AMP is now accepted as being a ubiquitous regulatory agent in most mammalian tissues (Robison <u>et al</u>., 1971; Gramer and Shultz, 1977).

The enzyme adenylate cyclase, located on the cytoplasmic face of the plasma membrane, forms cyclic AMP from ATP in the presence of Mg²⁺. The adenylate cyclase complex exists as an oligomeric unit consisting of at least three separate components. These include the hormone receptors, which discriminate the specificity of the hormone signals; the catalytic subunit, which catalyses the conversion of ATP to cyclic AMP; and guanine nucleotide (ATP) binding regulatory entities, which are involved in the coupling between the agonist receptors and the catalytic subunit (Ross and Gilman, 1980). It has been shown that the hormone receptors are molecularly separate from the rest of the adenylate cyclase system (Birnbaumer and Rodbell, 1976; Limbird and Lefkowitz,

- 38 -

1977) whilst cell fusion experiments have shown that the receptors float freely in the plasma membrane (Schram and Orly, Extensive research on the transduction mechanisms of 1976). stimulatory and inhibitory agonists has provided strong evidence that two GTP-binding proteins, termed N_S and N_i , are involved in the coupling between stimulatory and inhibitory agonist receptors respectively, and the catalytic subunit. N_{c} mediates the agonist-induced stimulation of the adenylate cyclase whereas N₁ is responsible for the hormone-induced inhibition of the adenylate cyclase (Ross and Gilman, 1980; Jakobs et al., 1984). The guanine nucleotide-binding regulatory units are oligomeric proteins composed of three non-identical subunits, termed an α -, β - and a χ -subunit (Northup <u>et al</u>., 1980; Codina et al., 1984; Hildebrandt et al., 1985). There appears homogeneity in the size of the β - and χ sub-units, however the α -subunit - the source of high affinity binding sites for GTP appears to vary somewhat in its molecular size.

Following agonist-receptor interaction, the receptor associates with N_S , facilitating the release of GDP and binding of GTP to the α subunit, which then dissociates into its α and β subunits (Gilman, 1984; Schram and Selinger, 1984; Jacobs <u>et al.</u>, 1986; Taylor and Merritt, 1986). The subunits remain dissociated for as long as GTP is bound to the α subunit; it is the dissociated α_s subunit which stimulates adenylate adenylate cyclase activity and hence elevates cytoplasmic levels of cyclic AMP. Agonists exhibiting a negative effect on the accumulation of cyclic AMP bind to separate receptors which preferentially combine with N_i , however the mechanism

- 39 -

whereby dissociation of N_i leads to inhibition of adenylate cyclase is not entirely clear; it may involve the inactivation of $N_s(\alpha_s)$ by the released β subunit (Gilman, 1984; Smigel et al., 1984). Mechanisms exist to terminate the effects on adenylate cyclase of stimulatory or inhibitory ligands. This is understood to be performed by GTPase's catalysing the conversion of GTP to GDP thereby promoting reassociation of Evidence the subunits and hence termination of the response. exists that the GTPase activity is an intrinsic feature of GTP-binding proteins (Ross and Gilman, 1980; Cassel and Selinger, 1976; Brandt et al., 1983; Jakobs et al., 1985; Birnbaumer et al., 1985; Jakobs et al., 1986) in particular with the α -subunit. The structural and functional identification of both N_s and N_i owes a considerable amount to two bacterial exotoxins, cholera toxin and pertussis toxin (IAP). The former toxin causes the NAD⁺-dependent ADP-ribosylation of the α -subunit of N_s (Van Heyningen, 1977; Gilman, 1984). Thi s process also depends upon GTP and leads to the activation and dissociation of N_s into its constituent subunits. It is believed that cholera toxin inhibits an inherent GTP-ase activity associated with α -N, hence allowing it to remain activated. In contrast, IAP ADP-ribosylates the α -subunit of N_i but, in this instance, it blocks the dissociation and activation of this regulatory protein and this obliterates inhibitory responses (Gilman, 1984; Jakobs et al., 1984; Ui, 1984). In this fashion, both these compounds can regulate the levels of cellular cyclic AMP. Both compounds however

need to gain access to platelets, being inactive in intact preparations. Nevertheless, cholera toxin and IAP should prove useful tools in delineating further the mechanism of action of the adenylate cyclase system in future studies.

As mechanisms exist to terminate the activity of adenylate cyclase, mechanisms must likewise exist to terminate the actions of its product, the second messenger cyclic AMP. In the case of cyclic AMP this is achieved by selective cyclic AMP phosphodiesterases which hydrolyse any cyclic AMP formed to the inactive 5' AMP. In human platelets three different types of cyclic nucleotide phosphodiesterases have been isolated (Hidaka and Asano, 1976). Rat platelets contain two forms, one is a cyclic GMP-binding phosphodiesterase, which hydrolyses cyclic AMP but acts preferentially on cyclic GMP and the other is specific for cyclic AMP (Coquil <u>et al.</u>, 1980).

In platelets an elevation of cyclic AMP is associated with inhibition of reactivity (Moncada and Whittle, 1985; Haslam et al., 1978). Agents known to elevate cytoplasmic levels of cyclic AMP are the most powerful known inhibitors of platelet function (Packham, 1983) and include the following:-Prostaglandins such as PGI_2 , PGE_1 and PGD_2 ; Adenosine, diterpenes such as forskolin; and non-specific phosphodiesterase inhibitors such as isobutyl methyl xanthine.

Platelets contain high affinity binding sites for PGI_2 , PGE_1 and PGD_2 (reviewed by MacIntyre, 1985). Studies (Whittle <u>et al</u>., 1978) have shown that PGI_2 and PGE_1 share a common receptor on the surface of platelets from various species, whilst PGD_2 acts at a distinct receptor (Mills and MacFarlane,

- 41 -

1974; Whittle <u>et al</u>., 1985). Studies examining the effects of prostaglandins on the elevation of cyclic AMP in rat platelets have revealed that, even at relatively high concentrations, PGD_2 is unable to inhibit agonist-induced aggregation (Hwang, 1980) implying that rat platelets lack receptor sites for PGD_2 .

Adenosine is also known to elevate cyclic AMP levels and inhibit platelet reactivity in both human and rat platelets (Michel <u>et al</u>., 1976). However, it appears that rat platelets are less sensitive to the inhibitory effects of adenosine compared to human platelets, an effect shown to be quantitative rather than qualitative (Michel <u>et al</u>., 1976).

The diterpene forskolin, isolated from the roots of Coleus forskolii, has proved immensely useful in probing the adenylate cyclase system in a variety of different cell types. Thiscompound can bypass the system at the receptor level and act directly on the catalytic unit of the enzyme (Seamon and Daly, 1981) and/or on the stimulatory GTP-binding protein, N_s (Durfler et al., 1982) thus elevating levels of cyclic AMP. Forskolin is a potent stimulant of platelet cyclic AMP formation and inhibitor of ADP induced platelet aggregation (Siegl et al., 1982). In human platelets, incubated with Quin 2, Sage and Rink (1985) demonstrated forskolin to elevate cytoplasmic levels of cyclic AMP with a concomitant inhibition agonist-induced aggregation and rises in cytoplasmic free calcium, the latter believed to be one of the mechanisms by which elevation of cyclic AMP inhibits platelet reactivity (vide infra).

- 42 -

Lastly, inhibition of the cyclic AMP phosphodiesterase is another mechanism by which cytosolic levels of cyclic AMP can be elevated. The non-specific phosphodiesterase inhibitors such as isobutyl methyl xanthine can inhibit platelet reactivity by elevating cyclic AMP levels. However several compounds exist which are allegedly specific for platelet cyclic AMP phosphodiesterase. One such compound is Trequinsin (Ruppert and Weithmann, 1982). These researchers found this compound to be at least 100 times more active as an inhibitor of arachidonic acid-induced platelet aggregation compared to PGI2, a compound previously thought to be amongst the most active compounds at inhibiting platelet activation (Moncada and Vane, 1979; Moncada and Whittle, 1985). Another compound shown to completely inhibit cyclic AMP phosphodiesterase in platelets is RO-15-2041, an analogue of anagrelide (Holck et al., 1984).

So far the effects of agents elevating levels of cyclic AMP and their ability to inhibit platelet reactivity subsequent to agonist addition has been discussed. What is not known is whether or not inhibition of adenylate cyclase and hence reduction in cyclic AMP levels plays any part in platelet reactivity. Several known pro-aggregatory agents are known to exert a negative effect on accumulation of cyclic AMP. Adrenaline and ADP have been demonstrated to reduce the concentration of cyclic AMP which has been previously elevated by adenylate cyclase stimulants (Mills, 1974). They can also inhibit basal and hormone-stimulated adenylate cyclase activity in platelet particulate fractions (Jakobs <u>et al.</u>, 1978;

- 43 -

Cooper and Rodbell, 1979). The inhibition of adenylate cyclase by agents such as adrenaline involves the participation of the regulatory protein N_1 and requires GTP (Jakobs <u>et al</u>., 1978). Other pro-aggregatory agents e.g. ADP (Mellwig and Jakobs, 1980); platelet-activating factor (PAF-acether) (Williams et al., 1983) and Thrombin (Aktories and Jakobs, 1984) have also demonstrated, to varying degrees, the inhibition of adenylate cyclase and decrease in levels of cyclic AMP in (human) platelets - this effect is observed only in platelet lysates when either Thrombin or PAF-acether are used. While these pro-aggregatory compounds have been shown to decrease levels of cyclic AMP in their various studies, what is not known is the physiological significance of the effect. Haslam et al. (1978a) suggested that the inhibition of adenylate cyclase is only physiologically significant in the context of reducing the stimulation of the enzyme by positive agonists such as PGI, and adenosine.

1.3.1.2 Mechanisms of action of cyclic AMP

Elevated levels of cyclic AMP elicit a wide range of responses in the eukaryotic cell from stimulating glycogenolysis in the liver, to water readsorption in the kidney to insulin secretion in the pancreas (Sutherland and Robinson, 1966; Sutherland <u>et al</u>., 1968; Rasmussen, 1970). Cyclic AMP mediates these diverse responses by stimulating the structural modification of pre-existing proteins within its target tissue (Kuo and Greengard, 1969). The nature of this modification is the formation of a phosphate ester linkage within the hydroxyl groups of specific serine and threenine residues of

- 44 -

a protein substrate using ATP as a donor. The specific phosphorylation of certain platelet proteins has been observed after elevation of cyclic AMP and are mediated by cyclic AMPdependent protein kinases. Several membrane proteins are phosphorylated (Steiner, 1975; Apitz-Castro et al., 1976) and one of these, a protein with an apparent mass greater than 400K, shows enhanced incorporation of labelled phosphate accompanying an elevated cyclic AMP (Apitz-Castro and Murcian, Haslam and coworkers (1979, 1980) observed that elevation 1978). of cyclic AMP led to the rapid phosphorylation of proteins with apparent molecular weight 24K and 50K and slower phosphorylation of several other proteins, including one of It thus appears that potentially many proteins can be 22K. phosphorylated in the presence of elevated levels of cyclic AMP. However, it is the steady state level or the rate and direction of change of these phosphoproteins which is more directly related to cellular reactivity. Dephosphorylation of the protein, by hydrolysis of the phosphate ester, is catalysed by protein phosphatases. Acting through these, as yet, largely unidentified protein substrates, elevated levels of cyclic AMP are able to inhibit or terminate cellular reactivity. Potential mechanisms by which cyclic AMP may inhibit platelet reactivity are outlined below.

(i) Inhibition of Ca²⁺ availability: One of the mechanisms by which cyclic AMP regulates platelet functions is by controlling the level of Ca²⁺ in the cytosol. Although this has long been assumed to be so, no direct evidence was available until

the introduction of Quin 2 made such experiments feasible (see Section 2.2.9.2). In Quin 2 loaded platelets, the elevation in cytoplasmic free calcium elicited by Thrombin is inhibited by prior exposure to PGD,, PGE, or forskolin (Feinstein et al., 1983a: Sage and Rink, 1985). When added at the peak response in Quin 2 fluorescence, either of the above compounds or the membrane permeant cyclic AMP derivative, 8-bromo-cAMP resulted in a rapid fall in Quin 2 fluorescence (Feinstein et al., 1983a; MacIntyre et al., 1985a). Finally Sage and Rink (1985) also demonstrated that forskolin could inhibit, with equal potency, the Ca²⁺ influx and mobilization induced by hrombin or platelet activating factor. These results strongly suggest that cyclic AMP is able to regulate Ca²⁺ availability in intact platelets. However, resting levels of Ca²⁺ do not appear to be regulated by cyclic AMP, since agents known to elevate cytoplasmic levels of cyclic AMP e.g. PGI2 and PGD2, do not affect resting Ca2+ . (MacIntyre et al., 1985a). Additionally, since platelet stimulatory agonists do not elicit cyclic AMP formation to any great extent, it is unlikely that unless exposed to an adenylate cyclase stimulant, endogenous cyclic AMP would subserve a physiological role in regulating Ca²⁺ availability. Platelets are thought to contain systems controlling calcium transport. Such transportation of calcium may be stimulated by platelet cyclic AMP-dependent protein kinases. Kaser-Glanzman and co-workers (1977, 1979) demonstrated that cyclic AMP could stimulate the active sequestration of Ca²⁺ into purified platelet intracellular membranes, thought to correspond to the dense tubular system, an organelle thought to store Ca²⁺ (White, 1972). Interestingly, using cytochemical techniques,

- 46 -

the existence of adenylate cyclase has been demonstrated in the dense tubular system (Cutler <u>et al.</u>, 1985). However, studies by Menashi <u>et al</u>. (1982) have failed to demonstrate a cyclic AMP-dependent accumulation of Ca²⁺ into intracellular platelet membrane fractions.

(ii) Inhibiting the generation of putative second messengers: The metabolism of inositol phospholipids, in particular phosphatidy inositol(4,5)P₂ (Ptdins(4,5)P₂) is believed to be involved in the generation of second messengers, in particular, 1,2 diacylglycerol (the endogenous activator of protein kinase C) and inositol(1,4,5)trisphosphate (thought to be involved in the elevation of cytosolic free calcium) (see Section 1.3.1.6 and 1.3.1.8).

Phospholipase C, a phosphodiesterase type enzyme is believed to catalyse the metabolism of inositol phospholipids. The finding that dibutyrlcAMP, a membrane permeant derivative of cyclic AMP can inhibit phospholipase C action in platelets (Rittenhouse-Simmons, 1979; Lapetina <u>et al</u>., 1977) is consistent with the idea that cyclic AMP may act by inhibiting the availability of calcium. Furthermore, numerous studies have demonstrated that agents elevating cytoplasmic levels of cyclic AMP can inhibit the metabolism of PtdIns(4,5)P₂ (Feinstein <u>et al</u>., 1985) and reduce the formation of $Ins(1,4,5)P_3$ (Watson <u>et al</u>., 1984). Interestingly cyclic AMP has also been demonstrated to stimulate the formation of PtdIns(4)P in intact platelets (de Chaff**o**y de Courcelles <u>et al</u>., 1986). Besides being a precursor for the formation of PtdIns(4,5)P₂, PtdIns(4)P has also been implicated in Ca²⁺ transport

- 47 -

(Varsanyi <u>et al</u>., 1983; Choquette <u>et al</u>., 1984). The possible PtdIns(4)P-induced depression of cytoplasmic levels of calcium might evidently contribute to the inhibitory effects observed with elevated levels of cyclic AMP.

(iii) Inhibition of protein phosphorylation: The phosphorylation of a wide, and varied collection of proteins occurs subsequent to agonist addition in platelets and is monitored using platelets prelabelled with $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$. Lyons <u>et al</u>. (1975) were the first to demonstrate that there were two major proteins phosphorylated on stimulation of platelets with thrombin. The first was a protein of molecular weight 40K, now known to be 47K (depending upon techniques used this protein can range from 40-47K). The second was a protein of molecular weight 20K. The identity of the latter protein was established by Daniel et al. (1981) as being myosin light chain. The former protein has been isolated by Imaoka et al. (1983) however its precise function in platelet reactivity is not clearly established. Recently it has been suggested that it may be a lipocortinlike entity, capable of inhibiting phospholipase A_{2} (Touqui et al., 1986). Upon agonist-induced phosphorylation of the 40K protein by protein kinase C, the inhibitory activity is suppressed and may possibly represent the mechanism underlying mobilization of arachidonic acid in platelets (see Section 1.1.3.4). In a separate study Connolly and Majerus (1986) proposed that the 40K protein may act as $Ins(1,4,5)P_3$ phosphatase, cleaving the 5 phosphate from $Ins(1,4,5)P_3$. Feinstein <u>et al</u>. (1983b) demonstrated that PGD_2 or forskolin could reverse the the Thrombin-induced phosphorylation of both the 20K, and 40K

proteins. In the presence of agents elevating cyclic AMP, phosphorylation of both proteins was again inhibited (Lyons et al., 1975; Fox et al., 1978).

Both myosin light chain and the 40K protein are believed to be intrinsically involved in platelet activation; 40K protein phosphorylation has been correlated with serotoning secretion, and 1,2-diacylglycerol formation (Ieysan <u>et al</u>., 1982; Sano <u>et al</u>., 1983; Kawahara <u>et al</u>., 1983); phosphorylation of myosin light chain is correlated with secretion and clot retraction (Daniel <u>et al</u>., 1977).

Although potentially many other mechanisms may be regulated by elevated levels of cyclic AMP the exact biochemical consequences of an increase in cytoplasmic cyclic AMP remains to be ascertained.

1.3.1.3 <u>Calcium</u>

In 1883, the British physiologist Sidney Ringer demonstrated that the calcium ion was required in the bathing medium to sustain contraction of an isolated frog heart. Almost a hundred years later, it was demonstrated that 'skinned' muscle fibres could be induced to contract following application of external ions (Natori, 1954). Within a few years considerable evidence accumulated in support of the concept that calcium ions play a critical role in excitation-contraction coupling, a term introduced by Sandow (1952). In 1961, Douglas and Rubin emphasized the importance of free ionized calcium as a ubiquitous intracellular coupling agent by showing that it was an essential component in the acetycholine-induced

- 49 -

release of catecholamines from the adrenal medulla. From this study the term "stimulus-secretion coupling" was derived to characterise stimulated exocytosis of hormones and secretory products. Since this time many events have been shown to be dependent on the presence of free ionized calcium ions and its role as a ubiquitous mediator of cellular reactivity is not disputed (Rasmussen and Barret, 1984; Rubin, 1982). Blood platelets can be stimulated by a wide range of chemically different compounds such as Thrombin (a proteolytic enzyme); collagen (a macromolecular complex); low molecular weight compounds such as ADP and thromboxane A_2 and vasoactive amines such as adrenaline. Nonetheless the response generated e.g. shape change, aggregation, secretion and icosanoid synthesis, seems virtually indifferent of the agonist used. This implies that a common "effector system" is operating. The role of cytosolic free calcium in platelet function has been extensively investigated and is generally accepted as the intracellular effector molecule in platelets (for reviews, see Detwiler et al., 1978; Feinstein, 1978; Gerrard et al., 1981). While the evidence for this is partly circumstantial and indirect, the essential evidence to indicate that calcium plays some regulatory role is available. The earliest evidence for a role in platelets was the work of Grette (1962) who demonstrated in porcine platelets a requirement for external Ca²⁺ for hrombin-induced secretion at 15° C, but Ca²⁺ was not required at higher temperatures. Murer (1972) confirmed these findings using human and porcine platelets. The increased requirement for Ca^{2+} at the lower temperatures was explained by the reduction

- 50 -
in effectiveness of an activator at this temperature. Indirect evidence of an essential role of Ca²⁺ in platelet activation derives from the following (i) Calcium ionophores:-The demonstration that the calcium ionophore A23187 could elicit the full platelet response e.g. shape change, aggregation, secretion and icosanoid synthesis (Feinman and Detwiler, 1974; White et al., 1974; Massini and Luscher, 1974) in a manner similar to that elicited by agonists interacting at specific platelet receptors proved to be amongst the first in a long line of indirect evidence expounding a role for Ca²⁺ in platelet activation. One notable difference in its effects on platelets appears to be its less efficacious nature of eliciting the metabolism of inositol phospholipids - a characteristic feature of many platelet agonists. The mechanism of action of ionophores, in particular A23187, is believed to involve the formation of a 'cage' by the combination of A23187 with Ca^{2+} . This 'cage' is lipophilic thus facilitating the transport of Ca²⁺ across membranes. However various workers (Holmsen and Dangelmaier, 1981; Lages and Kruger, 1983) have reported that only free A23187, and not divalent complexes, could activate platelets. (ii) $\frac{45_{Ca}^{2+}}{2}$ The use of this entity to measure Ca²⁺ fluxes in a variety of cells is hampered by the difficulty in interpreting data obtained from such studies. However, Brass (1984a,b) has recently used this probe to investigate Ca²⁺ homeostasis in human platelets providing useful information concerning intracellular stores of this ion.

- 51 -

(iii) Chlortetracycline CTC:- CTC can be used to measure mobilisation of membrane-associated divalent ions such as Ca²⁺ (Caswell and Hutchison, 1971a; Caswell and Hutchison, 1971b; Caswell, 1972). This compound forms a fluorescent complex with Ca^{2+} ion, the intensity being 20-100 times more intense in a nonpolar environment. Upon releasing Ca²⁺ the intensity of fluorescence decreases. Feinstein (1980) was the first to demonstrate release of intracellular Ca²⁺ in platelets using this compound. Upon Thrombin addition, release of intracellular Ca²⁺ occurred with a time-course consistent with its proposed role in activation-secretion coupling. Later studies by Thompson and Scrutton (1985) confirmed these observations. However as a probe for measuring intracellular Ca²⁺ they concluded, from their studies, that it was not possible to define which membrane site(s) were involved. Additionally, quantification of the CTC fluorescence signal is difficult. (iv) Calcium antagonists e.g. TMB-8:- Studies by Malagodi and Chiou (1974; 1974a) demonstrated that 8-(N,N-diethylamino)octyl 3,4,5-trimethoxybenzoate (TMB-8) was a potent inhibitor of the contractility of both smooth and skeletal muscle. Subsequent studies demonstrated that a possible mechanism of action of TMB-8 could be to reduce Ca^{2+} -availability by stabilizing Ca^{2+} binding to cellular Ca²⁺ stores and thereby inhibiting its release upon agonist addition (Chiou and Malagodi, 1975). By analogy in platelets, it is believed that TMB-8 acts as an intracellular Ca²⁺-antagonist inhibiting the agonist-induced mobilization of Ca²⁺ (Charo <u>et al</u>., 1976). Several groups

- 52 -

have used this compound to investigate activation processes in platelets (Charo <u>et al.</u>, 1976; Murer and Siojo, 1982; MacIntyre and Shaw, 1983). However, several effects of TMB-8 in platelets have been shown not to be entirely attributable to Ca^{2+} antagonism e.g. TMB-8 interferes with formation of thromboxane, apparently by inhibiting cyclooxygenase (Shaw, 1981). Additionally Simpson <u>et al</u>. (1984) demonstrated TMB-8 to be less effective at inhibiting Thrombin-induced Ca^{2+} flux or internal release in human platelets, but more effective at inhibiting the secretory response evoked by phorbol esters at basal Ca^{2+} . Phorbol esters are known activators of protein kinase-C, an enzyme thought to be involved in platelet reactivity (see Section 1.3.1.8.3). Hence, the use of TMB-8 in probing Ca^{2+} ion in platelets is debatable.

(v) <u>Permeabilized platelets</u>:- By rendering platelets permeable to Ca-EGTA (and other small molecules) using high-voltage electrical fields, the role of intracellular calcium in platelet activation can be examined (Knight and Scrutton, 1980; Knight <u>et al.</u>, 1982). The addition of buffered micromolar concentrations of Ca²⁺, without an agonist was sufficient to induce dense granule secretion (Knight and Scrutton, 1980). Results obtained using permeabilized platelets are plagued by the fact that such procedures allow entry of other low molecular weight compounds and the attendent possibility of internal damage. Nonetheless, they represent a valuable model for studying the role of Ca²⁺ ions in platelet activation.

• 53 -

(vi) Intracellular Ca²⁺ indicators:- Within the past couple of years the development of indicator dyes whose fluorescence changes upon chelation with Ca²⁺ ions has contributed vastly towards our understanding of the role of Ca²⁺ ions in platelet activation. In particular the custom built fluorescent probe Quin 2 (Tsien, 1981) has been of considerable value. This dye is introduced into cells in an ester form, Quin 2 AM. The hydrolysed form, Quin 2, reports intracellular levels of free Ca²⁺ (Tsien <u>et al</u>., 1982a, 1984; 1985, see also Section 2.2.9.2). Using this technique, in human platelets, Rink <u>et al</u>. (1982) demonstrated that Thrombin or ionomycin caused a very rapid increase in fluorescence, indicating an increase in cytosolic calcium ion activity. Since then numerous workers have shown

similar results with a variety of different agonists (see MacIntyre et al., 1985b, for a recent review).

It is clear from the above citations that a number of techniques are available to us for probing the role of Ca^{2+} in platelet activation. To date the most explicit information has come from using Quin 2 (however newer and better Ca^{2+} -indicator dyes have been developed and are discussed in Section 2.2.9.2). Using this technique, in human platelets, Rink and Coworkers (1981) recorded basal cytoplasmic free calcium concentrations Ca^{2+} , to be approximately 100 nM. In a subsequent study, Rink and coworkers (1982) recorded the threshold levels of Ca^{2+} , required for shape change, 5HT secretion and aggregation induced by A23187. They were 0.5 μ M; 0.8 μ M and 2 μ M respectively. In the same study however they

demonstrated the ability of Thrombin to evoke shape change and secretion at resting Ca^{2+} levels. They suggested that alternative pathways must exist for platelet activation other than elevation of Ca^{2+} . Similar conclusions were drawn by Knight <u>et al</u>. (1982) using permeabilised platelets. They observed that different concentrations of Thrombin were required to elicit dense granule and lysosomal release whereas a similar concentration of Ca^{2+} could elicit both processes, implying an additional pathway regulating platelet reactivity.

Assuming that platelet activation is mediated, at least in part, by calcium ions then a ready supply of these ions must be obtained, possible sources being either intracellular or extracellular in nature.

The homeostasis of calcium in human platelets has recently been subject to detailed analysis (Brass, 1984a,b,c). With respect to possible intracellular sources of Ca^{2+} , Brass demonstrated the existence of two pools of Ca^{2+} in human platelets one cytosolic in nature and the other possibly the dense tubular system. The cytosolic pool appeared to more rapidly exchanging pool with a t.1/2 of Ca^{2+} exchange approximately 17 min. All of the ${}^{45}Ca^{2+}$ that had entered this pool could be released by digitonin. However, the possibility exists that the calcium released acrued from a number of separate pools of calcium, the existence of which has been postulated by Johnson <u>et al</u>. (1985). For many years the dense tubular system in platelets has been proposed to subserve a similar role in storing calcium as does the sarcoplasmic reticulum of muscle cells (White, 1972). Various studies using intracellular

membrane fractions derived from human platelets - thought to correspond to the dense tubular system - have demonstrated the accumulation of Ca²⁺ in the presence of ATP (Kaser-Glanzmann et al., 1977; 1978; 1979; Menashi et al., 1982; Menashi et al., 1984). Other histological (Cutler et al., 1978) and immunological (Dean and Sullivan, 1982) evidence also support the existence of an ATP-dependent Ca²⁺ sequestrating activity in human platelet membranes, supposedly corresponding to the dense tubular system. In Brass's study, the second, more slowly exchanging pool (t.1/2 of several hundred minutes) was assumed to be the dense tubular system and acted to 'buffer' the size of the rapidly exchanging pool by helping to remove Ca²⁺ from the cytosol. The studies of Brass and those of Kaser-Glanzman et al. (1977; 1978; 1979) and Menashi and colleagues (1982; 1984) help confirm that the dense tubular system probably does represent the most important intracellular store for calcium. Other possible intracellular Ca^{2+} -stores in platelets are membrane-associated calcium mitochondria or dense granules. Membrane associated calcium has been demonstrated using the CTC fluorescence technique and may play an essential part in regulating Ca²⁺- availability after receptor occupation. Platelets contain many mitochondria and by analogy with other cells presumably accumulate calcium. However studies using inhibitors of mitochondrial respiration suggest they play no significant role in generating availability of Ca²⁺ for platelet reactivity (Friedman and Detwiler, 1975). In human platelets up to 60% of platelet Ca²⁺ is stored in dense granules (Murer, 1969; Skaer, 1975) (although this figure is questioned by

Feinstein (1978) in a secretory pool that is not in equilibrium with the cytosol and that presumably plays no role as a second messenger.

In plasma the free extracellular concentration of Ca^{2+} is approximately 1 mM and hence represents a potential source of calcium required for platelet reactivity. Mechanisms governing influx of Ca²⁺ into platelets are ill-defined. However, one possible mechanism by which Ca²⁺ could gain entry to the cell is through a calcium 'gate' or 'channel'. To date a few studies have been performed which suggest the existence of a calcium channel in the platelet plasma membrane. In his study, Brass (1984a) demonstrated that Ca^{2+} influx was maximal at free Ca^{2+} concentrations 20-fold less than in plasma and inhibited by moderate concentrations of Gd^{3+} and La^{3+} (< 100 μ M). He concluded that Ca^{2+} influx occurs through a selective Ca^{2+} channel. In a separate study Hallam and Rink (1985a) demonstrated that when added to Quin 2 loaded platelets, the divalent cation Mn²⁺ guenched the fluorescence signal as a result of its well known property to bind avidly to Quin 2 (Arslan et al., 1985). They postulated that Mn²⁺ and Ca²⁺ enter through some sort of Ca²⁺-channel in the plasma membrane.

Like most cells, platelets possess mechanisms which maintain levels of $[Ca^{2+}]$, within defined limits as well as restoring agonist-induced elevation of Ca^{2+}_{1} . In other cells both Ca^{2+} -sequestration and extrusion processes operate (Rasmussen and Barett, 1984). Similar processes may operate in platelets. While there appears to be good evidence for a Ca^{2+}/Mg^{2+} ATPase existing in the dense tubular system, acting to sequester

- 11 -

Ca²⁺ (Cutler <u>et al</u>., 1978; Menashi <u>et al</u>., 1982; 1984; Dean and Sullivan, 1982), the mechanisms governing Ca²⁺ extrusion from platelets is less well understood. The mechanism does not appear to be a Na-Ca or Ca-Ca exchange process since in his studies Brass (1984a,b) could not demonstrate a dependency on either Ca²⁺ influx or transmembrane Na⁺ gradient for Ca^{2+} efflux. The existence of an ATP-dependent Ca^{2+} sequestrating mechanism acting in platelet membrane lysates has been demonstrated (Cutler et al., 1978; Dean and Sullivan, 1982). However, histochemical studies have failed to demonstrate the enzyme in the platelet plasma membrane (Cutler et al., 1978). Finally, Brass (1984c) has recently suggested that glycoproteins 11b/111a may be involved in Ca²⁺ efflux in platelets. These proteins span the platelet plasma membrane and may form a Ca^{2+} -channel. It is clear that the mechanisms regulating Ca^{2+} availability in platelets are complex and that a variety of mechanisms exist by which agonists may perturb the homeostatic mechanisms generating Ca²⁺ availability required for platelet activation.

1.3.1.4 Mechanism of action of cytosolic free calcium, Ca²⁺

The targets for a regulatory increase in cytoplasmic free $\operatorname{calcium}, \left[\operatorname{Ca}^{2+}\right]_i$, are calcium receptor proteins. These proteins bind Ca^{2+} with high affinity (kd $10^{-8} - 10^{-6}$ M) and thus undergo a conformational change, which leads either to a change in function or a change in ability to interact with other proteins= to change their functions. There are two classes of Ca^{2+} receptor protein (i) true Ca^{2+} receptor proteins such as

calmodulin which is present in all cells in a membrane-bound and soluble form and troponin-C and parvalbumin, proteins found in muscle cells, and (ii) calcium-activated enzymes with no specific calcium receptor subunit such as calcium-activated, phospholipid-dependent protein kinase (Rasmussen and Barrett, 1984). The calcium target for which best evidence exists in platelets is calmodulin (for a recent review of the role of calmodulin in platelets, see Feinstein, 1982). Calmodulin has been identified in platelets (White et al., 1981) and is a low molecular weight calcium binding protein that mediates Ca²⁺ regulation of many processes (Cheung, 1980). It occurs in both the cytosol and bound to membranes and undergoes a conformational change upon the binding of 1-4 Ca²⁺ atoms per molecule which renders it capable of activating a wide range of enzymes. Processes which are believed to be Ca²⁺-calmodulin activated in platelets include (i) stimulation of glycogenolysis. Large amounts of ATP are consumed during platelet activation. The levels of ATP are redressed by stimulation of cellular metabolism. Ca²⁺-calmodulin activates phosphorylase kinase which in turn activates phosphorylase-b to phosphorylase-a, the latter catalysing glycogen breakdown to glucose. Glucose formed in this manner can then synthesize ATP via glycolysis and the citric acid cycle. (ii) Activation of platelet actomyosin. Platelet myosin light chain kinase has been isolated, and shows a requirement for calmodulin (Dabrowska and Hartshorne, 1978; Hathaway and Adelstein, 1979). Phosphorylation of myosin chains allows them to interact with actin filaments and ATP-dependent actomyosin contraction can take place which is involved in

- 59 -

platelet activation and clot retraction (Pollard et al., 1977; White and Gerrard, 1979; Cohen, 1979). Although potentially many Ca-calmodulin activated processes do exist, the only direct and convincing evidence for calmodulin regulating platelet enzymes is for the above cited case. (iii) <u>Regulation</u> of platelet protein phosphorylation. Various proteins are phosphorylated by platelet agonists such as Thrombin, collagen. and A23187 in cells prelabelled with $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ (Lyons <u>et al</u>., 1975; Haslam and Lynham, 1977). The demonstration that the calcium ionophore A23187, at low concentrations, can preferentially phosphorylate the 20K protein (Yamanishi et al., 1983) which corresponds to myosin light chain is definitive proof of an involvement of Ca²⁺ in protein phosphorylation. Other evidence originates from the use of purported calmodulin antagonists such as trifluoperazene (Weiss and Wallace, 1980). Feinstein and Hadjian (1981) demonstrated the inhibition of phosphorylation of a polypeptide of molecular weight 20K as well as one with molecular weight of 47K using trifluoperazine.

Another process thought to be regulated by Ca^{2+} -calmodulin in platelets include activation of phospholipase A_2 which is involved in release of arachidonic acid from certain platelet phospholipids (Wong and Cheung, 1979).

Given the high concentrations of calmodulin present in eukaryote cells, including platelets, it is primarily the intracellular level of free Ca²⁺ and its distribution that will be responsible for controlling platelet reactivity.

1.3.1.5 Metabolic pathways of inositol phospholipids

Before discussing the role of inositol lipid metabolism in stimulus-response coupling it is pertinent to consider the metabolic pathways of inositol phospholipids. The recognized pathways of inositol phospholipid metabolism are detailed in Figures 2a and 2b. The inositol phospholipids, comprising of phosphatidyl inositol (PtdIns); phosphatidyl (inositol-4phosphate (PtdIns(4)P) and phosphatidyl inositol-4,5- P_2 (PtdIns $(4,5)P_{2}$) are located in the inner leaflet of the plasma membrane and are in a state of perpetual turnover. Two different types of enzymes are involved in inositol phospholipid metabolism, namely phosphomonoesterases and phosphodiesterase(s) e.g. phospholipase C. The phosphomonosterases selectively remove phosphate from positions 4 and 5 of the inositol ring and convert $PtdIns(4,5)P_2$ to PtdIns(4)P and the latter to PtdIns. Together with PtdIns and PtdIns(4)P kinases these enzymes are responsible for the turnover of the phosphate in the 4 and 5 positions of the inositol ring, without affecting the mass of these inositol lipids in the steady state. There is some evidence that PtdIns kinase and PtdIns(4)P kinase may be controlled by receptor occupancy (Drummond et al., 1984), perhaps through the actions of protein kinase (since phorbol esters and synthetic 1,2-diacylglycerols can affect the synthesis and levels of the polyphosphoinositides (de Chaffoy Courcelles et al., 1984a,b). Relief of kinases from de product inhibition, as demonstrated for PtdIns(4)P kinase (Van Rooijen et al., 1985), may also account for an increase of flux through the kinases relative to metabolism by the



Metabolic pathways for synthesis and degradation of inositol phospholipids.

Figure 2a:



Figure 2b: Proposed metabolism and interconversion of inositol phospholipids and inositol phosphates.

Some of the known (bold line) and probable (broken line) interconversions of inositol phospholipids and inositol phosphates are depicted.

, who was glassed and the for and profess and

化磷酸钙医磷酸盐 法公共主义的复数形式 法公司法 美国人名

After Michell, 1986

- 63 -

phosphomonosterases. An important point to note is that a dynamic equilibrium exists between the three main inositol phorpholipids and that the kinases and phosphomonoesterases maintaining this equilibrium are amongst the most active enzymes in the cell (Hawthorne and Pickard, 1979). Any change in the level of one of the inositol phospholipids will be rapidly "buffered" by compensatory changes in the levels of other inositol lipid species.

Phosphodiesteratic attack, by phospholipase C, causes the metabolism of all three inositol phospholipids, with the concomitant generation of the water soluble inositol phosphates, InslP, Ins(1,4)P₂ and Ins(1,4,5)P₃, from PtdIns, PtdIns(4)P and $PtdIns(4,5)P_{2}$ respectively, which are then released into the cytosol. Various phosphatases are responsible for metabolising $Ins(1,4,5)P_3$ to $Ins(1,4)P_2$, $Ins(1,4)P_2$ to $InslP_3$ and InslP to Inositol (Segfred et al., 1984; Storey et al., 1984; Joseph and Williams, 1985 and Connolly <u>et al</u>., 1985). The sequential metabolism of the inositol phosphates is probably responsible for the appearance of Ins(1)P since, in a variety of studies, agonist-induced formation of Ins(1)P lags behind formation of Ins(1,4,5)P₃ (Berridge, 1983; Norten, 1983; Rebecchi and Gershengorn, 1983). The addition of lithium has been shown to inhibit the Ins(1)P and $Ins(1,4)P_2$ phosphatases not the Ins(1,4,5)P3 phosphatase (Storey et al., 1984). The consequence of this step is to convert an otherwise complex. cycle into one with one accumulating end product e.g. Ins(1)P, thus allowing easier interpretation of agonist-induced effects on inositol phospholipid metabolism (Berridge et al., 1982).

- 64 -

Recent evidence has been presented demonstrating the existence of a novel inositol trisphosphate, $Ins(1,3,4)P_3$ (Irvine et al., 1984a; figure 2b). The metabolism and kinetics of $Ins(1,3,4)P_3$ are entirely different from those of $Ins(1,4,5)P_3$ which is formed immediately following agonist exposure; $Ins(1,3,4)P_3$ formation tends to be somewhat delayed as its decline on removal of the agonist (Irvine et al., 1985; Burgess et al., 1985). Although it is conceivable that $Ins(1,3,4)P_3$ does subserve some second messenger function, only $Ins(1,4,5)P_3$ has been demonstrated to mobilize intracellular Ca²⁺ (Streb et al., 1983; Irvine et al., 1984b; Burgess et al., 1984a). Evidence has also been provided for the existence of a higher phosphorylated derivative of inositol phosphates, designated inositol 1,3,4,5-tetratrisphosphate $(Ins(1,3,4,5)P_{\mu}, in rat brain slices (Batty et al., 1985),$ hepatocytes (Hansen et al., 1986) and GH_{μ} pituitary cells (Heslop et al., 1985). With very long prelabelling periods even higher phosphorylated derivatives, designated ${\tt InsP}_5$ and InsP₆, have been isolated (Heslop <u>et al</u>., 1985; Figure 2b). However their potential role in cellular reactivity is unknown. The source of $Ins(1,3,4,5)P_{\mu}$ is of crucial significance to its possible role in receptor mechanisms. One possibility is a receptor-linked phospholipase C-induced hydrolysis of an appropriate phospholipid, phosphatidylinositol 3,4,5-trisphosphate $PtdIns(3,4,5)P_3$. Although preliminary evidence for its

existence of this phospholipid in animal cells has been presented (Santiago-Calvo <u>et al</u>., 1963) others have failed to confirm this finding. An alternative source could be that

- 65 -

derived from $Ins(1,4,5)P_3$. Evidence for an active and specific Ins(1,4,5)P₃-3-kinase has been presented (Irvine <u>et al</u>., 1986). The same authors (Irvine et al., 1986) propose that formation of $Ins(1,3,4,5)P_4$ and subsequent degradation to $Ins(1,3,4)P_3$ (via a 5-dephosphorylation of Ins(1,3,4,5)P4, see Batty et al., 1985) may provide an alternative route of metabolism of $Ins(1,4,5)P_3$ rather than the acclaimed dephosphorylation route. Although it seems quite likely that $Ins(1,3,4)P_3$ and/or $Ins(1,3,4,5)P_4$ will turn out to have intracellular messenger functions of their own, this idea remains to be tested because so far the compounds have only been obtained in trace amounts. Concomitant with the formation of inositol phosphates is the production of 1,2-diacylglycerol, which remains in the plasma membrane. In most cells, 1,2-diacylglycerol is metabolised primarily to phosphatidic acid (PtdA) by diacylglycerol kinase. Carrier proteins exist which transfer the formed PtdA to the endoplasmic reticulum (or analogous membrane in platelets) where it interacts with CTP to form cytidine diphosphate diacylglycerol. The latter compound combines with inositol to form PtdIns which is then carried back to the plasma membrane by carrier proteins thus completing the cycle. Itis important to note that, regardless of the parent compound, 1,2-diacylglycerol and hence PtdA will be formed. As such PtdA formation is often used as an index of inositol phospholipid metabolism. 1,2-diacylglycerol may also be attacked sequentially by diacylglycerol lipase and monoacylglycerol lipase thus producing free arachidonate. Mauco et al. (1984) calculated that as much as 60% of 1,2-diacylglycerol may be

- 66 -

consumed in this fashion, the remaining 40% returning to the metabolism of inositol phospholipids, forming a common backbone to all inositol phospholipids. Subsequent studies (Mahadevappa and Holub, 1986) have proposed that the diacyl/monoacylglycerol lipase pathway is a minor source of released arachidonate.

In passing, a special mention for phospholipase C is apt since, as the wealth of information suggests, this is perhaps the major enzyme regulating agonist induced metabolism of inositol phospholipids. Various studies have demonstrated that the phospholipase C enzyme is capable of metabolising not only PtdIns (Mauco et al., 1979; Rittenhouse-Simmons, 1979; Stess and Lapetina, 1983) but also PtdIns (4)P (Rittenhouse, 1983; Graff et al., 1984) and PtdIns(4,5)P₂ (Agranoff et al., 1983; Rittenhouse, 1983). One difficulty that had to be resolved was how the inositol phospholipids, located in the inner leaflet of the plasma membrane, could be attacked by a cytosolic enzyme. One possibility is that upon combining with the receptor the agonists induce a conformational change in it which in turn perturbs the membrane sufficiently to make the inositol phospholipids accessible to the phosphodiesterase (Irvine et al., 1984). An alternative proposal is that the phosphodiesterase may also be controlled by coupling of receptors to the enzyme through a GTP-binding protein in an analogous manner to the adenylate cyclase system (Gomperts, 1983; Gilman, 1984). An early clue to this came with the demonstration that GTP reduced the affinity of noradrenaline for α_1 -receptors (Goodhardt <u>et al</u>., 1982), carbachol for muscarinic receptors (Florio and Sternweiss, 1985), and chemotactic peptides to neutrophil membranes (Koo et al.,

- 67 -

1983). More definitive evidence indicating an involvement of GTP-binding proteins in metabolism of inositol phospholipids has been obtained by observing the effects of introducing nonhydrolysable analogues of GTP into the cells. Cockcroft and Gomperts (1985) have shown that metabolically stable GTP analogues stimulate polyphosphoinositide metabolism in neutrophil plasma membranes. Using platelets that had been made permeable to small molecules, Haslam and Davidson (1984) demonstrated that GTP-analogous promote 1,2-diacylglycerol formation and the secretion of serotonin. Similar results have been obtained in homogenates and plasma membrane preparations from insect salivary glands (Litosh et al., 1985), hepatocytes (Wallace and Fain, 1985) and GH₃ pituitary cells (Lucas <u>et al.</u>, 1985). This data suggests that a novel GTP-binding protein may be involved in coupling receptors to phospholipase C. However the identity of this GTP-binding protein has not yet been ascertained. Numerous GTP-binding proteins probably exist, the best characterised being N_s and N_i from the adenylate cyclase system. Addition of pertussis toxin inhibits the activity of N_i by catalysing the ADP-ribosylation of the 41K α -subunit of this protein (Katada and Ui, 1982). Addition of pertussis toxin to mast cells (Nakamura and Ui, 1985), neutrophils (Verghese et al., 1985) and human leukemic HL60 cells (Brandt et al., 1985; Krause et al., 1985) inhibited agonist-induced cellular activation, including amongst, other things, metabolism of inositol phospholipids, suggesting that the GTP-binding protein may be N_i. However, addition of pertussis toxin to 3T3 fibroblasts (Murayama and Ui, 1985), pituitary GH₃ cells

- 68 -

(Schlegel <u>et al</u>., 1985) and hepatocytes (Pobin er <u>et al</u>., 1985) with concentrations that apparently cause a complete ADP-ribosylation of N_i and prevention of its inhibitory effects on adenylate cyclase, do not prevent agonist-induced increases of polyphosphoinositide breakdown or Ca²⁺ mobilization. Hence the exact identity of the GTP-binding protein coupled to phospholipase C remains elusive. However, the possibility of a GTP-binding protein that differs from N_i in its susceptability to ribosylation and inactivation by pertussis toxin in different cells, but which uniquely couples to phospholipase C in all cells may hold true.

Finally, the dependency of phospholipase C on Ca^{2+} has been the subject of much discussion. Various studies (Simon <u>et al.</u>, 1984; Chap <u>et al.</u>, 1985) have demonstrated, using Quin 2 loaded platelets, the ability of thrombin to induce platelet activation at resting levels of cytoplasmic Ca^{2+} . In a separate study Wilson and colleagues (1984) demonstrated, using purified phospholipase C reconstituted along with radiolabelled inositol lipid substrates into phospholipid vesicles, that whereas Ca^{2+} stimulated the breakdown of all three inositol phospholipids, only the polyphosphoinositides were hydrolysed in the presence of EGTA. Nevertheless, one must be cautious when stating the dependency of an enzyme on the presence of Ca^{2+} since it may simply require the presence of Ca^{2+} for activation (see Michell <u>et al.</u>, 1981).

- 69 -

1.3.1.6 <u>Metabolism of inositol phospholipids - their role in</u> stimulus-response coupling

The major unanswered question in cells which use Ca^{2+} ions to elicit their response is how the Ca^{2+} ions are made available to the cell. Unlike the adenylate cyclase system, where receptors are coupled in either a stimulatory or inhibitory manner thus regulating the intracellular second messenger cyclic AMP, the mechanism regulating Ca^{2+} availability is unknown or at least speculative.

Where cells depend upon the extracellular medium for their source of Ca^{2+} ions the immediate problem is to overcome the penetration of the plasma membrane. This problem can be overcome by the opening of Ca^{2+} "channels" in the plasma membrane allowing Ca^{2+} ions to flood in down its concentration gradient. There may be as many as three such channels (Reuter, 1983). The two major ones are the receptor-operated channels or 'ROC's and voltage-operated channels or 'VOC's. Receptor mediated cellular activation in the absence of a change in membrane potential, as is the case with platelets (MacIntyre and Rink, 1982), suggests that platelets rely on ROCs for gating extracellular Ca^{2+} . As outlined in Section 1.3.1.3 the existence of a specific Ca^{2+} channel in platelets has been proposed. Whether these channels represent the ROC's is unknown.

Mobilization of intracellular Ca²⁺ as a consequence of cell surface receptor occupation demands the presence of intracellular mediators capable of releasing stored Ca²⁺. These mediators may or may not be involved in the influx of extra cellular Ca²⁺ through the plasma membrane.

Over the past years, the search for these mediators has centred primarily on the stimulated metabolism of inositol phospholipids. These are rather minor components of the plasma membrane, representing 5-10% of the total phospholipids, but are the most metabolically active. The one present in the largest amounts is phosphatidyl inositol (PtdIns) while the phosphorylated derivatives, phosphatidyl inositol-4-phosphate (PtdIns(4)P) and phosphatidylinositol 4,5-bisphosphate (PtdIns $(4,5)P_2$) each contain about 1% of the total inositol lipid (see Figure 2c). Stimulated inositol phospholipid metabolism was discovered by Hokin and Hokin (1953), who demonstrated that the incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ into phospholipids in pancreas was stimulated by acetylcholine. Metabolism of the polyphosphoinosotides was first demonstrated by Durell et al. (1969) who suggested that this stimulated metabolism may be a part of receptor function. The first indication of an involvement between agonist-stimulated metabolism of inositol phospholipids and elevation of cytoplasmic free calcium was cited by Michell in 1975. He demonstrated that occupation of receptors which elicited the elevation of cytoplasmic free calcium also brought about the metabolism of PtdIns giving rise to the phrase the "PtdIns response". Although the hypothesis was generally accepted there were some points of contention such as the apparent calcium-dependency of the "PtdIns response" in neutrophils (Cockcroft <u>et</u> <u>al</u>., 1980a,b; Cockcroft, 1981; see also Hawthorne, 1982; Michell, 1982 for critical appraisal). Nevertheless, substantial - be it circumstantial - evidence now exists leading to the widespread acknowledgement that changes in cytoplasmic free calcium concentration and inositol

- 71 -



Ptd Ins 4.5P2

Figure 2c: Structures of the inositol phospholipids.

Phosphatidyl inositol (PtdIns); Phosphatidyl inositol-4-phosphate (PtdIns(4)P); Phosphatidyl inositol-4,5-bisphosphate (PtdIns(4,5)P). At position two on the diglyceride side chain Arachidonic Acid is most common. At position 1, stearic acid is most common.

phospholipid metabolism are closely linked. Some of the earliest data pertaining to this association included the work of Fain and Berridge (Fain and Berridge, 1979; 1979a; Berridge and Fain, 1979). Using flowfly salivary glands they demonstrated that upon prolonged stimulation with 5HT, the tissue was unable to transport Ca²⁺ (Berridge and Fain, 1979). This effect was associated with an inability to resynthesize PtdIns (due to an apparent Ca²⁺-dependent block in the synthetic pathway between PtdIns and PtdA) (Berridge and Fain, 1979). Addition of inositol to the preparation restored the ability of 5HT to elicit Ca²⁺-transport (Berridge and Fain, 1979; Fain and Berridge, 1979a), thus reinforcing the proposed link between inositol phospholipid metabolism and Ca²⁺ mobilization. 0ther evidence includes the many coincidences between calciummobilizing agonists and stimulated inositolphospholipid turnover (Michell and Kirk, 1981) and equally important numerous negative controls (that is, receptors which do not mobilize calcium do not stimulate the metabolism of inositol phospholipids). Many studies have demonstrated a strong correlation between occupation of specific receptors, metabolism of inositol phospholipids and elevation of cytoplasmic free calcium (Michell et al., 1976; Kirk et al., 1981; Evans et al., 1985) suggesting that the phenomenon is a receptor operated event and not a non-specific action at the plasma membrane.

Recent evidence suggests however that it is not the metabolism of PtdIns that is the initial event occurring after agonist-receptor interaction but rather the metabolism of the

- 73 -

phosphorylated derivatives namely PtdIns(4)P and PtdIns(4,5)P2. In rat hepatocytes, Michell et al. (1981) demonstrated that addition of vasopressin resulted in a decrease in $PtdIns(4,5)P_2$ which preceded any changes in PtdIns and was largely independent of extracellular calcium. Furthermore, the demonstration that the formation of $Ins(1,4,5)P_3$, the water soluble product of $PtdIns(4,5)P_2$ metabolism, precedes that of Ins(1)P (the water soluble product of PtdIns metabolism) in insect salivary glands (Berridge, 1983; Berridge et al., 1984a) and GH3 pituitary tumor cells (Rebecchi and Gershengorn, 1983; Martin, 1983) provides additional evidence that $PtdIns(4,5)P_2$ is in fact the primary substrate in agonist stimulated metabolism of inositol phospholipids. Perhaps the most tantalizing evidence linking Ca²⁺ fluxes and metabolism of inositol phospholipids, especially PtdIns(4,5)P2 originates from studies where levels of polyphosphoinositides have been depleted by lowering levels of cellular ATP. Under these conditions, agonist addition is unable to exert effects on intracellular Ca²⁺ levels (Poggioli et al., 1983). However, an alternative interpretation of this particular data could be that by lowering intracellular levels of ATP one is interfering with an ATP-dependent mechanism regulating Ca²⁺ availability. This ATP-dependent mechanism may or may not be dependent on the absolute levels of the polyphosphoinositides. In simplistic terms, depletion of cellular ATP levels may merely interfere with energy dependent processes leading to an inability of the cell to respond to external stimuli (Akkermann et al., 1983).

Nevertheless, there are now many studies supporting the notion that the metabolism of $PtdIns(4,5)P_2$ is a common response in a variety of different cell types using primarily Ca^{2+} ion to elicit their responses (see Berridge, 1984 for comprehensive list).

1.3.1.7 Agonist-induced metabolism of inositol phospholipids

The first evidence for the metabolism of inositol phospholipids in platelets was reported as early as the 1970's by Loyd and coworkers. They demonstrated the incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ into PtdA, PtdIns, PtdIns(4)P and PtdIns(4,5)P₂ in rabbit platelets exposed to ADP (Lloyd et al., 1972, 1973) and collagen and Thrombin (Lloyd and Mustard, 1974). Later a rapid but transient Thrombin-induced increase in 1,2-diacylglycerol and decrease in PtdIns was reported (Rittenhouse-Simmons, 1979; Bell and Majgrus, 1980). Metabolism of PtdIns and a concomitant formation of PtdA was demonstrated in human platelets by Broekman et al. (1980) . As described previously (Section 1.3.1.6) the initial event in agonist-induced metabolism of inositol phospholipids is believed to be the metabolism of $PtdIns(4,5)P_2$. Consequently many studies have demonstrated a rapid and transient agonist-induced decreases in labelled polyphosphoinositides. In human platelets using Thrombin, a decrease in labelled $PtdIns(4,5)P_{2}$ followed by an increase has been demonstrated (Imai <u>et</u> <u>al</u>., 1983; Agranoff, 1983).

Measuring the mass of inositol phospholipids, Vickers <u>et al</u>. (1982) in rabbit platelets demonstrated a decrease in both mass and radiolabelled phosphorous of PtdIns(4,5)P₂ in platelets activated by ADP. Using Thrombin, in human platelets,

Broekman (1984) similarly demonstrated a decrease in mass of $PtdIns(4,5)P_2$ followed by an increase in this inositol As would be expected a concomitant production of lipid. inositol phosphates can be demonstrated in platelets labelled either with $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ or $\begin{bmatrix} 3 H \end{bmatrix}$ inositol. Agranoff <u>et al</u>. (1983) demonstrated a very rapid increase in 3^{2} Plabelled Ins(1,4,5)P within seconds after thrombin stimulation. Thrombin-induced formation of $[^{3}H]$ labelled $Ins(1,4)P_{2}$ and $Ins(1,4,5)P_{3}$ have been detected before significant accumulation of InsIP in human platelets (Watson et al., 1984) and rabbit platelets (Vickers et al., 1984). The above citations support the concept that metabolism of $PtdIns(4,5)P_2$ is amongst the initial events that follow receptor occupation and inherent in this theory (as expounded by Michell et al., 1981) disappearance of PtdIns occurs as a result of compensatory resynthesis of depleted polyphosphoinositides.

However recent data suggests that the metabolism of inositol phospholipids, at least in platelets, may not be as straightforward as first seems. Wilson <u>et al</u>. (1985) demonstrated using non-equilibrium labelling techniques, that the flux or rate of conversion of PtdIns to PtdIns(4P) and PtdIns(4,5)P₂ in human platelets stimulated with thrombin was not fast enough to explain the decrease in PtdIns. They proposed that all three (i.e. PtdIns, PtdIns(4)P and PtdIns(4,5)P₂) be metabolised. However, this proposal is inconsistent with the fact that no significant amounts of $\begin{bmatrix} 3 \\ H \end{bmatrix}$ InsIP are produced in platelets labelled with $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol soon after agonist addition (Seiss, 1985).

1.3.1.8 <u>Metabolism of inositol phospholipids and generation of</u> <u>effectors or second messengers</u>

While there appears little doubt that the metabolism of inositol phospholipids is a recurrent feature of receptors acting predominantly through calcium, the exact nature of the mechanisms governing cellular reactivity in these cells is just beginning to unfold. It now appears that roles for most of the intermediaries in the inositol phospholipid metabolic cycle exist.

1.3.1.8.1 Role of PtdIns(4,5)P2, PtdIns and PtdA

 $PtdIns(4,5)P_2$ besides being a substrate for phospholipase C, may also be a potential source of Ca²⁺ required for cellular reactivity. PtdIns(4,5)P₂ is known to bind Ca^{2+} with high affinity (Dawson and Hauser, 1970) and it is possible that upon its metabolism it may release its source of Ca^{2+} (Billah and Lapetina, 1982). The original hypothesis put forward by Michell (1975) envisaged that metabolism of PtdIns in some way resulted in the opening of a calcium channel in the plasma membrane thus allowing an influx of Ca²⁺. Subsequent theories included an ionophoric effect for PtdA (Putney et al., 1980). Lyso PtdA (produced from PtdA by the action of phospholipase A_2) was also proposed as an ionophore in platelets (Lapetina et al., However the discovery by Holmes and Yoss (1983) that 1981a). neither PtdA nor lysoPtdA could translocate Ca²⁺ across the membranes of purified phosphatidyl choline vesicles suggested they might not be important regulators of calcium flux. The possible role of inositol phospholipid metabolism in mediating

 Ca^{2+} influx is, at present, unknown, however recent evidence suggests that $Ins(1,3,4)P_3$ or indeed $Ins(1,3,45)P_4$ may play a role in Ca^{2+} influx through the plasma membrane (see Section 1.3.1.8.2).

1.3.1.8.2 Role of inositol phosphates

While there still remains some doubt as to the nature of the relationship between the metabolism of inositol phospholipids and the influx of Ca^{2+} , there appears little doubt, from the now copious literature, that the link between such metabolism and mobilization of Ca^{2+} from intracellular stores is $Ins(1,4,5)P_3$ - the water soluble product from metabolism of PtdIns(4,5)P₂.

The first suggestion that this may be the case was put forward by Berridge and colleagues (1983, 1984). They observed, in insect salivary glands, that upon addition of serotonin, accumulation of $Ins(1,4,5)P_3$ occurred with no apparent lag, however the physiological Ca^{2+} -dependent response was delayed by approximately 7s. They concluded that $Ins(1,4,5)P_3$ may act to mobilize intracellular calcium. Increases in $Ins(1,4,5)P_3$ following cell stimulation also precede physiological responses in liver (Thomas <u>et al</u>., 1984) and GH_3 pituitary tumor cells (Drummond <u>et al</u>., 1985) thus confirming their proposal. The finding that $Ins(1,4,5)P_3$ was metabolised equally quickly suggested that it may in fact be the effector or second messenger which could mobilize intracellular Ca^{2+} .

In 1983 Streb and coworkers demonstrated the ability of exogenously added $Ins(1,4,5)P_3$ to mobilize intracellular calcium in a preparation of leaky rat pancreatic acinar cells. Similar

- 78 -

observations were later found in a variety of cell preparations including liver (Burgess <u>et al</u>., 1984; Joseph <u>et al</u>., 1984) GH3 pituitary cells (Gershengorn et al., 1984) Swiss 3T3 cells (Berridge et al., 1984) as well as human platelets (O'Rourke et al., 1985; Authi and Crawford, 1985; Adunyah and Dean, 1985). Perhaps the most definitive work demonstrating a second messenger role for $Ins(1,4,5)P_3$ is the observation that direct introduction of this water-soluble metabolite, by microinjection, into sea urchin eggs caused exocytosis of cortical granules and the appearance of a fertilization envelope (Whitaker and Irvine, 1984). Additionally, introduction into intact photoreceptor cells of Limulus evoked a voltage response similar to that produced by light (Fein et al., 1984; Brown et al., 1984). The store from which $Ins(1,4,5)P_3$ releases Ca^{2+} is most likely to be the endoplasmic reticulum (or dense tubular system in platelets) since inhibition of mitochondrial function by removing metabolic substrates or by adding inhibitors has no effect on the ability of $Ins(1,4,5)P_3$ to release calcium (Streb <u>et al</u>., 1983; Joseph et al., 1984; Gershengorn et al., 1984). This conclusion is supported by cell fractionation experiments, which demonstrate that $Ins(1,4,5)P_3$ can release calcium from microsomes but not from isolated mitochondria (Prentki et al., 1984).

The mechanism by which $Ins(1,4,5)P_3$ releases intracellular Ca^{2+} is thought to involve its interaction with specific sites or receptors on the endoplasmic reticulum. Spat <u>et al</u>. (1986) have demonstrated that $Ins(1,4,5)P_3$ binds to a specific and saturable site in permeabilized guinea pig hepatocytes and

rabbit neutrophils. The properties of this binding site were consistent with it being the physiological receptor for $Ins(1,4,5)P_3$. Recent studies by Dawson <u>et al</u>. (1986) have demonstrated that a protein, phosphorylated by GTP, has to bind to microsomal membranes before $Ins(1,4,5)P_3$ can stimulate Ca^{2+} release. Whether these two mechanisms are mutually exclusive remains to be seen. The exact concentrations of $Ins(1,4,5)P_3$ that can be attained after cell stimulation is unknown. However, several estimates suggest that micromolar levels, which are within the levels active on permeabilized cells, can be attained (see Berridge and Irvine, 1984). Termination of calcium mobilization induced by $Ins(1,4,5)P_3$ apparently involves the action of a phosphomonoesterase on $Ins(1,4,5)P_3$ (Authi and Crawford, 1985; Storey et al., 1986). Recently a new dimension to the Ca^{2+} -mobilizing role of $Ins(1,4,5)P_3$ has been introduced by Putney (1986). He proposed that besides mobilizing Ca²⁺ from intracellular stores, $Ins(1,4,5)P_3$ may also provide a mechanism for Ca²⁺ influx into the cell. Based on a capacitance theory, the generation of $Ins(1,4,5)P_3$ facilitates both Ca^{2+} mobilization from intracellular stores and Ca²⁺ influx through the plasma membrane. Whether this theory is upheld remains to be seen. Preliminary experiments performed in rat liver plasma membrane vesicles (Hansen et al., 1986) ascribe a similar role to the higher phosphorylated derivative $Ins(1,3,4,5)P_{\mu}$. Again, this observation awaits further confirmation.

In addition to mobilizing Ca^{2+} from intracellular (and possibly extracellular) sites $Ins(1,4,5)P_3$ has been demonstrated to induce platelet activation (Brass and Joseph, 1985) possibly vis arachidonic acid metabolites (Authi <u>et al.</u>, 1986; Watson <u>et al.</u>, 1986). $Ins(1,4,5)P_3$ has also been demonstrated to induce phosphorylation of 40K and 20K proteins (Watson <u>et al.</u>, 1986; Lapetina <u>et al.</u>, 1984). Whether or not these effects are mediated directly by $Ins(1,4,5)P_3$ or via released Ca^{2+} and the biochemical/physiological relevance of such effects is not known.

In the majority of cases where cells have been demonstrated to utilize Ca^{2+} as part of their stimulus-response coupling mechanism, it appears that $Ins(1,4,5)P_3$ is the second messenger releasing intracellular Ca²⁺. Possible exceptions could be in smooth and skeletal muscle where, although $Ins(1,4,5)P_3$ has been demonstrated to induce release of Ca²⁺ from intracellular stores (Volpe et al., 1985; Volpe et al., 1986; Yamamoto and Breemen, 1985), its involvement here is by no means certain. Other water soluble products of inositol phospholipid metabolism such as inositol 1:2-cyclic 4,5-triphosphate also appear to have calcium mobilizing activities (Wilson et al., 1985). Despite difficulty in isolating this compound, due to its rapid metabolism under the acidic conditions used to extract inositol phosphates, it appears that this compound is as likely as its non-cyclic derivative to subserve a second messenger function in stimulated cells (Wilson et al., 1985).

1.3.1.8.3 Role of 1,2-diacylglycerol

As indicated in Section 1.3.1.3 the addition of the calcium ionophore A23187 to platelets results in their activation, thus indicating a role for Ca²⁺ ions in platelet reactivity.

- 81 -

However, the studies by Rink <u>et al</u>. (1981) who demonstrated that Thrombin could evoke platelet shape change and secretion at resting levels of cytoplasmic free calcium and those of Knight <u>et al</u>. (1982) who demonstrated that different concentration of Thrombin were required to evoke the same degree of secretion from dense granules and lysosomal granules whereas similar concentrations of Ca^{2+} could elicit both processes (see Section 1.3.1.3) suggested that processes, other than an elevation of cytoplasmic free calcium, must operate in platelets resulting in their activation. This additional process is believed to involve the interaction of 1,2-diacylglycerol and Ca^{2+} dependent, phospholipid-dependent protein kinase C.

Protein kinase C was first isolated by Inoue et al. (1977) as a proteolytically-activated protein kinase. Subsequent studies demonstrated it to be a Ca-dependent, phospholipiddependent enzyme (Takai et al., 1979). Experiments carried out by Nishizuka and colleagues later demonstrated that the lipid 1,2-diacylglycerol could greatly increase the affinity of protein kinase C for Ca²⁺ thereby activating it (Takai <u>et al</u>., 1979a; Kishimoto et al., 1980; Nishizuka, 1980). Very rapid and transient increases in 1,2-diacylglycerol are detected soon after activation of a variety of cell types, including platelets, which previously were assumed to utilize Ca^{2+} alone as a second messenger (Rittenhouse-Simmons, 1979). It is believed that 1,2-diacylglycerol produced in activated cells plays a physiological role in activating protein kinase C. This belief originated from the above observation that 1,2diacylglycerol, a product of agonist-induced inositol phospholipid metabolism, could increase the affinity of protein kinase

- 82 -

for Ca²⁺ thereby activating it (Kawahara <u>et al</u>., 1980; Takai et al., 1982; Nishizuka, 1984). Such observations were enhanced by the work of Castagna et al. (1982) and Kaibuchi et al. (1982). Castagna et al. (1982) demonstrated that a number of phorbol esters could stimulate protein kinase C in a manner analogous to 1,2-diacylglycerol. Indeed it can be demonstrated that phorbol esters and 1,2-diacylglycerol can compete for binding sites on protein kinase C (Sharkey et al., 1984), and as such protein kinase C has been proposed as the receptor for phorbolesters (ergo 1,2-diacylglycerol) (Kikkawa <u>et</u> <u>al</u>., 1983; Nishizuka, 1984). Only biologically active 4β -phorbol esters such as TPA, phorbol dibutyrate and phorbol dibenzoate are able to activate protein kinase C, while others such as 4α phorbol didecanoate - which is not biologically active - fail to affect protein kinase C (Couturier et al., 1984). Kaibuchi et al. (1982) demonstrated that synthetic mimetics of 1,2diacylglycerol such as 1-oleoy1-2-acety1-sn3-glycerol (OAG) can mimic the effects of naturally produced 1,2-diacylglycerol. Again this effect proved highly specific since monoacylglycerol or triacylglycerol or fatty acids fail to activate the enzyme (Castagna et al., 1985).

That activation of protein kinase C can occur only when biologically active phorbol esters and mimetics of 1,2diacylglycerol are used suggests that activation of protein kinase C is a highly specific event. Furthermore, addition of phorbol esters to samples of platelets does not result in either metabolism of inositol phospholipids or formation of endogenous 1,2-diacylglycerol (Nishizuka, 1984; Castagna <u>et al</u>.,

- 83 -

1985; Castagna et al., 1982). However, studies have been performed which demonstrate that the effects of phorbol esters and synthetic analogues of 1,2-diacylglycerol are not directly comparable (Ashby <u>et al.</u>, 1985; Kiss and Luo, 1986). One further dimension to the enzyme protein kinase C, is that upon agonist-induced elevation of 1,2-diacylglycerol, the enzyme is translocated from the cytosol to the plasma membrane (Drust and Martin, 1985), a phenomenon possibly regulated by cytoplasmic levels of calcium (Wolf <u>et al.</u>, 1985; May <u>et al.</u>, 1985).

The addition of phorbol esters or OAG to human platelets results in secretion and aggregation in the absence of changes in basal cytoplasmic free calcium levels (Rink et al., 1983; at least as reported by Quin 2). This independence from changes in cytosolic free calcium levels has recently been challenged by Ware et al. (1985) who suggest that small discrete changes in cytosolic free calcium evoked by agents like OAG may not be detected using Quin 2, which is diffused throughout the cell, but may be detected using aqueorin which tends to report localised changes in cytoplasmic calcium concentration. Whether or not phorbol esters or OAG do in fact elicit the elevation of cytoplasmic free calcium, the biological (cellular) response induced by these agents is qualitatively different compared to that evoked by a receptor-directed agonist; being delayed and somewhat slowed (Rink et al., 1983; Rink and Hallam, 1984). Subsequent studies demonstrated that low concentrations of calcium ionophore or OAG which alone could induce very little secretion can produce a full secretory

- 84 -

response when added in combination with each other (Kaibuchi et al., 1982; 1983; Rink et al., 1983; Rink and Hallam, 1984). The synergistic effect of the calcium-dependent pathway (mediated by elevation of Ca^{2+}) and the calcium-independent pathway (mediated by 1,2-diacylglycerol activated protein kinase C) represents a bifurcating signal pathway. Such a system can provide the versatility necessary to introduce subtle variations in the control mechanism and is observed not only in platelets, but in a variety of other cells including lymphocytes (Mastro and Smith, 1983) and adrenal glomerulosa cells (Kojima <u>et al</u>., 1983) to name but two. (See also Nishizuka, 1984).

The degree to which individual agonists utilize each pathway is unknown. However, in human platelets, some way towards understanding the mode of action of physiological agonists such as collagen, Thrombin and ADP was carried out by Rink and Hallam (1984). They envisaged collagen acting primarily through the 1,2-diacylglycerol-protein kinase C pathway, ADP through the Ca²⁺-dependent pathway and Thrombin acting through both, thus possibly accounting for the fact that Thrombin is one of the most efficacious platelet However such observations - at least with agonists known. respect to collagen and ADP - diverge from the widely held belief that an integral aspect of agonist-receptor interaction in a variety of different cell types is in fact the synergistic action of $Ins(1,4,5)P_3$ - ergo mobilization of Ca^{2+} - and 1,2-diacylglycerol - ergo activation of protein kinase C.

- 85 -

Collagen has been demonstrated to elicit the formation of 1,2-diacylglycerol at basal levels of Ca²⁺ (Rink et al., 1983). One possible explanation for such an effect, as proposed by Rink and Sanchez(1984) could be that collagen preferentially hydrolyses a compound other than $PtdIns(4,5)P_2$, possibly PtdIns, such that no formation of $Ins(1,4,5)P_3$ occurs but still allows formation of 1,2-diacylglycerol. However, Watson et al. (1985a) since demonstrated the formation of $Ins(1,4,5)P_3$ in collagen treated (human) platelets. Nevertheless, these authors could find no detectable mobilization of Ca²⁺ as monitored by Quin 2. The possibility remains that as an indicator of small, localised transients of Ca²⁺, that Quin 2 may not be the most sensitive probe to use (see Section 2.2.9.2). Hence whether or not collagen can induce elevation of cytosolic Ca^{2+} remains to be ascertained. Using ADP as the agonist, a number of workers have demonstrated that this agonist elicits an elevation in cytosolic Ca²⁺ without the concomitant metabolism of inositol phospholipids - ergo $Ins(1,4,5)P_3$ and 1,2diacylglycerol formation (MacMillan et al., 1984; MacIntyre et al., 1985c; Fisher et al., 1985). Recent studies demonstrated that the response to ADP, in human platelets, was not voltage dependent (Hallam and Rink, 1985). Hence the mechanism whereby ADP elevates cytosolic Ca²⁺ remains an enigma.

Addition of Thrombin (Kawahara <u>et al</u>., 1980), collagen (Sano <u>et al</u>., 1983) or platelet activating factor (Ieyasau et al., 1982) to platelets results in the phosphorylation of two proteins of molecular mass 20K and 40K. In intact platelets
one of the major substrates for protein kinase C appears to be the 40K protein (Ieyasau et al., 1982; Sano et al., 1983). Phosphorylation of the 40K protein has been associated with platelet secretion (Ieysau et al., 1982; Sano et al., 1983; see also Nishizuka, 1984). However the exact relationship between phosphorylation and secretion is not clear since an increase in 40-47K protein phosphorylation can be shown to occur in the absence of secretion if permeabilized platelets are challenged with 1,2-diacylglycerol in the presence of 0.01 µM Ca²⁺ (Knight et al., 1984; Haslam and Davidson, 1984). The effects of protein kinase C has been demonstrated to be suppressed by the addition of trifluoperazine, chlorpromazine (Sanchez et al., 1983) or 1-(5-isoquinoline-sulfony1)-2methylpiperazine (H7) (Inagaki <u>et al</u>., 1984). The use of such compounds, especially the latter, will be useful in discerning whether or not activation of protein kinase C by 1,2-diacylglycerol does fulfil any physiological role in cellular activation.

Besides exhibiting synergism with Ca^{2+} ions, activation of protein kinase C, by 1,2-diacylglycerol, is now thought to be involved in inhibitory feedback mechanisms in a number of different cell types. Lagast <u>et al</u>. (1984) demonstrated, in neutrophils, that phorbol esters could suppress agonistinduced elevation of cytoplasmic free calcium and attributed it to an activation of a Ca^{2+} -extrusion process. Prior addition of phorbol esters to human platelets was also shown to suppress the increase in cytoplasmic free calcium induced

- 87 -

by platelet agonists such as Thrombin (MacIntyre et al., 1985d; Zavoico et al., 1985). However, since basal cytoplasmic concentrations of Ca²⁺ were not affected, this effect of phorbol esters in human platelets can not be attributed to the activation of an extrusion process. Addition of phorbol esters at the peak of agonist-induced elevation of cytoplasmic free calcium has been demonstrated to accelerate the decline in Quin 2 fluorescence (MacIntyre et al., 1985a; MacIntyre et al., 1985b) and a number of other cells including GH3 pituitary tumor cells (Drummond, 1985), rat basophilic leukemic cells (Sagi-Eisenberg <u>et al</u>., 1985) and PC 12 cells (Vincintini <u>et al</u>., 1985). Addition of agonist to a sample of platelets (MacIntyre et al., 1985a) or GH₃ pituitary tumor cells (Drummond, 1985) preincubated with Quin 2 often exhibit an increasing rate of decline as the agonist concentration is increased. This suggests that the cells are limiting their response to high concentrations of agonists by activating an intracellular negative feedback mechanism. In view of the ability of phorbol esters to alter or reduce agonist-induced changes in cytoplasmic free calcium consequently, it was postulated (Drummond and MacIntyre, 1985), that such a regulatory mechanism may operate vis activation of protein kinase C.

Not only are agonist-induced changes in cytoplasmic free calcium attenuated or inhibited by addition of phorbol esters, but the metabolism of inositol phospholipids also seems to be affected by these compounds. In platelets, thrombin, vasopressin, and PAF induced metabolism of inositol phospholipids, monitored as $\begin{bmatrix} 3^2P \end{bmatrix}$ -PtdA formation, was impaired by phorbol esters

(MacIntyre <u>et al</u>., 1985d; Zavoico <u>et al</u>., 1985). Such effects on inositol phospholipid metabolism are observed in a number of other cell types including neutrophils (Naccache et al., 1985) PC12 cells (Vinientini et al., 1985) and GH₃ Pituitary tumor cells (Drummond, 1985). In their study MacIntyre et al. (1985d) suggested that the mechanism by which activated protein kinase C results in the inhibition of inositol phospholipid metabolism could be attributed to an inhibiton of phospholipase C. However, the design of their experiments could not preclude alternative actions e.g. on 1,2-diacylglycerol kinase or PtdIns(4)P phosphomonoesterase. Subsequent studies by Rittenhouse and Sasson (1985) demonstrated that phorbol esters attenuated Thrombin-induced formation of $Ins(1,4,5)P_3$ - an effect consistent with inhibition of phospholipase C. Given the proposed link between $Ins(1,4,5)P_3$ and Ca^{2+} mobilization, inhibition of inositol phospholipid metabolism by activated protein kinase C may represent a mechanism by which agonist-induced changes in cytoplasmic free Ca²⁺ are attenuated or inhibited.

The possibility that an intermediary in agonist-induced metabolism of inositol phospholipids e.g. 1,2-daiacyglycerol could both facilitate and inhibit cellular reactivity led to the concept that protein kinase C may subserve a bidirectional role in cellular activation (Drummond and MacIntyre, 1985). One interesting divergence from this concept is the observation made by Krishnamurthi <u>et al</u>. (1986) who demonstrated that OAG could potentiate the actions of Thrombin in human platelets, but observed no inhibitory effect with this compound. OAG is a synthetic analogue of 1,2-diacylglycerol and, like its

- 89 -

endogenous counterpart, is rapidly metabolised. Consequently, studies utilizing this compound may prove more physiologically relevant than those using phorbol esters. The ability of synthetic mimetics of 1,2-diacylglycerol and phorbol esters to stimulate the formation of polyphosphoinositides (de Chaffoy de Courcelles et al., 1984a, b; Halenda and Feinstein, 1984; Taylor et al., 1984), apparently by increasing the activity of PtdIns and PtdIns(4)P kinase by an unknown mechanism, represents another divergence from the proposed bidirectional control of cellular reactivity by protein kinase C. $\mathbf{B}\mathbf{y}$ stimulating the formation of polyphosphoinositides - rather than inhibiting cellular reactivity as inherent in the bidirectional control proposal - this effect of protein kinase C could conceivably act as a positive feedback mechanism by generating and/or maintaining a supply of inositol phospholipids ergo generation of second messengers. However no convincable evidence has been presented to suggest that increased levels of polyphosphoinositides have any effect on the levels of second messengers. Additionally, the ability to stimulate formation of polyphosphoinositides has been demonstrated only in platelets (de Chaffoy de Courcelles, 1984a, b; Halenda and Feinstein, 1984) and thymocytes (Taylor et al., 1984).

Other effects of 1,2-diacylglycerol, presumably acting through protein kinase C, include modulation of adenylate cyclase activity and receptor regulation. In platelets, using phorbol ester-activated protein kinase C, both stimulation and inhibition of cyclic AMP formation has been demonstrated.

- 90 -

The former may result from phosphorylation of the N_i regulatory unit (Katada <u>et al.</u>, 1985; Watanabe <u>et al.</u>, 1985) and latter from phosphorylation of receptors that mediate an increase in cyclic AMP (Bushfield et al., 1986).

Modulation of receptors by phorbol ester-activated protein kinase C has also been demonstrated. In some cells agonist binding is inhibited. This may be a direct consequence of phosphorylation of the receptors by protein kinase C since receptors for somatomedin, insulin, EGF, transferin and histone H_1 have been shown to be substrates for the enzyme (Anderson et al., 1985). Internalisation of receptors for EGF and transferin (Beguinot et al., 1985) has also been demonstrated. Protein kinase C activation may be involved in receptor internalisation or desensitization as a general phenomenon for termination of agonist-mediated transduction processes (Macara, 1985).

Because the biological effects of activated protein kinase C have been evaluated mainly using phorbol esters or synthetic analogues of 1,2-diacylglycerol, which may not act in a similar fashion to endogenously produced 1,2-diacylglycerol and the doubt as to whether sufficient 1,2-diacylglycerol is produced in cells with physiological agonists to fully activate protein kinase C, much more investigation is required before a complete understanding of the role of protein kinase C in cellular activation is attained.

- 91 -

1.3.1.9 <u>Inositol phospholipid metabolism and generation of</u> arachidonic acid and cyclic GMP

In addition to metabolising inositol phospholipids and elevating cytoplasmic free calcium, receptors which act predominantly through calcium generate significant amounts of arachidonic acid and elevate cytoplasmic levels of cyclic GMP (Berridge, 1981).

Inositol phospholipids are enriched in the Sn-2 position with arachidonic acid, but oleate or linoleate can also be detected, albeit in significantly lower amounts. During agonist-induced metabolism of inositol phospholipids there appears a preferential degradation of molecules such that the 1,2-diacylglycerol and subsequently the PtdA, produced by the receptor mechanism predominantly carriers of arachidonic acid in the Sn-2 position (Bell et al., 1979; Mahadevappa and Holub, 1983). The release of arachidonic acid from inositol phospholipids and its subsequent metabolism to icosanoids has been outlined (Section 1.1.3.4). Released arachidonic acid and/or its metabolites has been demonstrated to activate guanylate cyclase (Graff et al., 1978; Craven and DeRubert, 1980; Gruetter and Ignarro, 1979). The role of cyclic GMP in cellular reactivity, unlike the role of cyclic AMP is widely contested. None more so in platelets. Addition of membrane-permeant derivatives of cyclic GMP such as 8-bromocGMP results in the inhibition of thrombin-induced inositol phospholipid metabolism, elevation of cytoplasmic free calcium, and secretion of serotonin (Nakashima et al., 1986). Addition

- 92 -

of sodium nitroprusside, which elevates levels of cyclic GMP selectively, at the peak of the Quin 2 fluorescence signal induced by platelet activating factor in human platelets resulted in an accelerated decrease in fluorescence signal (MacIntyre <u>et al</u>., 1985a) suggesting that cyclic GMP may play a role in calcium mobilization. Certain platelet stimulatory agonists have been reported to cause cyclic GMP formation, this occurs relatively late after agonist addition and may represent a mechanism by which platelets can regulate their reactivity (Haslam <u>et al</u>., 1980). However, data exists which demonstrates that an increase in cyclic GMP is ineffective (Weiss <u>et al</u>., 1978) or that its addition may actually enhance platelet reactivity (Chiang <u>et al</u>., 1976).

1. -

· "你们,你们的你们都是我的问题?"他说道:"你们我们就是你们的我们的?"

an en la sel de la perside en les en la seconda de la s

and a constant section of the second states in the second states in the

n an Alexandra (Alexandra) Nashin ya shekara ya shina ya Nashina ya shina ya s

2. MATERIALS AND METHODS

2.1 Materials

The following materials from the indicated sources were used in the undertaking of this project. Unless otherwise stated, all drugs were dissolved in distilled water to give stock solutions in the mM concentration range with subsequent dilutions being carried out in 0.9% saline.

2.1.1 Drugs and reagents

Bovine thrombin, adenosine, adenosine 3'-monophosphate acid (AMP); Adenosine 5'-diphosphate (ADP); Adenosine 5'triphosphate (ATP); Prostaglandin E₁ (PGE₁) and Digitonin were purchased from Sigma (U.K.).

Hirudin; Adrenaline-tartrate and Adenosin-5'-O-[3thiotriphosphate] (ATP(S) were purchased from Serva, Heidelberg; Koch-Light Laboratories (U.K.) and Boehringer Corporation (London) respectively. Both prostaglandin I_2 (PGI₂) and 9,11-epoxymethane prostaglandin H_2 (U44069) were donated by Dr J.E. Pike (Upjohn, U.S.A.). PGI₂ was dissolved in Tris-HC1 buffer, pH 9.2. Other prostaglandins were dissolved initially in ethanol and diluted to 3.3 mM stock by addition of 9 vol. Na₂CO₃(3 mM) with subsequent dilutions in saline. Quin-2acetoxy methyl ester (Quin 2 AM), purchased from Lancaster synthesis (Morecambe), was dissolved in dimethyl-sulphoxide (DMSO).

Bovine serum albumin; HEPES; phospholipid standards trichloro-acetic acid (TCA); ethylene glycol bis-(^β-aminoethyl ether)N,N,N'-tetraacetic acid (EGTA); trypan blue and Dowex 1 x 8(C1⁻ form, 100-200 mesh) were purchased from Sigma (U.K.). All compounds used in tissue culture (Dubecco's modified eagles medium (DME); Hanks balanced salt solution HBSS; foetal calf serum (FCS); Penicillin-Streptomycin (5000 μ /ml) and Neomycin (10,000 μ g/ml) were purchased from Gibco-Europe. Rat promegakaryoblasts (RPM) were a gift from Drs R. Weinstein and T. Maciag, Harvard Medical School, Boston, U.S.A.

TLC plate macherney Nagel "Polygram Silica G" precoated glass plates (Camlab, Cambridge, U.K.) and Merck Silica Gel 60 HP-TLC plates (Macfarlane Robson Ltd., Glasgow, U.K.) were purchased. The following compounds were also purchased: Folin and Gocaltean's phenol reagent (BDH Chemicals Ltd.); triton X-100 and scintol 2 (Koch-Light Laboratories Ltd.); Ecoscint (National Diagnostics, New Jersey); X-ray developer and fixer (Agfa Ltd.). All organic solvents and other laboratory reagents were of analytical grade.

2.1.2 Radiochemicals

Carrier-free $\left(3^{2}P\right)$ -orthophosphate (P_{i}) was kindly supplied, gratis, by the regional isotope dispensary (Western Infirmary, Glasgow, U.K.). 5-Hydroxy G-³H tryptamine creatinine sulphate (17 mCi/mmol); 8-³H Adenine (24 Ci/mmol); myo-(2-³H) inisitol (13.8 Ci/mmol) were from Amersham International (Amersham, U.K.).

2.2 Methods

2.2.1 Rat Platelets

2.2.1.1. Preparation of rat platelets

Blood was obtained from the vena cava of rats of either sex under light ether anaesthesia and collected into plastic

- 95 -

tubes containing 10% (v/v) trisodium citrate (0.1M) and centrifuged (800g; 10 min; 15-20°C). The supernatant plateletrich plasma was aspirated into plastic tubes with care taken to avoid contamination with red or white blood cells. Platelets were then pelleted by centrifugation (2000g; 4 min, 15-20°C) in the presence of PGI₂ (100 nM) to prevent platelet activation (Moncada <u>et al</u>., 1982). The supernatant plasma was aspirated and the platelets suspended in the appropriate buffer depending upon the nature of the experiment (vide infra).

2.2.1.2 Radioisotope labelling of washed platelets

Metabolism of inositol phospholipids in rat platelets was monitored in cells prelabelled with $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$. Rat platelets obtained as described (Section 2.2.1.1) were suspended in calcium-free, phosphate-free buffer (NaCl, 150 mM; KCl, 4 mM; MgCl₂, 1 mM; Na-Hepes, 5 mM; dextrose, 10 mM; Bovine serum albumin, 0.3%; pH 7.4) at 10⁹ cells/ml and incubated with carrier-free $\begin{bmatrix} 3^2 P \end{bmatrix}$ -orthophosphate (30 µCi/ml) for 90 min at 37°C, at which point labelling of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA approached steadystate (equilibrium) conditions (Figure 3). Platelets were then pelleted by centrifugation (2000g, 4 min, 15-20°C) in the presence of PGI₂ (100 nM) and resuspended in approximately 1.5 times the original labelling volume of fresh buffer.

2.2.1.3 <u>Measurement of platelet cytosolic free calcium</u>, (Ca^{2+}) loading with calcium indicator dye Quin 2 <u>AM</u>

Ca²⁺ was measured using the fluorescent indicator dye Quin-2-acetoxy methylester (Quin 2 AM). Rat platelets obtained as described (Section 2.2.1.1) were suspended in a modified

- 96 -



Figure 3: Incorporation of ³²P P_i into rat platelet phospholipids with time.

Rat platelets ($\sim 1 \times 10^9$ cells/ml) were incubated with $\begin{bmatrix} 32p \end{bmatrix}$ Pl (30 μ Ci/ml) for the times indicated, prior to extraction of phospholipids (Section 2.2.3.2.1):- phosphatidicacid (D); phosphatidylinositol (D); phosphatidylcholine (Δ); phosphatidyl ethanolamine (\bullet). Results are expressed as DPM (mean $\stackrel{+}{-}$ SD). The experiment shown is typical of one other experiment.

- 97 -

Hepes-buffered tyrodes (NaCl, 120mM; Na₃ citrate, 8.9mM; NaHCO₃, 8.9mM; DExtrose, 5.5mM; Hepes, 5mM; KH₂PO₄, 0.8mM; MgCl₂, 0.84mM; CaCl₂, 24mM, pH 7.4) at 10⁸ cells/ml. Platelets were incubated with Quin 2AM (20µM final concentration; 37° C; 10 min) before pelleting by centrifugation (2000g; 4 min; 15-20°C) in the presence of PGI₂ (100 nM). Quin 2 labelled platelets were suspended in fresh buffer at a concentration of 5 x 10⁸ cells/ml, and were kept at room temperature throughout the duration of the experiment.

Note that the above procedure differs from that used in human platelets (Rink <u>et al.</u>, 1982; Pollock <u>et al.</u>, 1984). Most importantly, incubation with Quin 2 AM is performed in plasma-free suspensions of washed platelets, as opposed to platelet rich plasma, because preliminary experiments indicated that plasma esterases, or other plasma constituents, interfere with the uptake of dye into the cell. Incubation of rat platelets with Quin 2 AM resulted in an intracellular accumulation of 2.54 \pm 0.39 mmol/litre (mean \pm SD, n = 4) Quin 2.

2.2.2 RPM

Rat promegakaryoblasts, or RPM, are a cell line originally obtained by Cicoria and Hempling (1980) from the bone marrow of a female Long-Evans rat. They have been characterised, both morphologically and biochemically, by Weinstein <u>et al</u>. (1981) as being similar to normal rat promegakaryoblasts.

RPM are grown in suspension cultures with Dulbecco's Modified Eagles Medium (DME) supplemented with 10% foetal calf serum (FCS) (vide infra). Under these conditions, the cells are in a proliferative mode. However, should they be grown under conditions of serum deprivation, or exceed a cell density of approximately $4 \ge 10^5$ cells/m (approximately 7 days continuous culture) they stop proliferating and undergo a process of maturation, passing through three stages, Stage I or promegakaryoblast; Stage II or megakaryoblast and Stage III or mature megakaryocyte (Weinstein <u>et al</u>., 1981). During the latter stages of Stage III, small vesicular bodies are visible, these have been designated Stage IV. Representative illustrations of each stage in RPM maturation, using light microscopy, are depicted in Figure 4. The following outlines the procedures used to maintain RPM in cell culture and their preparation for experimentation.

2.2.2.1 Methods of cell culture

Rat promegakaryoblasts (RPM) were grown and maintained under conditions essentially as outlined by Weinstein <u>et al</u>. (1981). RPM were grown in stationary suspension culture stocks in 100 mm cell culture dishes (Nunc) in Dulbecco's Modified Eagle's Medium (DME) and 10% foetal calf serum (FCS) at 37° C in an automatic CO₂ incubator (Forma Scientific, Marietta, Ohio, USA) under a humidified atmosphere of 95% air/5% CO₂. Antibiotics benzyl penicillin (100 iu/ml) and streptomycin (100 µg/ml) were included to cope with minor bacterial contaminations which could have arisen from non-sterile culturing techniques. Cells were monitored regularly using a light microscope (Labovent, Leitz).







<u>Figure 4</u>: Light microscopy of the RPM cell line in various stages of maturation. Top left: Stage I (promegakaryoblasts); middle right: Stage II (immature megakaryocytes); middle left: Stage III (megakaryocytes); Bottom right: Stage IV (vesicular bodies).

> After Weinstein <u>et al</u>. (1981).



TOO -

2.2.2.1.1 Maintenance of aseptic conditions

All aseptic manipulations were performed in a vertical laminar flow hood (Microflow Ltd., Andover, Hants, U.K.). Every operation using sterile solutions included the use of either 1 ml sterile graduated pastettes (Alpha Labs., Eastleigh, England); 5 ml sterile graduated movettes (Nunc, Roskilde, Denmark); or 10 ml sterile graduated plastic pipettes (Falcon) when necessary. Operations with 10 ml pipettes were made easier by the use of an automatic "Pipetus" obtained from Flow Labs., Irvine, Scotland.

2.2.2.1.2 Passaging of RPM

RPM were seeded at 1×10^6 cells/100 mm plastic petridishes in 10 ml of culture medium unless otherwise stated. On the day of passaging - usually seven days from previous passage and after 4 "doubling times" - the cells from one dish were gently aspirated and transferred to individually wrapped 17 x 100 mm sterile plastic tubes (Falcon) and pelleted (700g; 8 min; 25° C). The resulting cell pellet was suspended in fresh DME (10 mls) and 1 ml of cells aliquoted into sterile petri dishes containing the complement of medium outlined in Section 2.2.2.1.

2.2.2.2 Preparation of RPM

RPM were harvested by centrifugation (700g; 8 min; $25^{\circ}C$) and suspended in Hanks Balanced Salt Solution (HBSS) [NaCl, 137 mM; KCl, 5.4 mM; MgSO₄, 0.4 mM; Na₂HPO₄, 0.3 mM; NaHCO₃, 4.1 mM; KH₂PO₄, 0.4 mM; MgCl₂, 0.49 mM; Glucose, 5.5 mM; CaCl₂, 0.1 mM; pH 7.4] at a cell concentration of approximately 6 x 10^6 /ml. Cell viability, assessed by trypan blue exclusion, was routinely approximately 90% in both treated and untreated cells. After suspension cells were ready for immediate use. The methodology for labelling RPM with $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ and $\begin{bmatrix} ^3 H \end{bmatrix}$ inositol is outlined in the following sections.

2.2.2.3 Radioisotope labelling of RPM with ³²P P.

On day 6 after previous passage RPM were harvested as in Section 2.2.2.2 and seeded at a cell concentration of 1.1 x 10^6 /ml in 260 ml sterile plastic bottles (Nunc) containing DME and FCS. The cells were incubated with carrier free $\begin{bmatrix} 32\\ 2 \end{bmatrix} P_i$ (3 µCi/ml; 24 hrs; 37° C). RPM were then pelleted by centrifugation (700g; 8 min, 15-20°C) before suspension in HBSS at a cell concentration of approximately 6 x 10^6 cells/ml. After 24 hrs incubation, $\begin{bmatrix} 32\\ 2 \end{bmatrix} P_i$ approached equilibrium conditions with respect to PtdA, Figure 5.

2.2.2.4 Radioisotope labelling of RPM with ³H Inositol

On day 4 after previous passage $[{}^{3}H]$ inositol (10 µCi/dish) was added to several dishes and allowed to grow as normal for a further 3 days. On the seventh day, the cells were harvested as before (Section 2.2.2.2) and suspended in HBSS, containing 10 mM ICl, at a cell concentration of approximately 6 x 10⁶ cells /ml. The inclusion of ICl prevents the hydrolyses of inositol phosphates and thus facilitates their detection (Berridge et al., 1982; see also Section 1.3.1.5). Incorporation of ³H inositol into inositol phospholipids was detected, values being 310176 \pm 33416 dpm for PtdIns(4,5)P₂; 49940 \pm 19696 dpm for PtdIns(4)P and 10092 \pm 3460 dpm for PtdIns (results means





RPM (1×10^6 cells/ml) were incubated with 32 P P₁ (1.5 Ci/ml) for the times indicated, prior to extraction of phospholipids (Section 2.2.3.2.2):- Phosphatidic acid (0); phosphatidylinositol ([]); phosphatidyl choline (Δ); phosphatidyl ethanolamine (\bullet). Results are expressed as DPM (mean $^{-1}$ SD). The experiment shown is typical of one other experiment.

- 103 -

[±] SEM, n = 2). Thus accumulation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol phosphates can be assumed to be derived from these $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol phospholipids.

2.2.2.5 <u>Measurement of RPM cytosolic free calcium</u>, Ca²⁺ loading with Quin 2 AM.

RPM prepared as described (Section 2.2.2.2) were suspended in HBSS at a cell concentration of approximately 6 x 10^6 cells/ml. They were incubated (37° C, 35 min) with Quin 2 AM (10 pM, final concentration) before pelleting by centrifugation (700g, 8 min, $15-25^{\circ}$ C) and washed twice in fresh HBSS to remove unincorporated Quin 2 AM before final suspension at a cell concentration of approximately 6 x 10^6 cells/ml. Under such incubation conditions, intracellular accumulation of Quin 2 was 0.69 ± 0.34 mmol/litre (mean \pm SD, n = 4). RPM were kept at room temperature throughout the duration of the experiment.

2.2.3 Phospholipid Studies

The experimental procedure for monitoring inositol phospholipid metabolism in cells prelabelled with $\begin{bmatrix} 3^2P \end{bmatrix} P_i$ was essentially similar for both rat platelets and RPM. The experimental technique for monitoring accumulation of inositol phosphates is described in Section 2.2.3.4.

2.2.3.1 <u>Metabolism of inositol phospholipids in cells prelabelled</u> with $[3^2p]_{P}$

Radiolabelled cells (0.5 ml platelets, containing approximately 0.94 mg protein; 0.4 ml RPM, containing approximately 0.54 mg protein) were dispensed into plastic tubes at 37° C. The maximum number of samples in any single experiment was 30 whilst within each experiment triplicate determinations for each drug/vehicle treatment were performed. Reactions were initiated by the additions of agonists (usually 5 μ 1) in the presence of, or subsequent to, the addition of antagonists (usually approximately 5 μ 1) or the appropriate vehicle as a control. The design of the individual experiments was such that from start to finish of each one, a period of approximately 20-30 min had elapsed. Once the reactions had been terminated, extraction of phospholipids were carried out immediately.

2.2.3.2 Phospholipid Extraction

In order to optimize the efficiency of phospholipid extraction from rat platelets and RPM, separate extraction procedures for each cell type were utilized.

2.2.3.2.1 Extraction of phospholipids from rat platelets

Extraction of phospholipids from rat platelets was performed essentially as described by Schacht (1981). Reactions were terminated by removal of the entire platelet sample into a glass test tube containing 1.5 ml of Chloroform:Methanol (1:2) (v,v) on ice. Partition of aqueous and organic phases was facilitated by the addition of 0.5 ml, 2.4N HCl and 0.5 ml chloroform. The lower (organic) phase was then removed to a clean glass test tube and the upper (aqueous) phase and protein interface was washed with 1 ml chloroform. The second lower phase was combined with the first and the combined lower phases were washed with 2 mls of methanol: 1N HCl (1:1) (v/v). Between each partitioning step of the extraction procedure the tubes vortexed briefly and centrifuged (1000 RPM, 5 min, 5° C). The final lower phase was removed into a glass vial, dried at 45° C under nitrogen and stored at -20° C until use, usually within 24 hrs.

2.2.3.2.2 Extraction of phospholipids from RPM

Extraction of phospholipids from RPM was carried out essentially as described by Method B of Lloyd <u>et al</u>. (1972). Reactions were terminated by removal of the entire sample to glass test-tubes containing 2 ml of Chloroform:Methanol: 10N HC1 (100:200:8, v/v/v) at room temperature. Phospholipids were extracted by partitioning of the aqueous and organic phases following the addition of 0.625 ml of chloroform and 0.625 ml of distilled water. After vortexing the tubes and centrifugation (700g, 3 min, 15-20°C) the lower (organic) phase was removed into a glass vial, dried at 45° C under nitrogen and stored at -20°C until use, usually within 24 hrs. The above procedure was performed when only PtdA and PtdIns were required to be monitored. When resolution of the polyphosphoinositides e.g. PtdIns(4,5)P₂ and PtdIns(4)P were required, the more efficient Schacht extraction (Section 2.2.3.2.1) was used.

2.2.3.3 Separation of phospholipids by thin layer chromatography

Two techniques were utilized to resolve the phospholipids extracted from rat platelets and RPM. When only PtdA and PtdIns were required to be resolved, the technique described by Yavin and Zutra (1977) was used. The technique as outlined by Jolles <u>et al</u>. (1981) was performed to resolve the polyphosphoinositides, namely PtdIns(4,5)P₂, PtdIns(4)P as well as PtdIns and PtdA.

2.2.3.3.1 <u>Separation by thin-layer chromatography of PtdA and</u> <u>PtdIns</u>

Lipids were dissolved in 0.15 ml of Chloroform:Methanol (9:1) (v/v) and spotted on silica-gel t.1.c. plates (10 cm x 10 cm) (Silica Gel, 25; Camlab) for two-dimensional separation of phospholipids (Yavin and Zutra, 1977). This t.1.c. system employs a basic solvent in the first dimension - Chloroform: Methanol:40% aqueous methylamine (13:6:1.5) (v/v/v); diethylether: acetic acid (glacial) (19:1) (v/v) for the intermediary run (through the second dimension) and Chloroform:acetone:methanol: acetic acid (glacial):distilled H_20 (10:4:2:3:1) (v/v/v/v/v) for the second direction. The glass chromatographic chamber was lined at both ends with Whatman I filter paper, and the tanks were equilibrated with the solvent mixtures (approximately 200 mls in total) 2 hrs before commencing the separation. All three organic mixtures were freshly prepared and a total of 30 plates could be run without significant loss of resolution.

Phospholipid samples were applied at the lower left hand corner (2 cm in), using a 10 µl disposable micro-pipette. The solvent was dried in a stream of warm air. A total of 8 plates could be run in one batch. After a period of approximately 25 min in the first solvent, by which time the solvent front was within 1 cm from the edge of the plate, the plates were removed and dried in a stream of warm air for 10-15 min whilst placed on a tray. Each plate was then exposed to the fumes of a concentrated solution of HCl for approximately 5 min. The plates were then dried in warm air for approximately 5 min, then cool air for an additional 5 min. The plates were then placed in the second solvent running in the second dimension (origin at lower right corner) for a period of 10-15 min. Following this ether:acetic acid wash the plates were dried for 5-10 min in a stream of cool air before placing in the third solvent in the same dimension as that immediately above. A period of approximately 25 min was needed to allow the solvent front to reach within 1 cm of the edge of the plate. At this point the plates were removed from the solvent and allowed to dry, thoroughly, in a stream of cool air. After drying, the phospholipids were visualized by placing the plates in an iodine vapour tank. The spots corresponding to PtdAs and PtdIns (see Figure 6) were scraped into polythene scintillation vials and counted for radioactivity in a scintillation counter (see Section 2.2.6).

2.2.3.3.2 <u>Separation by thin-layer chromatography of polyphospho-</u> inositides

Lipids were redissolved in 0.15 ml of Chloroform:Methanol (9:1, v/v) and spotted as 1.5 cm bands (1.5 cm from edge) on potassium oxalate $(1\% \text{ in Methanol:distilled } H_2O(2:3, v/v))$ impregnated HP-TLC plates (Silica Gel, 60; Merck, 20 x 20 cm) under a stream of warm air. Prior to spotting the plates were 'activated' by placing in an oven for 15 min at $110^{\circ}C$. The plates were developed, in the longitudinal dimension, in

Chloroform:acetone:methanol:glacial acetic acid:distilled H_2^0 (40:15:13:12:8; v/v/v/v) until the solvent front was 12-14 cm from the base line. The solvent was always freshly prepared.

- 108 -



Figure 6: Schematic representation of phospholipid separation by two-dimensional thin layer chromatography.

Phospholipids identified by iodine staining are indicated. 0 - origin; l. Neutral lipid; 2. phosphatidyl ethanolamine; 3. phosphatidyl choline; 4. sphyngomyelin/lyso-phosphatidyl ethanolamine; 5. phosphatidyl inositol; 6. phosphatidyl serine; 7. phosphatidic acid; 8. plasmalogen lysophosphatidyl ethanolamine. Individual phospholipids were visualised by autoradiography (X-Omat S plates, Kodak) and identified using phospholipid standards. Phospholipids separated by this method are illustrated schematically in Figure 7. $PtdIns(4,5)P_2$ and PtdIns(4)P migrate as discrete bands as does PtdIns. PtdA migrates with, or just behind, the solvent front. The spots corresponding to the relevant phospholipids were scraped into polythene scintillation vials and counted for radioactivity as before.

2.2.3.4 Metabolism of inositol phospholipids in cells prelabelled with $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol

Rat platelets, like human platelets, fail to incorporate $[^{3}H]$ inositol into their phospholipids in a cost-effective manner (Section 2.2.9.1.2). However, due to the fact that RPM are grown and maintained in culture, $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol can be incubated with the cells and incorporated into the phospholipids over a period of days. RPM, prelabelled with [³H] inositol, (0.5 ml, containing approximately 0.67 mg protein) were dispensed into plastic reaction tubes at 37° C. The maximum number of samples in any single experiment was 24, whilst within each experiment triplicate determinations for each drug/vehicle treatment were Reactions were initiated by the additions of agonists performed. (usually $< 5 \mu$) or appropriate control. The design of the individual experiments meant that the duration of each experiment lasted 20-30 min. Once the reactions had been terminated, extraction of inositol phosphates was carried out as detailed in Section 2.2.3.5.



Figure 7: Schematic representation of phospholipids separated by one dimensional thin layer chromatography.

Phospholipids identified by iodine staining are indicated. O - origin; PtdIns4,5P (Phosphatidyl inositol-4,5-bisphosphate; PtdIns4P (Phosphatidyl²inositol-4-phosphate; PtdIns (Phosphatidyl inositol); PtdA (Phosphatidic Acid); NL - Neutral Lipids; SF - Solvent Front. Extraction of inositol phosphates from RPM was performed as follows. The reactions were terminated by adding 0.5 ml of 20% (w/v) trichloroacetic acid (ice-cold). After 10 min at 0°C, precipitated proteins were removed by centrifugation (1000g; 10 min; 4° C). The acid soluble fraction was collected and analysed immediately or frozen (-20°C) until (³H] inositol phosphates were to be analysed. Unfrozen or thawed aliquots of [³H] inositol phosphates were extracted three times with 4 volumes of water-saturated diethylether. After driving off the residual ether in a boiling water bath, the (³H) inositol phosphates in the extracts were then separated on dowex anion-exchange columns.

2.2.3.6 <u>Separation by anion-exchange chromatography of inositol</u> phosphates

Samples of $[{}^{3}H]$ inositol phosphates were collected and distilled water was added (usually < 200 µl) to obtain a 1 ml aliquot. The 1 ml sample was then applied to columns containing approximately 1 ml of Dowex 1 x 8 anion-exchange resin (formate form) and the phosphate esters eluted, essentially as described by Berridge <u>et al.</u> (1983) by the stepwise addition of solutions containing increasing levels of formate. The following procedure was routinely followed: $[{}^{3}H]$ inositol eluted by addition of 7 ml distilled water; $[{}^{3}H]$ glycerophosphoinositol with 8 ml 5mMdisodiumtetraborate/60mM-sodium formate; $[{}^{3}H]$ InslP with 8 ml 0.1M formicacid/0.2M-ammonium formate; $[{}^{3}H]$ Ins $(1,4)P_{3}$ with 0.1M formic acid/0.4M ammoniumformate and $[{}^{3}H]$ Ins $(1,4,5)P_{3}$ with 0.1M formic acid/10M ammonium formate. 2 ml samples of each fraction was taken for liquid-scintillation counting.

2.2.4 <u>Measurement of cytosolic free calcium, [Ca^{2+]}</u>,

 $\left[\operatorname{Ca}^{2+}\right]_{i}$ was measured using the fluorescent calcium indicator dye Quin 2 (Tsien, 1982). The polycarboxylic anion permeates the cell membrane in the form of acetoxymethyl Quin 2 or Quin 2 AM. Once it has passed through the cell membrane it readily distributes throughout the entire cytoplasm where it is cleaved by nonspecific esterases, generating and trapping the active form of the dye, Quin 2, in the cytoplasm.

Quin 2 binds Ca^{2+} preferentially, compared to Mg^{2+} , in a 1:1 stiochiometry with an apparent dissociation constant of 115 nM. Upon binding Ca^{2+} the dye exhibits an increase in fluorescent intensity which can be calibrated to reveal $[Ca^{2+}]_i$ (vide infra).

In both cell types, 0.5 ml aliquots together with 1.5 ml buffer (modified Hepes buffered tyrodes or HBSS depending upon cell type) were placed in a 1 cm square quartz cuvette and fluorescence was monitored in a Perkin Elmer - LS3 Fluorescence Spectrometer, thermostatted at 37° C, with standard monochromator settings of 339 nm - excitation and 492 nm emission. To record the development of response a chart recorder (Linseis) was linked to the spectrometer. Prior to addition of agonist to rat platelet samples, the external free calcium was adjusted to 1 mM by addition of CaCl₂. For RPM studies, sufficient Ca²⁺ was present in the HBSS and no further addition was required. Changes in fluorescence consequent upon agonist addition, were recorded and calibrated to levels of $\lfloor Ca^{2+} \rfloor_i$, using the following formula:-

$$\begin{bmatrix} Ca^{2+} \end{bmatrix}_{i} = Kd(F - F_{min}) \\ F_{max} - F)$$

where F is the fluorescence recorded, F_{min} and F_{max} are the fluorescence recordings at very low and very high Ca²⁺ respectively. Kd is the dissociation constant = 115 nM.

 F_{max} was determined by saturating Quin 2 with Ca²⁺. This was achieved by lysing the cells with digitonin (50 µM for rat platelets and 100 µM for RPM) which exposed the entrapped dye to the high extracellular calcium concentration. The addition of EGTA (0.02M, final concentration) to both rat platelets and RPM chelated all available Ca²⁺ thus allowing determination of F_{min} .

Preliminary investigations were carried out to test the validity of this technique, for measuring levels of $\left[Ca^{2+}\right]_{i}$ in both rat platelets and RPM.

Tsien <u>et al</u>. (1982a) demonstrated that upon hydrolysis of Quin 2 AM there is a spectral shift and peak fluorescence at 492 nm (when excited at 430 nm) indicating the formation of the active dye Quin 2. Studies following the spectral characteristics of cells loaded with Quin 2 AM (Figures 8a and b) demonstrates a peak in fluorescence at around 492 nm, suggesting the presence of Quin 2 in these samples. In addition to being present inside the cells, an optimal intracellular concentration of Quin 2 - in the millimolar region - is desired in order to report accurately concentrations of $[Ca^{2+}]_i$. Reference to Sections 2.2.1.3 and 2.2.2.5 demonstrate that this criterion is met with, intracellular concentrations of Quin 2 of 2.5 mM and 0.69 mM being present in rat platelets and RPM respectively. As already noted, Quin 2 is a Ca^{2+} -chelator (see also Section 2.2.9.2) and may in fact blunt transient increases in $[Ca^{2+}]_i$ or





Aliquots (2 ml) of rat platelets incubated in the presence of Quin 2 AM (closed symbols) as outlined in Methods (Section 2:2:1:3) or absence of Quin 2 AM (open symbols) were excited at 339 nm and their emission spectra followed. Presence of Quin 2 is indicated by subtracting (auto) fluorescence detected in platelets not incubated with Quin 2 AM (open symbols) from fluorescence detected in Quin 2 AM loaded platelets (closed symbols) to yield the difference spectrum which is attributed to Quin 2 (dotted line). Fluorescence measured in arbitary units.



Figure 88. Spectral characteristics of RPM incubated with Quin 2 AM.

Aliquots (2 ml) of RPM preincubated in the presence of Quin 2 AM (closed symbols) as outlined in Methods (Section 2:2:2:5) or absence of Quin 2 AM (open symbols) were excited at 339 nm and their emission spectra followed. Presence of Quin 2 is indicated by subtracting (auto) fluorescence detected in RPM not incubated with Quin 2 AM (open symbols) from fluorescence detected in Quin 2 AM loaded RPM (closed symbols) to yield the difference spectra which is attributed to Quin 2 (dotted line). Fluorescence measured in arbitary units.

and the second second

and the second and the states and the states of the second second second second second

slow down rapid increases induced by agonists. This phenomenon may be expected to be more pronounced in studies involving rat platelets compared to those utilizing RPM. With hindsight studies designed to estimate the importance of this phenomenon by performing the experiments with a range of intracellular concentrations of Quin 2 should have been performed. However, time did not permit this. Finally, Quin 2 AM has been reported to be non-toxic in lymphocytes up to a concentration of 2-3 mM (Tsien et al., 1982a). In order to demonstrate that - at the concentrations used in the present experiments - Quin 2 AM was non-toxic - several indices of cellular viability were recorded. Figures 14 and 31 in the Results Section demonstrate platelet aggregation in the presence of Quin 2 AM, the responses obtained suggest no untoward damage to the cells has occurred. In the present studies it was difficult to demonstrate biological activity of RPM subsequent to agonist addition (see Section 3.2.1.1). However, using Trypan Blue exclusion, pretreatment with Quin 2 AM routinely revealed approximately 90% viability. Similar degrees of viability were reported in the absence of dye.

2.2.5 Platelet Functional Tests

Washed platelets prepared under labelling conditions, but with the label $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ or Quin 2 AM excluded, were examined by platelet aggregometry as an index of functional responsiveness under the experimental techniques used. Aliquots (0.45 ml) of rat platelets prepared as described were placed in plastic cuvettes in a thermostatted (37°C) aggregometer. Aggregation, measured photometrically using an aggregometer (Malin Electronics) was initiated by the addition of agonist or appropriate vehicle,

- 117 -

usually <5 μ l volumes were added. The development of aggregatory response was followed on a Linseis chart recorder. The extent of aggregation was measured by expressing the net maximal increase in aggregation induced by an agonist as a percentage of maximal aggregation observed using 50 nM (1 μ /ml) Thrombin.

2.2.6 Quantification of Radioactivity

After iodine had sublimed, identified spots/bands were scraped using a surgical blade and incorporated radioactivity quantified by liquid scintillation spectrometry (Packard, Model 3390). Scintillant used was either 10 mls toluene:triton-X 100: Scintol 2 (1223:667:100, v/v/v) or 5 ml Ecoscint (National Diagnostics, N. Jersey). 2 ml aliquots of the separated $\begin{bmatrix} ^{3}H \end{bmatrix}$ inositol phosphates were quantified in a similar manner using either 20 ml of toluene:triton-X100; scintol 2 or 10 ml Ecoscint.

2.2.7 Calculation of Results

Results obtained from the scintillation counter were corrected for background radiation then expressed as counts per minute (cpm); disintegrations per minute (dpm) or as fold stimulation of control levels where a value of 1, by definition, is equivalent to basal levels. Simple statistical analyses of replicate values within an experiment was carried out to give mean values \pm standard deviation (S.D.) or \pm Standard error of the mean (S.E.M.). Statistical significance between the effect of an agonist and the control (vehicle or distilled water or saline), or the effect of an agonist and the effect of an (pAugA) antagonist, was calculated using Student's t-test. Significant

- 118 -

differences are indicated by probability (P) values; * = P < 0.05; ** = P < 0.005; *** = P < 0.0005. The term EC_{50} is used to describe the concentration of agonist that elicits 50% of the maximum response to that agonist.

2.2.8 Other Assays

Trypan Blue Exclusion

Viability of RPM were monitored before and after each labelling procedure by obtaining a 1:100 dilution of RPM with trypan blue (0.5% trypan blue in 0.9% saline). Cells were counted using a haemocytometer (Hawksley, England) and a light microscope (Labovent, Leitz).

Protein Determination

Proteins were determined by the method of Lowry <u>et al</u>. (1951) using bovine serum albumin (Sigma) as standard.

en galante en providente en providente para entre la composición de

to the state of th

2.2.9 Aim of Study and Rationale behind Studies

Besides acting at the level of the platelet, antithrombotic drugs, such as aspirin, may also act at the level of the progenitor cell, the megakaryocyte. Studies by Demers et al. (1980) and Worthington and Nakeff (1982) demonstrated a lack of cyclooxygenase activity in rat platelets after aspirin treatment at times long after the drug would have been removed from the animal. They proposed that aspirin, in addition to inhibiting platelet cyclooxygenase, may also inhibit cyclooxygenase in megakaryocytes. In a separate study, using human tissue, Bye et al. (1979) demonstrated the inability of arachidonic acid to cause platelet aggregation four days after administration of aspirin. They also concluded that aspirin may have an effect on platelet progenitor cells. Before further studies can be undertaken in investigating a possible dual site of action for antithrombotic drugs, or even a single action at the level of the megakaryocyte, a thorough background knowledge of the mechanisms underlying cellular activation in both these cell types is required. Such investigations will aid further studies into whether or not megakaryocytes are a potential site for antithrombotic therapy. However, any such investigation would require a ready supply of viable megakaryocytes. The difficulty of obtaining human bone-marrow tissue and the paucity of cells obtainable from such sources (see Section 1.2) precluded the use of human tissue in the present studies. However, to resolve this quantitative limitation I used a cell line of rat bone

marrow derived promegakaryoblasts or RPM. These cells were derived from the bone-marrow of a long-Evans female rat by Cicoria (Cicoria and Hempling, 1980) and can be grown and maintained in culture thus allowing limitless supply of tissue for investigation. Having assured myself of the similarity between normal rat promegakaryoblasts and the RPM line (see Section 2.2.2) I undertook an

the second second states and the second seco

investigation into the mechanisms underlying cellular reactivity in both rat platelets and RPM. Although the RPM are representative of a maturational stage occurring early in the sequence of rat megakaryocyte development, it would prove interesting to observe whether or not the mechanisms (biochemical and/or ionic) regulating cellular reactivity were established in these cells and to compare them with those occurring in their progeny, rat platelets. Mechanisms thought to govern cellular reactivity in a number of different cell types, including human platelets, include the following: i) generation of cyclic AMP; ii) production of arachidonic acid metabolites; iii) inositol phospholipid metabolism and iv) changes in cytosolic free calcium $\left[\operatorname{Ca}^{2+}\right]_{i}$. Preliminary studies, carried out in this department using RPM, demonstrated that these cells could both synthesise cyclic AMP after exposure to a suitable stimulus and synthesize icosanoids (MacIntyre et al., 1983a,b). Table 1 demonstrates that after addition of 0.1-10 µM agonist, RPM were capable of synthesizing significant amounts of cyclic AMP. Interestingly, 1-10 µM PGD, failed to elicit significant formation of cyclic AMP suggesting that like rat platelets (Hwang, 1980), RPM do not possess receptors for PGD₂. Table 2 demonstrates the ability of RPM to synthesize a range of icosanoids when labelled with an appropriate marker. Significant amounts of PGE,, TXB, and PGD₂ are formed with lesser amounts of 6 keto $PGF_{1\alpha}$ and $PGF_{2\alpha}$ being detected. Again these results are similar to those attainable when using rat platelets.

The present study aims to investigate the nature of the remaining two mechanisms, namely inositol phospholipid metabolism and changes in cytoplasmic free calcium $(Ca^{2+})_i$, in both rat platelets and RPM, subsequent to agonist addition.
ILOSANDID	Concentration (µM)	Cyclic AMP (pmoles/10 ⁶ cells)
PGE ₁	0.1	0.83 <u>+</u> 0.44
PGE1	1.0	1.87 <u>+</u> 0.6 9
PGE1	10.0	4.90 <u>+</u> 1.30
PGI2	0.1	0.09 + 0.09
PGI2	1.0	0.45 ± 0.23
PGI2	10.0	2.3 9 <u>+</u> 0.30
PGD ₂	0.1	0
	1.0	0.07 <u>+</u> 0.07
PGD ₂	10.0	0.23 <u>+</u> 0.14
PGF ₂₂	0.1	0
PGF23	1.0	0
PGF22	10.0	0
6 Keto PGE₁	0.1	0.1 <u>+</u> 0.10
6 Keto PGE	1.0	0.53 ± 0.50
6 Keto PGE 1	10.0	2.3 <u>+</u> 0.83
PGEo	0.1	0.58 <u>+</u> 0.43
PGE ₂	1.0	1.33 <u>+</u> 0.44
PGE ₂	10.0	9.00 <u>+</u> 1.61

Table 1: Icosanoid-induced cyclic AMP formulation by RPM.

Icosanoids were incubated with RPM for 60 sec, at $37^{\circ}C$ and reactions terminated by the addition of ethanol. Total cyclic AMP was measured by competitive protein binding assay. Results are mean values - SEM, above basal levels of cyclic AMP, of 3-4 experiments each performed in triplicate.

> After MacIntyre <u>et al</u>., (1983a)



Table 2: Icosanoid production by RPM.

RPM, or medium alone in control samples, were incubated with ${}^{3}\text{H}$ arachidonic acid at 37°C for 2 hours. Lipids were then extracted from the cell-free supernatent, separated by TLC and identified using authentic PG standards. Percentage conversion of $[{}^{3}\text{H}]$ - arachidonate by RPM was estimated by subtracting the corresponding control values. The results are the mean values \pm SEM of 4 experiments each performed in triplicate and are expressed as percentages of the total radioactivity added to the TLC plate.

After MacIntyre et al. (1983;

The methods used were in the first instance gleaned from the literature and subsequently modified in order to optimise the experimental procedure. To monitor inositol phospholipid metabolism the use of radioactive isotope markers was employed; changes in cytosolic free calcium were monitored using the fluorescent indicator dye Quin 2. A brief discussion on the benefits and failings of these techniques now follows.

2.2.9.1 Radioactive markers and inositol phospholipid metabolism

The use of radioactive markers in monitoring agonistinduced inositol phospholipid metabolism is widespread. However, such a technique inevitably involves a compromise between the duration of prelabelling - allowing the radioisotope to be incorporated into the molecular species under investigation and the length of time that the tissue has been removed from the body to allow such prelabelling. This presents a problem when using blood platelets which have a finite lifespan (a few hours) outwith the corporeal circulation. However, such compromises are obviated when cells that can be maintained in culture (such as RPM) are used. Under such conditions, radioactive isotopes can remain in contact with the cells for long periods (up to many days) until they come into isotopic equilibrium with radioactive markers.

The metabolic pathways of the inositol phospholipid interconversions are described in Section 1.3.1.5. The following lists various methods by which radioactive markers are utilized to follow the metabolism of inositol phospholipids.

- 124 -

2.2.9.1.1 Incorporation of [32P] P.

The initial observation by Hokin and Hokin (1953) was that in pancreatic slices, acetylcholine caused a rapid and apparently specific incorporation of $\begin{bmatrix} 3^2 p \end{bmatrix} p_1$ into PtdIns and Since this initial study many experiments have been PtdA. performed monitoring the incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ into inositol phospholipids following agonist addition. However, this method effectively monitors the rate of resynthesis of inositol phospholipids rather than the proposed initial event: inositol phospholipid metabolism. Additionally, resynthesis of inositol phospholipids may be prone to interference thus producing artefactual responses. For example Egawa et al. (1981) demonstrat. ed that low concentrations of calcium (10 µM) substantially reduced the activity of CDP diglyceride inositol (PtdA) transferase thus giving an apparent increase in inositol phospholipid metabolism without actually enhancing the rate of metabolism. Similar inhibitory effects of calcium have been noted in platelets (see Broekman et al., 1980). To overcome this problem one could label up the inositol phospholipids to isotopic equilibrium with cellular $\begin{bmatrix} 32 \\ P \end{bmatrix}$ ATP and monitor the metabolism of inositol phospholipids. However, in most cells so far examined, PtdIns by far represents the larger proportion of inositol phospholipids, being approximately 80-85% of the total inositol lipid species. PtdIns(4)P and PtdIns(4,5)P₂ are usually present in approximately equal amounts (Downes and Michell, 1982). Under resting conditions PtdA is present in trace amounts. Given this situation it may take several hours (or days) to label PtdIns to isotopic

equilibrium. However, given the relatively small amounts of PtdIns(4)P and $PtdIns(4,5)P_2$ and the fact that their 4- and 5phosphate groups turnover extremely rapidly (Hawthorne and Pickard, 1979; Downes et al., 1982) and derive their phosphates directly from ATP via kinase activity, these inositol phospholipids will label up relatively quickly. In human platelets $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA is known to be in isotopic equilibrium with cellular $\begin{bmatrix} 32 \\ P \end{bmatrix}$ ATP within 1-2 hrs (Holmsen et al., 1984), an event reflecting its presence in exceedingy small amounts relative to the inositol phospholipids. During agonist-induced metabolism of inositol phospholipids, PtdA is derived from 1,2-diacylglycerol. Using techniques in which $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA is in isotopic equilibrium with $\begin{pmatrix} 3^2 P \end{pmatrix}$ ATP, any changes in levels of 1,2diacylglycerol will result in a change in levels of $\begin{bmatrix} 3^2 P \end{bmatrix} Pt dA$. In lymphocytes it has been demonstrated that calcium can activate triacylglycerol lipase to produce 1,2-diacylglycerol which in turn is then formed into PtdIns (Allen and Michell, 1978). Clearly this represents another artefactual response when one is monitoring the perturbation of the inositol phospholipid signal pathway. However, at least in human platelets, triacylglycerol lipase is present in negligible amounts (Lote and Lowery, 1979). Assuming that inositol phospholipids are the only source of 1,2-diacylglycerol, then changes in agonist-induced formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA reflects an increase in PtdA mass. However, the same can not hold true for PtdIns. Since, under conditions where $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA is in isotopic equilibrium with cellular $\begin{bmatrix} 3^2 P \end{bmatrix}$ ATP, due to its presence in much larger amounts, only a small percentage of the PtdIns pool will be labelled up with $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$.

Hence, subsequent to agonist addition, more $\begin{pmatrix} 3^2P \end{pmatrix}P_i$ will be incorporated into PtdIns, thus increasing its specific activity, ipso facto, changes in $\begin{pmatrix} 3^2P \end{pmatrix}PtdIns$ do not necessarily reflect changes in PtdIns mass. When monitoring changes in $\begin{pmatrix} 3^2P \end{pmatrix}PtdIns$ (4)P and $\begin{pmatrix} 3^2P \end{pmatrix}PtdIns(4,5)P_2$ this complication is relatively insignificant as a result of the higher degree of labelling attained in these lipids by virtue of the rapid turnover of their 4- and 5-phosphates.

To date, most studies utilizing $\begin{bmatrix} 3^2 p \end{bmatrix} P_i$ are performed under non-equilibrium conditions and monitor either disappearance of the polyphosphoinositides (the proposed initial event after agonist-receptor interaction) or the appearance of $\begin{bmatrix} 3^2 p \end{bmatrix} PtdA$.

2.2.9.1.2 Incorporation of [3_H] inositol

As outlined in Section 1.3.1.6 the initial event occurring following occupancy by agonist of receptors known to act predominantly through changes in cytosolic free calcium, is a decrease in level of $PtdIns(4,5)P_2$ with a concomitant increase in level of $Ins(1,4,5)P_3$. Thus a more realistic approach to studying the metabolism of inositol phospholipids would be to monitor these two events. The former event has been discussed elsewhere. However, one additional drawback in monitoring changes in levels of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdIns $(4,5)P_2$ is that the extent of agonist induced breakdown of this lipid is minor. Consequently a small decrease in radioactivity against a relatively high background is difficult to detect. However, against a low, if not negligible background of $Ins(1,4,5)P_3$, any increment in this compound will be easily detected. As such measurement of $Ins(1,4,5)P_3$ formation represents perhaps the most sensitive index of inositol phospholipid metabolism.

Unfortunately, a drawback to using inositol in platelets appears to be its relatively slow uptake into the cell. Compared to other tissues, human platelets require up to ten times as much inositol label compared to other cells (compare Watson <u>et al.</u>, 1984 and Rubin <u>et al.</u>, 1984). Rat platelets also fail to adequately incorporate inositol into their inositol-containing lipids (present study). Due to the prohibitively high cost of this label, studies monitoring accumulation of inositol phosphates in platelets (human and rat) are limited. Conversely, RPM, when incubated with $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol over a period of days, do incorporate sufficient amounts of label in order for it to be used while monitoring agonist-induced accumulation of inositol phosphate.

2.2.9.1.3 Incorporation of [3_H] glycerol and fatty acid labelling

A recent study demonstrated that the inositol phospholipids share a common "backbone" namely stearoylarachidonoylglycerol (Mauco <u>et al.</u>, 1984) from which all inositol phospholipids may be formed. Subsequent to its phosphorylation to glycero-3phosphate, $\begin{bmatrix} 3\\ H \end{bmatrix}$ glycerol enters the inositol phospholipid metabolic cycle at PtdA. This label is of use if one wishes to monitor changes in mass of the individual inositol phospholipid species since incorporation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ glycerol into inositol phospholipids reflects their relative mass. However, one drawback of using this label under non-equilibrium conditions is that whilst metabolism of inositol phospholipids may be detected, - 129 -

the corresponding increase in $\begin{bmatrix} 3\\H \end{bmatrix}$ PtdA may yield an underestimate of the actual amounts of $\begin{bmatrix} 3\\H \end{bmatrix}$ PtdA formed.

An alternative ${}^{\dagger}_{1}$ using ${}^{3}_{H}$ glycerol would be to use fatty acid labelling techniques. Fatty acids are easily incorporated into the phospholipids of blood platelets. In particular ${}^{3}_{H} - \text{or } {}^{14}_{C} - \text{arachidonic acid is often used: arachidonate}$ being the predominant fatty acid esterified in the 2-position of the glycerol backbone in inositol phospholipids (Mahadevappa and Holub, 1982). However, one major drawback to using ${}^{3}_{H}$ or ${}^{14}_{C} - \text{arachidonic acid is that it can be released from}$ phospholipids, including inositol phospholipids, either by phospholipase A_{2} attack, by transacylation, or after hydrolysis of the inositol phospholipids to 1,2-diacylglycerol, by the sequential actions of diglyceride/monoglyceride lipases (see Section 1.3.1.4).

2.2.9.1.4 Direct measurement of inositol phospholipid metabolism

Metabolism of inositol phospholipids may be monitored by chemical assays such as a phosphorus assay (Bowyer and King, 1977). However the sensitivity of this technique is exceedingly low (0.5 - 1 nmole inorganic phosphorus) and would require large amounts of freshly drawn blood to perform an average sized experiment. This, coupled to the fact that phosphorus assays are notoriously laborious, time consuming and prone to artefacts, precludes this assay being routinely used. Resolution of the mass of inositol-containing metabolites, using gas chromatography has recently been reported by Rittenhouse and Sasson (1985). However, such a technique also necessitated large quantities of biological tissue and requires sophisticated apparatus,

2.2.9.2 Measurement of cytosolic free calcium, $\left[Ca^{2+}\right]_{1}$.

In the present studies, the fluorescent indicator dye Quin 2 was used to monitor the concentration of cytosolic free calcium, $\left[\operatorname{Ca}^{2+}\right]_{i}$. Other calcium-indicators might also have been used: these include the bioluminescent indicator, Aequorin, and fluorescent indicators such as Chlortetracycline. A brief outline of the properties, advantages and disadvantages of these various calcium-indicators including Quin 2 and the newer Fura 2 derivatives now follows.

2.2.9.2.1 <u>Aequorin</u>

Aequorin is a Ca²⁺-sensitive photoprotein of 20,000 daltons obtained from the photocytes of jellyfish, Aequorea aequorea. Previously acquorin could only be used in cells that were sufficiently large to allow microinjection of the dye, howeve the technique has now been modified for use in (human) platelets (see Johnson et al., 1985). Advantages of using aequorin include rapidity of response (5-10 mgc); great range of sensitivity with calcium dependent measurable light output over the entire physiological range of calcium concentration from $10^{-8} - 10^{-4}$ molar and it is relatively nontoxic. Compared to Quin 2 (vide infra) whose fluorescence intensity increases only twice over its range of useful sensitivity, the light output of aequorin increases with $\left[Ca^{2+}\right]_{i}^{2\cdot 5}$, thus aequorin can provide bright signals from small zones of the cytosol exposed to local $\begin{bmatrix} Ca^{2+} \end{bmatrix}_i$ transients. However, this property of aequorin proves to be one of its disadvantages in that basal levels of $\begin{bmatrix} ca^{2+} \end{bmatrix}$

and agonist-induced elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ reported by this compound are much higher than those reported by Quin 2 (see Johnson <u>et al.</u>, 1985; Salzman <u>et al.</u>, 1985), presumably a result of aequorin reporting both mean cytosolic levels of $\left[\operatorname{Ca}^{2+}\right]_{i}$ and predominantly local zones of elevated calcium in the platelet. In some circumstances, the capacity of aequorin to report local zones of calcium may be used to one's advantage, particularly in studies examining the Ca²⁺-dependency of an agonist. In studies utilizing Quin 2, which is distributed homogenously throughout the cytoplasm, any localized increase in calcium may not be detected. However, using aequorin such localised increases would be recorded. Another disadvantage of aequorin is that it is consumed during the generation of the calcium signal.

2.2.9.2.2 Chlortetracycline

Chlortetracycline was one of the first indicators of intracellular $[Ca^{2+}]i$. It can bind Ca^{2+} in 1:1 stochiometry to form a fluorescent complex and has a Kd of 4 x 10^{-4} M(Caswell and Hutchinson, 1971a). The fluorescence intensity is very much more intense in a non polar than in a polar environment (Caswell and Hutchinson, 1971b). This resulted in concept that chlortetracycline reported levels of 'membrane bound' calcium. Experiments performed by Thompson and Scrutton (1985) confirmed this claim in their studies in human platelets. However, they were unable to define the membrane site(s) monitored by chlortetracycline. Separate studies performed by Jy and Haynes (1984) suggest that chlortetracycline may be best utilized in monitoring the internal free $[Ca^{2+}]_i$ in the lumen of organelles. 2.2.9.2.3 Quin 2

Quin 2, a polycarboxylate anion analogue of EGTA, was. custom-synthesized by Tsien in 1980. Properties of Quin 2 which make it useful as an indicator of $\left[\operatorname{Ca}^{2+}\right]_{i}$, include an ability to bind Ca^{2+} with marked selectivity over Mg^{2+} and an indifference to intracellular pH. It binds Ca²⁺ with a 1:1 stochiometry and has a Kd of 115 nM which is near to resting values of $\left[\operatorname{Ca}^{2+}\right]_{i}$ in many cells. Upon binding calcium it rapidly undergoes an apparent 6 fold increase in fluorescence intensity with no spectral shift (Tsien et al., 1982a). 0ne drawback to the general use of Quin 2 in measuring $\left[\operatorname{Ca}^{2+}\right]_{i}$ was its inability to permeate across the plasma membrane. However, by esterifying the compound, Tsien (1981) developed a nondisruptive technique by which one could load population of suitable cells with the dye. Once the esterified dye has permeated the plasma membrane, non-specific esterases splice the ester bond thus regenerating the active quinoline dye.

Although Quin 2 has found much success as a monitor for $\left[\left(\operatorname{Ca}^{2+}\right)_{i}\right]_{i}$, it has always had severe and acknowledged limitations (Tsien <u>et al.</u>, 1982a). Firstly, the dye saturates at $\left[\operatorname{Ca}^{2+}\right]_{i}$ greater than 1 µM, concentrations observed to occur in activated cells. Consequently values reported above 1 µM are subject to error; b) Quin 2 loadings of at least several tenths of milli-molar are required to obtain well-calibrated responses. Such high concentrations are required to counteract the auto-fluorescence obtained when the cells are exposed to 339 nm wavelength, the optimal wavelength for exciting Quin 2. However,

- 13.2 -

at such high loading concentrations Quin 2 may act to buffer any evoked changes in $[Ca^{2+}]_i$; c) Quin 2 signals $[Ca^{2+}]_i$ only as a change in the fluorescence intensity, which can be highly dependent on illumination intensity or dye concentration. A much better situation would be to have an indicator that responded to calcium by shifting wavelengths while maintaining strong fluorescence; d) finally, several reports of side effects during Quin 2 loading have been reported including a lowering of ATP concentration, an increase in lactate production, and some degree of mitogenic stimulation in murine lymphocytes (Tsien <u>et al</u>., 1982; Hesketh <u>et al</u>., 1983); activation of phosphorylase in hepatocytes (Charesk <u>et al</u>., 1983) and inhibition of glycolysis in human red cells and increased calcium permeability (Tiffert et al., 1984).

Whilst Quin 2 has been of use in monitoring $\left[\operatorname{Ca}^{2+}\right]_{i}$ in a variety of different cell types, the above limitations dictate that newer and better probes of $\left[\operatorname{Ca}^{2+}\right]_{i}$ be designed. Such probes may be the newly developed stilbene chromophores such as fura 2.

2.2.9.2.4 Fura 2

By combining a stilbene fluorophore with the octacoordinate, tetracarboxylate pattern of liganding groups characteristic of EGTA, Grynkiewicz et al. (1985) have developed new $[Ca^{2+}]_i$ indicators with better fluorescence properties than Quin 2. Typical of these new indicators is Fura-2 which can be incubated with and loaded into populations of cells in a manner similar to that used with Quin 2.

- 133 -

The salient advantages these new indicators appear to have over Quin 2 are a) an approximate 30 fold brighter fluorescene; b) major changes in wavelength upon binding Ca^{2+} . Consequently calibration of the signal is performed by taking the ratio of fluorescence intensities of the dye at two distinct excitation wavelengths; c) slightly higher dissociation constants, thus improving the resolution of $[Ca^{2+}]_i$ levels above 1 μ M; d) slightly longer wavelengths of excitation, thus decreasing the degree of autofluorescence and finally e) a much improved selectivity to Ca²⁺ over other divalent ions.

These properties of fura-2, and like compounds, (e.g. Indo-1) should prove of great advantage in further elucidating the mechanisms underlying calcium homeostasis in platelets and other cell types.

3. RESULTS

The mechanisms underlying cellular reactivity in both rat platelets and RPM were investigated. A number of known platelet agonists (for human platelets) were screened namely ADP, Thrombin, Adrenaline and U44069 (a TxA, mimetic). Agonists observed to produce a reasonable and measurable response in both cell types namely ADP and Thrombin - were further investigated. Such investigations included observing (i) the agonist-induced metabolism of inositol phospholipids with respect to both its time-course and concentration-dependence. This was assessed by (i) monitoring the accumulation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA; disappearance of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdIns(4,5)P₂ or accumulation of $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol phosphates, depending upon the cell type; (ii) the inhibition of such agonist-induced responses using agents known to inhibit platelet function; (iii) the concentration-response relationship for agonist-induced changes in cytosolic free calcium-monitored using the quinoline dye Quin 2 - and its inhibition by agents known to affect platelet function. Lastly, an attempt was made to characterise the nature of the receptor (purinoreceptor) recognizing ADP in both rat platelets and RPM.

ADP

A dialysable, heat-stable factor present in acid extracts of red cells and platelets was shown to cause platelet adhesion to glass (Hellem, 1960) and platelet aggregation (\emptyset llgaard, 1961). This factor was identified as ADP by Gaarder <u>et al</u>. (1961).

When added to platelets ADP causes aggregation which is preceded by shape change (Born, 1962, 1970). In human platelets,

depending upon concentration of ADP, biphasic aggregation and platelet release reaction can occur. In rat platelets only single phase aggregation and no release reaction is observed (Mills, 1970; Sinakos and Caen, 1967). The reason(s) for this species differences is unknown. However, since the release of platelet granule constituents can initiate the activation of other platelets and hence play an integral role in haemostasis, ADP may be more important in haemostasis in man than in the rat. Exposure of ADP to human platelets also results in the inhibition of platelet adenylate cyclase, a response believed to be mediated by the same receptor eliciting platelet aggregation (Cusak <u>et al.</u>, 1985). The effect of ADP, in human platelets, can be inhibited competitively by ATP (Macfarlane and Mills, 1975) and non-competetively by 2-alkylthic analogues of AMP and ATP (Cusack and Hourani, 1982; Cusack <u>et al.</u>, 1985).

Besides platelet dense granules where it is present along with 5HT, Ca²⁺ and ATP, other potential sources of ADP include damaged erythrocytes or vascular cells. Indeed Born (1962) reported that ADP is released from damaged cells into the blood at a sufficient rate to account for the aggregation of platelets in haemostases.

ADP belongs to the family of compounds known as purines. Receptors for the action of purines on different cell types were classified in 1978 by Burnstock (Burnstock, 1978) using several criteria such as rank order of potency of a range of purines in different cell types and the use of selective antagonists. Further distinctions were based on activation of adenylate cyclase and activation of prostaglandin synthesis. Although first established in 1978 the classification has been upheld in a range of different tissues (Burnstock and Buckley, 1985).

- <u>136</u> -

Under Burnstock's classification two receptors are identified. P₁-purinoreceptors exhibit a rank order of potency of Adenosine > AMP > ADP > ATP; they are antagonised by methyl xanthines such as theophyline and caffeine; occupation of these receptors leads to an activation of adenylate cyclase and subsequently increase in intracellular cyclic AMP. The P_2 purinoreceptors, in contrast, have a rank order of potency of ATP > ADP > AMP > Adenosine; no specific antagonists exist although quinidine; apamine; 2,2'-pyridylisatogen and the photo affinity compound arylazidoaminopropionyl ATP have all been cited as P₂-purinoreceptor antagonists (Burnstock and Buckley, 1985); occupation of these receptors have been shown to elicit the production of prostaglandins in various tissues. Both the P1- and P2-purinoreceptor exhibit strict structureactivity relationships: the free -NH2 group at C6 position in the imidazole ring; the ribose moiety and the steric configuration of both purine and ribose rings being essential for activity (Hahn and Olsson, 1975). Like most other receptors which have been pharmacologically characterised in detail e.g. adrenoreceptors (Lands et al., 1967; Langer, S.Z., 1974; Carlsson, 1979; Langer and Shepperson, 1982; McGrath, J.C. 1982) there appears to be reason for further subclassification of both P_1 - and P_2 -purinoreceptors.

The subclassification of P_1 -purinoreceptors is based on the order of potency of a series of adenosine receptor agonists and stereospecificity with respect to the diasteromers. L-N⁶phenylisopropropyladenosine(N⁶-R(-)-L-methyl-2-phenethyl adenosine (L-PlA) and D-N⁶-phenylisopropyladenosine (D-PlA).

- 137 -

In 1980 Londos and colleagues examined the actions of three agonists, namely L-PlA; 5'-N-ethyl carboxanudoadenosine (NECA) and adenosine on the adenylate cyclase system of several cell types (Londos et al., 1980). He found that for stimulation of adenylate cyclase the rank order of potency NECA > Adenosine > PlA was observed while the converse was true for inhibition of adenylate cyclase i.e. rank order of potency was PlA > Adenosine > NECA was observed. The terms R and R, were proposed to describe the receptors mediating the stimulatory and inhibitory actions on adenylate cyclase respectively. Van Calker et al. (1979) similarly concluded from studies on cultured glial cells from mouse brain that adenosine acted at two separate cell surface receptors, which were termed A_1 and A_2 representing inhibition and stimulation of adenylate cyclase respectively. The use of stereoselectivity in the A_1/R_1 and A_2/R_a classification is a more recent addition. In rats, various workers have demonstrated using L-P1A and D-P1A, that the inhibition of lipolysis is an A_1/R_1 receptor mediated action whereas the accumulation of cyclic AMP is thought to be an A_2/R_a action (Trost and Schwabe, 1981; Dunwiddie and Fredholin, 1982). In addition to the A_1/R_i and A_2/R_a sites a "**p**" site has been proposed (Londos) et al., 1983). Unlike the A_1/R_i and A_2/R_a sites, this 'P' site is located on the internal surface of the membrane and when activated by adenosine leads to the inhibition of adenylate cyclase, an effect which is not antagonized by methylxanthines.

The basis for a subclassification of P_2 -purinoreceptors rests, amongst other evidence, upon the divergence of agonist

- 138 -

potency ranking for purinoreceptors located on different cell types e.g. smooth muscle membranes, platelets and mast cells (Burnstock, 1981). The purino-receptor on human platelets, which promotes aggregation, is highly specific for ADP with neither ATP nor AMP being able to activate the receptor. In fact, in human platelets, ATP is a competitive antagonist of the action of ADP (Macfarlane and Mills, 1975). In rat mast cells ADP is unable to mimic the histamine releasing action of ATP (Diamant and Kruger, 1967). The strict specificity of these receptors for either ADP or ATP is in marked contrast with the purinoreceptor located on smooth muscle where both ADP and ATP are equipotent. This lack of homogeneity amongst purinoreceptor.

In a recent review, Gordon (1986) proposed, based on the available data, 4 subclasses of purinoreceptor recognizing ATP, namely the 'T'; 'X'; 'Y' and 'Z' purinoreceptor. The P_{2T} receptor is located on (human) thrombocytes, platelets and megakaryocytes. P_{2X} and P_{2Y} purinoreceptors are located on smooth muscle cells and elicit respectively contraction or relaxation. Finally the P_{2Z} purinoreceptor is apparently present on mast cells and lymphocytes.

Attempts to establish the number of ADP receptors on human platelets, using radiolabelled analogues of ADP, have revealed the existence of one class of receptor with between 400-1200 sites/plate (Macfarlane <u>et al</u>., 1982; Macfarlane <u>et al</u>., 1983). In an earlier study, using the adenine nucleotide affinity analogue 5'-p-flurosulfonylbenzoal adenosine, Bennett <u>et al</u>.

- 139 -

(1978) identified a polypeptide with molecular weight of 100000 daltons as being a putative ADP receptor in human platelets.

A review of the actions of purines on tissues would not be complete without considering the possibility of prior metabolism of these compounds, by ectonucleotidases, prior to eliciting a response. Ectonucleotidases have been found on the surface of endothelial cells; blood platelets and leukocytes (see Pearson for review, 1985). The existence of such enzymes can complicate the interpretation of data using for example ATP as an agonist, since any effect observed with ATP may be due to a direct action of ATP or due to the actions of one of its metabolites. Such possibilities may explain discrepancies in the literature concerning classification of purinoreceptors. Indeed, in guinea pig ileum, ATP has been shown to exert an inhibitory response, partly mediated by its metabolites and partly due to a direct action (Moody et al., 1984). Clearly caution is required when interpreting data obtained while studying the effects of purines on tissues.

Thrombin

Thrombin, a serine protease, has a substantial degree of structural homology with other known serine proteases such as trypsin and chymotrypsin, but differs in its side chain specification (Blow, 1971; Keil, 1971; Magnusson, 1971). Thrombin is formed by the enzymatic cleavage of prothrombin, a component present in plasma, and is one of the most important endogenous stimuli involved in haemostasis - its most important role being the initiation of clot formation by cleaving

- 140 -

fibrinogen into fibrin monomer which may then be converted by factor XIIIa into fibrin polymer. As a platelet agonist, Thrombin is one of the most efficacious agents known, capable of evoking platelet shape change, aggregation, release of dense granule, α granule and lysosomal enzymes and icos anoid synthesis at very low (nM) concentrations. Platelets themselves can catalyse the production of Thrombin in several ways (Walsh, 1974), and it is possible that sufficient Thrombin may be generated within platelet aggregates to trigger the release of platelet granule contents without formation of fibrin. The effects of Thrombin on platelets can be inhibited by Hirudin, an anticoagulant substance derived from leech salivary glands (Markwardt, 1960; Holmsen et al., 1981) and by other serine protease inhibitors such as Leupeptin (Holmsen et al., 1984; Ruggiero and Lapetina, 1985). The interaction of Thrombin with platelets is unusual in that it exhibits properties of both an agonist-receptor equilibrium reaction and an enzyme catalysed reaction (Tollefsen et al., 1974; Detwiler and Feinman, 1973; McGowan and Detwiler, 1983; Shuman et al., 1978; Martin et al., 1975; Detwiler, 1981; Davey and Luscher, 1967).

Thus concentration-response and time-course curves are more consistent with an agonist-receptor equilibrium model (Detwiler and Feinman, 1973; Martin <u>et al.</u>, 1975). The catalytic activity of Thrombin has been demonstrated to be essential for platelet activation i.e. blocking the active site serine inhibits Thrombin-induced platelet activation (Davey and Luscher, 1967). However, the finding that active site blocked Thrombin competes with native Thrombin for binding sites to platelets, but does

- 141 -

not inhibit Thrombin-induced aggregation of platelets (Philips, 1974; Tollefsen <u>et al.</u>, 1974) suggests that all attempts to gain insight into the Thrombin receptor through analysis of such receptor binding data is compromised by the fact that the detected binding sites may not be identical with the functional receptor.

The exact mechanism of action of Thrombin in inducing platelet activation thus remains enigmatic. However a model of Thrombin-induced activation of platelets incorporating elements of both an agonist-receptor equilibration and enzymecatalysed event seems most likely (Martin <u>et al</u>., 1975; Tollefsen <u>et al</u>., 1974; Detwiler, 1981).

Putative "receptors" or more accurately "binding sites" for Thrombin have been identified. Phillips and Poh Agin (1977) presented evidence that the proteolytic substrate for Thrombin was glycoprotein V, since following addition of ^Thrombin to platelets only glycoprotein V was hydrolysed.

Subsequent studies have shown that the complete removal of glycoprotein V from the platelet surface by chymotrypsin revealed no inhibition of platelet activation (McGowan <u>et al</u>., 1983). While this suggests that hydrolysis of glycoprotein V alone does not constitute the signal event in Thrombin-induced platelet activation, some role for glycoprotein V hydrolysis may exist.

Radioligand binding studies have revealed a high degree of heterogeneous binding, interpreted as evidence for three distinct classes of binding sites: a high affinity, low capacity site (Kd 1 nM, 500 sites/cell); a low affinity, high capacity site

- 142 -

(Kd 100 nM, 5000-50,000 sites/cell and a non-saturable site (Detwiler and McGowan, 1985)). In a separate study, Harmon and Jamieson (1985) estimated that the high affinity site approached a size of approximately 90 K Daltons and suggested, due to its size, that this binding site for thrombin may involve a multimolecular complex of membrane components somewhat analogous to the adenylate cyclase system.

Although no identifiable receptor (substrate) for Thrombin has yet been identified on the platelet surface, it appears that the response to Thrombin, in human platelets, may involve more than one such binding site. Studies by Holmsen et al. (1981) have shown a requirement for the continued presence of active Thrombin for such events as acid hydrolase secretion, arachidonic acid metabolism and phosphatidic acid formation. 0 therconsequences of Thrombin-induced platelet activation such as aggregation, dense-granule secretion and phosphatidyl inositol. metabolism do not have this requirement. These results have been interpreted as being the consequence of either separate receptors or separate effectors mediating the response of Thrombin It is interesting to note that McGowan and Detwiler (1986) have recently presented results suggesting the existence of two separate receptors. Whether they correspond to the observations of Holmsen et al. is, at present, unknown. Finally, perhaps the most convincing evidence for a true receptor for Thrombin on platelets stems from the studies of Ruda and colleagues (1985) who showed that the tripeptide \$C40476 exhibits properties of a partial agonist at the putative Thrombin receptor.

- 143 -

3.1.1 Agonist-induced metabolism of inositol phospholipids

The metabolism of inositol phospholipids in rat platelets was monitored using $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ (30 µCi/ml) prelabelled rat platelets. Within approximately 90 min incorporation of $\begin{bmatrix} 32 \\ P \end{bmatrix} P_i$ into PtdA was approaching steady-state conditions, whereas incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ into PtdIns was still increasing at this stage (Figure 3 in Section 2). Due to the relatively high turnover of the 4- and 5-phosphate moieties of PtdIns(4)P and $PtdIns(4,5)P_2$, incorporation of $\begin{bmatrix} 3^2 p \end{bmatrix} P_i$ into these lipids was assumed to approach steady state by 90 mins. Thus a time point of 90 min was used to prelabel rat platelets with $\begin{bmatrix} 3^2 \mathbf{p} \end{bmatrix} \mathbf{P}_i$ in all subsequent experiments monitoring the metabolism of inositol phospholipids. Under these labelling conditions, the basal levels of $\begin{bmatrix} 3^2 P \end{bmatrix} f_1$ prelabelled inositol phospholipids routinely monitored were 112 995 ± 52 248 dpm; 36 378 ± 6384 dpm, 81 459 ± 40 272 dpm and 3648 ± 468 dpm (mean ± SD) for $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂; $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4)P; $\begin{pmatrix} 3^2 P \end{pmatrix}$ PtdIns and PtdA respectively. The following sections outline the effects of ADP on inositol phospholipid metabolism in rat platelets. The effects of Thrombin on inositol phospholipid metabolism are discussed in Section 3.1.4.

3.1.1.1 ADP-induced platelet aggregation

In order to establish that rat platelets prepared under radiolabelling conditions are still responsive, agonist-induced platelet aggregation was monitored. Figure 9 depicts the response obtained while monitoring platelet aggregation induced by $ADP(1-100 \mu M)$. Aggregation is preceded by shape change - lasting approximately 10-15s - followed by a concentration-dependent



and the second second second second

Figure 9: Aggregation of washed rat platelets in response to ADP.

ේදුම අංකාරයේ කත

in the state where the content is

a a strandska state serve sa s

Plasma free suspensions of platelets (0.45 ml) prepared as described in the Methods were exposed to ADP at the concentrations indicated and aggregation was monitored photometrically with continuous stirring. primary aggregation response. The addition of $CaCl_2$ (extracellular calcium concentration 21 mM) 60s before ADP addition facilitated aggregation.

3.1.1.2 Effect of ADP on platelet inositol phospholipid metabolism : Time course

If the metabolism of inositol phospholipids following agonist-receptor interaction is a determinant of cellular activation in rat platelets then the time-course of such an effect should be consistent with the rapidity of the cellular response. Figure 10 depicts the time-course of ADP-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation. In the absence of agonist the levels of [32]PtdA do not change significantly throughout the course of the experiment. No changes in levels of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdIns were observed, either in the presence of absence of agonist (results not shown). ADP (10 μ M) induced a rapid increase in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA which was maximal (43% above basal levels in experiment shown) within 10 sec of agonist addition. The percentage increase, above basal levels, in a number of experiments ranged from 43 - 240%. After the initial peak, levels in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA declined towards basal. The lack of change in levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns relative to changes observed in levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA may be explained by the presence of a large metabolic pool of PtdIns, changes in the size of which - consequent of agonistreceptor interaction - would be difficult to detect. Conversely, under resting conditions, the size of the PtdA metabolic pool is very small and thus any changes to its size, consequent upon agonist-receptor interaction, would be readily detectable.





3.1.1.3 Effect of ADP on platelet inositol phospholipid

metabolism: Concentration-response relationship

Having established the time-course of ADP-induced inositol phospholipid metabolism, the concentration-response relationship for ADP-induced changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA was examined. The timepoint of 30 sec was used since it proved more convenient to use and $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA levels were still elevated at this time point. Figure 11 depicts a typical example of the concentrationresponse relationship of ADP-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation. The threshold for stimulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation was < 0.1 μ M; with maximal stimulation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA formation occurring at approximately 10 µM. In this experiment the maximal stimulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation was approximately 2.3 fold of control, the levels varying between 2-3 fold in different experiments. The EC₅₀ value was 0.9 \pm 0.2 μ M (mean \pm SEM, n = 4). Similar experiments have been performed using both rabbit (Lloyd et al., 1972; 1973a) and human (Lloyd et al., 1973b; MacMillan et al., 1984; MacIntyre et al., 1985c; Fisher et al., 1985) platelets. However, whereas similar results as those obtained here are found in experiments using rabbit platelets, several workers (MacMillan et al., 1984; MacIntyre et al., 1985c; Fisher et al., 1985) have been unable to confirm the findings of Lloyd et al. (1973b) who reported that ADP could elicit changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA in human platelets. The reason(s) for these paradoxical findings is unknown but may merely reflect the insensitivity of the assay techniques available to monitor changes in inositol phospholipid metabolism in human platelets. However, of the techniques available to monitor inositol phospholipid metabolism,



Figure 11: Concentration-response relationship for ADP-induced stimulation of [32P] PtdA formation in rat platelets.

0.5 ml samples of $\begin{bmatrix} 3^2 P \end{bmatrix} P_1$ prelabelled platelets prepared as described (Section 2:2:1:2) in Methods were exposed to ADP at the concentrations indicated. Results measured in terms of fold stimulation - basal levels equal to 1. $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA levels were measured at 30s after agonist addition. The results are mean ⁺ SEM of triplicate determinations. The experiment shown is typical of 4 similar experiments. measurement of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA (Irvine <u>et al</u>., 1983), in addition to measurement of inositol phosphates (Fisher <u>et al</u>., 1985), appear to be the most sensitive.

3.1.1.4 Inhibition of ADP-induced changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ phosphatidate

The most obvious site at which to inhibit an agonist effect is at its receptor. Several receptor antagonists acting specifically at the human platelet ADP receptor are known e.g. 2-methylthio-AMP, 2-methylthio ATP and 2-chloro-ATP (Cusack <u>et al.</u>, 1985). However these were not available in sufficient quantity to test their specificity and use as potential antagonists at receptors recognizing ADP in rat platelets. An alternative mechanism by which one can inhibit agonist-induced platelet activation is to elevate cytoplasmic levels of cyclic AMP (see Section 1.3.1.2). Consequently, where no specific receptor antagonist was available, agents elevating cytoplasmic cyclic AMP levels were used throughout this study to demonstrate inhibition of agonist-induced platelet activation.

3.1.1.4.1 Effect of PGE

PGE₁ has been demonstrated to elevate rat platelet cyclic AMP levels and is a potent inhibitor of platelet (aggregation) activation (Michel et al., 1976). The effect of PGE₁ on ADPinduced inositol phospholipid metabolism was examined. Figure 12 demonstrates the effect of PGE₁ on submaximal stimulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA by ADP. Prior exposure to PGE₁ (10 µM) inhibited the subsequent response to ADP (1 µM) by 30 \pm 3% (mean \pm range, n = 2). It appears therefore that in rat platelets cyclic AMP dependent processes exist which can inhibit ADP-induced metabolism of inositol phospholipids.



Figure 12: Effect of PGE, on ADP-induced [32p] PtdA formation in rat platelets.

0.5 ml samples of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ P prelabelled platelets prepared as described (Section 2:2:1:2) in Methods were exposed to 1 μ M ADP in the absence or presence of PGE as indicated. PGE was added 60s prior to ADP and the levels of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdA measured 30s thereafter. Results measured in terms of fold stimulation - basal levels equal to 1. The experiment shown is typical of 2 similar experiments.

3.1.1.5 Effect of ADP on the polyphosphoinositides

In Section 3.1.1.3 it was demonstrated that ADP stimulates inositol phospholipid metabolism as indicated by the accumulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA. Although agonist-induced accumulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA remains a sensitive index of agonist-induced metabolism of inositol phospholipids (Irvine <u>et al</u>, 1983) $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA could, theoretically, be derived from the phosphorylation of 1,2-diacylglycerol formed as a consequence of phospholipase C activity on any one of the inositol phospholipids i.e. PtdIns (4,5)P₂; PtdIns(4)P or PtdIns, 1,2-diacylglycerol may also arise from the deacylation of triglyceride. Consequently agonistinduced accumulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA gives no indication of the initial event occurring subsequent to agonist-receptor inter-Michell's original hypothesis (1975) was that cell action. surface receptors were coupled to calcium gating via stimulation of PtdIns turnover and that agonist induced metabolism of PtdIns was amongst the initial events triggered by agonist-receptor However, Kirk et al. (1981a) demonstrated, in interaction. rat hepatocytes, that the metabolism of $PtdIns(4,5)P_{2}$ preceded that of PtdIns and was the event initiated by agonist occupation of Ca^{2+} -mobilizing receptors, such as the vasopressin, V_1 receptor in hepatocytes. Consequently they proposed that the initial consequence of receptor occupation was in fact the metabolism of PtdIns(4,5)P₂ (Michell <u>et al</u>., 1981). Data is now available from a variety of different cell types, including human platelets, indicating that the initial events occurring subsequent to agonist-receptor interaction 'do in fact include a rapid decrease in levels of $PtdIns(4,5)P_2$ and a parallel

increase in its water soluble metabolite, $Ins(1,4,5)P_3$ (see Berridge, 1984; Berridge and Irvine, 1984; Farese, 1984; for reviews and section 1.3.1.5). Thus in addition to monitoring the accumulation of 'PtdA' as an indice of agonist-receptor interaction, one could also monitor the decrease in levels of $PtdIns(4,5)P_2$ or the accumulation of $Ins(1,4,5)P_3$ (see Section However, a limitation to the use of the latter 2.2.9.1). technique in platelets (see Section 2.2.9.1.2) lies in the apparent slow uptake of $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol into the inositol phospholipids of platelets including human and rat platelets. Preliminary results for the latter cell type indicated that when incubated with $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol (20 μ Ci/10⁹ cells), basal levels of inositol phosphates measured were 111 - 19 dpm/ 2×10^8 cells, $37 \pm 8 \text{ dpm}/2 \times 10^8$ cells and $21 \pm 6 \text{ dpm}/2 \times 10^8$ cells for $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Ins(1,4,5)P₃; $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Ins(1,4)P₂ and $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Ins 1P respectively. Due to this inadequate rate of $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol incorporation into rat platelets and the prohibitively high cost of this label, studies designed to monitor the accumulation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol phosphates, in rat platelets prelabelled with $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol, subsequent to agonist addition were not performed. However, studies designed to monitor the metabolism of the polyphosphoinositides - in rat platelets prelabelled with $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ - subsequent to agonist addition did prove successful and the results are discussed below.

3.2.1.5.1 Effect of ADP on rat platelet polyphoinositide metabolism: Time course

Figure 13 depicts the time-course of ADP-induced metabolism of $\begin{pmatrix} 3^2 P \end{pmatrix}$ PtdIns(4,5)P. Within 5 sec ADP (10 µM) induced an

- 154 -





0.5 ml samples of $\begin{bmatrix} 3^2 P \end{bmatrix} P_1$ prelabelled platelets prepared as described (Section 2:2:1:2) in Methods were exposed to 10 μ M ADP and the levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂ measured at the times indicated after agonist addition. The results are mean values $\stackrel{+}{=}$ SD of triplicate determinations expressed as a percentage of zero time control; 100% $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂ 112 995 $\stackrel{+}{=}$ 52 248 dpm. The experiment shown is typical of 3 similar experiments. approximate 12% loss in $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdIns(4,5)P₂ (10.8 ± 1.3, mean ± range, n = 3). No changes in either $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdIns(4)P or $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdIns are observed throughout the course of the experiment (data not shown). At the longer time points (> 60s) there is a slight, but insignificant, increase in $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdIns(4,5)P₂ levels above basal. These results are similar to those obtained in rabbit platelets using ADP (Vickers <u>et al</u>., confirming the concept that amongst the initial events initiated by agonist-receptor interaction is the metabolism of polyphosphoinositides.

3.1.2 <u>ADP-induced changes in cytosolic free calcium</u>, $\begin{bmatrix} Ca^{2+j} \end{bmatrix}_{i}$ 3.1.2.1 ADP-induced platelet aggregation

To ensure that platelets prepared under Quin 2 loading conditions maintained their functional responsiveness, platelet aggregation in Quin 2 loaded cells was monitored. Quin 2 loaded cells retain their characteristic disc shape, as monitored by the swirling (schlieren) observed when the cells are shaken in bright light prior to stimulation. Figure 14 demonstrates the aggregatory response obtained with ADP (1-100 μ M). Comparison with unloaded or vehicle (DMSO) treated platelets reveals that Quin 2-loaded platelets exhibit pronounced shape change only at the lower agonist concentration (1 µM) tested. Additionally the aggregatory response observed at this concentration appears to be somewhat enhanced compared to the response obtained using the same agonist concentration in unloaded or vehicle treated platelets. Disruption of the Ca^{2+} homeostasis by Quin 2 may, in some manner, account for this response.



Figure 14: Aggregation of washed rat platelets in response to ADP.

Plasma free suspensions of platelets (0.45 ml) preincubated with either Quin 2 AM (extreme right) vehicle (DMSO) (middle) or unloaded (extreme left) were exposed to ADP at the concentrations indicated and aggregation was monitored photometrically with continuous stirring.

> . بد رد

化二溴化基合物酶 紧张机器装饰机 建化合金

an Antoireire

3.1.2.2 Effect of ADP on rat platelet cytosolic free calcium, [Ca²]

Subsequent to agonist-receptor interaction, changes in cytosolic free calcium are believed to mediate cellular activation in a number of different cell types (Sections 1.3.1.3 -1.3.1.8). For this to be true of rat platelets such changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$ should be consistent with the time-course of known platelet functional responses. Figure 15a depicts the observed changes in $\left[Ca^{2+} \right]$; (monitored as changes in Quin 2 fluorescence) in rat platelets following exposure to ADP (0.03 - 10 μ M). Α rapid and concentration-dependent elevation in $\left[\operatorname{Ca}^{2+}\right]_{i}$ is Significant elevation above basal levels being detected evident. within 10 sec - in keeping with the time-course of plateletfunctional responses. With increasing agonist concentration, the rate of increase and decline of response is visibly quicker. Such responses may represent a mechanism by which the cell regulates its function - by terminating its response more quickly in the presence of a large concentration of stimulant. The processes responsible for terminating the elevation of $[Ca^{2+}]_i$ under normal conditions are not fully understood, but may involve the action of protein kinase C (see Section 1.3.1.8.3 plus Drummond and MacIntyre, 1985). Figure 15b depicts the cumulative concentration-response curve for ADP-induced elevation in $[Ca^{2+}]$, From basal levels of approximately 70 nM, maximal elevation in $\left[\operatorname{Ca}^{2+}\right]_{i}$ of approximately 1200 nM above basal levels is detected at a concentration of 3 μ M. The EC₅₀ value was $0.63 \pm 0.21 \mu M$ (mec \pm SEM, n = 3-7). Similar studies carried out in human platelets (Hallam et al., 1983; Hallam and Rink, 1985) also show a concentration-dependent elevation in $\left[Ca^{2+j}\right]_{i}$


Figure 15a: ADP-induced elevation of cytosolic free calcium $\left[\operatorname{Ca}^{2+}\right]_i$ in rat platelet

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described in Section 2:2:1:3 were incubated with ADP at the concentrations indicated. Changes in Ca²⁺ were estimated from the observed changes in dye fluorescence (Section 2:2:4). The traces above are from a single representative experiment typical of 5 others.

- 158 -



Figure 15b: Concentration-response relationship for ADP-induced elevation of cytosolic free calcium [Ca²⁺], in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with ADP at the concentrations indicated. Results shown are cumulative observations from several experiments (n = 7) and are expressed as mean values - SEM, above basal levels.

- 159 -

after exposure to ADP. The unusual phenomenon of ADP eliciting the elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ in human platelets in the absence of effects on inositol phospholipid metabolism suggests that, unlike the situation in rat platelets, the link between metabolism of inositol phospholipids and calcium gating in human platelets, using this particular agonist, is more tenuous.

Studies carried out in calcium free buffer in the presence of 2mM EGTA (to chelate residual Ca^{2+}) showed that the majority of the rise in $\left[Ca^{2+}\right]_{i}$ detected after ADP addition could be attributed to an influx of extracellular Ca^{2+} , Figure 16. This is similar to the situation observed in human platelets where it has been demonstrated that approximately 80% of the increase in $\left[Ca^{2+}\right]_{i}$ is due to Ca^{2+} influx (Rink <u>et al.</u>, 1982; MacIntyre <u>et al.</u>, 1985c).

Observations in other tissues of the rat e.g. hepatocytes have shown a similar concentration-dependent increase in $[Ca^{2+}]_{i}$ when exposed to ADP. The EC₅₀ for ADP-induced elevation of $[Ca^{2+}]_{i}$ in hepatocytes was 0.3 µM which is lower than that observed for the same event in rat platelets (Charest <u>et al</u>., 1985a). This indicates that ADP either has a lower affinity for the platelet receptor than hepatocyte ADP receptors or that there are different efficiencies in receptor/effector coupling in different cell types.

3.1.2.3 Inhibition of ADP-induced changes in $\left[Ca^{2+} \right]_{i}$

Use of agents elevating cytoplasmic levels of cyclic AMP were examined in an attempt to demonstrate inhibition of ADPinduced $\left[\operatorname{Ca}^{2+}\right]_{i}$ elevation in rat platelets. Figure 17 demonstrate



Figure 16: ADP-induced elevation of cytosolic free calcium in the presence and absence of extracellular free calcium in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with ADP at the concentrations indicated, either in the presence or absence of extracellular free calcium. Changes in $LCa^{2+}J_i$, were estimated from the observed changes in dye fluorescence. Above represents typical responses from a single experiment, representative of 3 others.



Figure 17: Effect of PGE₁ on ADP-induced elevation of cytosolic free calcium Ca²⁺ i in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were exposed to ADP, at the indicated concentrations, in the presence or absence of PGE₁ as indicated. PGE₁ (10μ M) was added 60s prior to ADP and changes in fluorescence monitored thereafter were converted to changes in LCa^{2+} . The above responses are obtained from a single experiment and are representative of 2 similar experiments.

n se an ingele and se and s

that, in the presence of PGE_1 (10 μM) responses to a maximal (3 $\mu M)$ and submaximal (0.3 $\mu M)$ concentration of ADP were inhibited by 73 \pm 6% and 100 \pm 0% (mean \pm range, n = 2) respectively. In the presence of PGE1, the response to ADP (3 μ M) was notably slowed as well as attenuated. These results are suggestive of a cyclic AMP-dependent process, existing in rat platelets, acting to inhibit agonist-induced changes in Ca²⁺ . Kaser-Glanzman <u>et al</u>. (1977; 1979) have demonstrated that cyclic AMP stimulates the active sequestration of Ca^{2+} into purified platelet intracellular membranes corresponding to the dense tubular system. However, Menashi et al. (1982) reported that they were unable to demonstrate any effect of dibutyrlcyclic AMP on Ca^{2+} accumulation into the dense tubular Nevertheless, this effect of cyclic AMP may represent system. one of the possible mechanisms by which it can inhibit agonistinduced platelet activation (see Section 1.3.1.2).

3.1.3 <u>Characterization of the receptors recognizing ADP in</u> rat platelets

Having demonstrated that ADP is capable of elicting the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA and elevating $\begin{bmatrix} Ca^{2+} \end{bmatrix}_{i}$ in rat platelets, I next attempted to characterise the nature of the receptor recognizing ADP in this cell.

Receptors may be characterised by several pharmacological techniques: - 1) Comparison of the rank order of potency of a series of similar chemical compounds, tested in a range of tissues; 2) The use of selective antagonists. (Tallarida and Jacob, 1979); 3) Radioligand binding studies, where only the

- 103 -

binding shown to be saturable, reversible and pharmacologically specific as well as consistent with the time-course and concentration-dependency of the response under investigation, is deemed to reflect receptor occupancy (Birnbaumer <u>et al</u>., 1974) and 4) Analyses of transduction processes. Transduction processes exist to transfer information received on the external surface of the cell to the inside of the cell in order to elicit a cellular response. Receptors may then be classified on the nature of this transduction process (see Section 1.3.1 plus Berridge, 1980; 1981).

From previous sections the nature of the transduction process in rat platelets, activated by ADP-receptor occupancy, would appear to involve the inositol phospholipid signal pathway (ergo metabolism of inositol phospholipids and elevation of $\left[\operatorname{Ca}^{2+}\right]_{1}$. The possibility exists, but is not proven, that these effects may be causally interrelated.

3.1.3.1 Effect of ATP, ADP, AMP and Adenosine on rat platelet inositol phospholipid metabolism: Concentration-response relationship

Figure 18 depicts the effects of ATP-, ADP-, AMP- and Adenosine-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation in rat platelets. Both ADP(10 nM - 100 μ M) and ATP (1 μ M - 1 mM) but not AMP or Adenosine when tested at a single concentration of 100 μ M

- 164 -



<u>Figure 18:</u> <u>Concentration-response relationship for agonist-induced</u> <u>stimulation of [³²P]PtdA formation</u> in rat platelets.

0.5 ml samples of $\begin{bmatrix} 3^2 p \end{bmatrix} P$, prelabelled platelets prepared as described (Section 2:2:1:2) were exposed to ADP (Δ); ATP (0); AMP (Ω) or Adenosine (**1**) at the concentrations indicated. Results are measured in terms of fold stimulation - basal levels equal to 1. $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA levels were measured 30s after agonist addition. The results are means [±] SD of triplicate determinations. The data shown for ATP and ADP is taken from individual experiments representative of 3-4 similar experiments.

an an an an an a

elicited, in a concentration-dependent manner, the formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA. The threshold concentration for stimulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation, using ADP as the agonist, was < 0.1 μ M and < 1 μ M using ATP. In the results shown, the maximal extent of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation was approximately 2.3 fold of control (range 2-3 fold in different experiments) using ADP as the agonist and 1.6 fold of control (range 1.4-2.0 in different experiments) using ATP as the agonist. The EC₅₀ values for ADP and ATP are 0.9 \pm 0.2 μ M and 3.8 \pm 1 μ M (mean \pm SEM, n = 3-4) respectively. In rat platelets it appears that ADP is more potent and efficacious, when compared to ATP, at eliciting the formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA. In contrast AMP and Adenosine are inactive.

Numerous studies have shown the presence of ectonucleotidases, capable of metabolising purines, on the surface of various cells, including platelets (Pearson, 1985). In order to negate the possibility, that the observed responses of ATP are a result of the actions of its metabolites - subsequent to its hydrolyses - rather than due to a direct action of ATP, studies were carried out comparing the effects of ATP and the non-hydrolysable ATP analogue, Adenosine-5'-3 -thiotriphosphate in the same experiment. Figure 19 demonstrates the results obtained from a single experiment of the effects of ATP and Adenosine 5'-3-thiotriphosphate , ATPOS, on inositol phospholipid metabolism in rat platelets. Both ATP and ATP χ S (1 μ M - 1 mM) elicit, in a concentration-dependent manner, the formation of Maximal fold stimulation of ATP-induced $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdA ³²_P PtdA. formation is 2 (range 1.4 - 2 in different experiments) and for

- 167 -

ATP \S S-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation it is 1.75 (range 1.6 - 1.75 in different experiments). Calculated EC₅₀ values are 3.1 \pm 0.98 μ M and 3.1 \pm 0.92 μ M (mean \pm SEM, n = 3) for ATP and ATP \S S respectively. No statistical difference exists between these two values and from the degree of fold stimulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation detected using either ATP or ATP \S S, it appears that ATP \S S may be slightly less efficacious when compared to ATP at eliciting the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA. These results are suggestive that, when exposed to rat platelets, both ATP and ATP \S S behave in a similar manner and lends credence to the suggestion that responses elicited by ATP on rat platelets are primarily a result of ATP interacting at given receptor(s) and not due to the products of its metabolism.

3.1.3.2 Effect of ATP, ADP, AMP and Adenosine on rat platelet cytosolic free calcium, [Ca²⁺];

Figure 20 depicts the effects of ATP-; ADP-, AMP- and Adenosine-induced elevation in cytosolic free calcium. Both ADP (30 nM - 10 μ M) and ATP (3 μ M - 1 mM) but not AMP or Adenosin (1 μ M - 100 μ M) elicited, in a concentration-dependent manner, the elevation of $[Ca^{2+}]_i$ from a basal value of approximately 70 nM to approximately 1300 nM and 550 nM for ADP and ATP respectively. The EC₅₀ values are 0.63 \pm 0.2 μ M and 23 \pm 5 μ M (mean \pm SEM), n = 3-7) for ADP and ATP respectively. Comparison of Figure 15a, using ADP as the agonist, with Figure 21, demonstrating the effects of ATP on cytoplasmic calcium levels, will reveal no discernible difference in the nature of the response to either of the two agonists - suggestive that the observed response to ATP in rat platelets is primarily due to



Figure 19: <u>Concentration-response relationship for ATP- and ATPVS-induced</u> <u>stimulation of (³²P) PtdA formation</u> in rat platelets.

0.5 ml samples of $\begin{bmatrix} 3^2 p \end{bmatrix} P_1$ prelabelled platelets prepared as described (Section 2:2:1:2) were exposed to ATP (open symbols) or ATP SX(closed symbols) at the concentrations indicated. Results are measured in terms of fold stimulation - basal levels equal to 1. $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA levels were measured 30s after agonist addition. The results are means - SEM of triplicate determinations. The experiment shown is typical of 3 similar experiments.



Figure 20: Concentration-response relationship for agonist-induced changes in cytosolic free calcium [Ca²⁺], in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with ADP (Δ); ATP (0); AMP([]) or Adenosine (\blacksquare) at the concentrations indicated. Results measured in terms of increase in $[Ca^{2+}]_i$ above basal levels (means - SEM). The data shown are cumulative results from 3-7 experiments.



Figure 21: ATP-induced elevation of cytosolic free calcium (Ca²⁺)_i in rat platele

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with ATP at the concentrations indicated. Changes in $[Ca^{2+j}]$, were estimated from the observed changes in dye fluorescence (Section 2:2:4). Above depicts typical responses from a single experiment representative of 4 others.

a direct effect of ATP itself and not as a result of its metabolites subsequent to its hydrolysis. To substantiate this point, studies were performed to compare the effects of ATP and ATPXS, in the same experiment, at eliciting the elevation of $\left[\operatorname{Ca}^{2+}\right]_{1}$. The results are depicted in Figure 22. EC₅₀ values are 22 $\stackrel{+}{=}$ 2 μ M and 17 $\stackrel{+}{=}$ 1.6 μ M (mean $\stackrel{+}{=}$ SEM, n = 2) for ATP and ATPXS respectively. There was no statistical difference between EC_{50} values or maximal response when either ATP or ATPYS was used as the agonist. These results extend those obtained while monitoring changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA induced by ATP/ATP S and indicate that ATP S behaves in a similar fashion to ATP with respect to elevating [Ca²⁺]. Hence responses elicited by addition of ATP appear to be due primarily to the direct action of ATP and not to metabolites generated following ATP hydrolysis. Figure 23 depicts the actual responses obtained in a representativ experiment; they demonstrate the similarity in response of ATP and ATPYS.

Figure 24 demonstrates the lack of response to AMP and Adenosine in rat platelets with respect to changes in cytosolic free calcium.

In conclusion, with regard to the characterisation of the nature of the receptor recognizing purines on rat platelets, the data obtained while monitoring $\begin{pmatrix} 3^2P \end{pmatrix}$ PtdA formation and elevation of $\begin{pmatrix} Ca^{2+} \end{pmatrix}_{i}$ indicate that rat platelets contain receptors acted upon by ADP and ATP but not AMP or Adenosine.

- 171 -



Figure 22: Concentration-response relationship for ATP- and ATPY S induced elevation of cytosolic free calcium (Ca²⁺); in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with ATP (open symbols) or ATP/S (closed symbols) at the concentrations indicated. Results are expressed in terms of increase in Ca^{2+} , above basal levels, (means $\stackrel{+}{-}$ SEM). The data shown is typical of that obtained in 2 other experiments.



Figure 23: ATP- and ATPVS-induced elevation of cytosolic free calcium, $C_{Ca}^{2+j}_{i}$ in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with ATP (upper trace) or ATPYS (lower trace) at the concentrations indicated. Changes in LCa^{2+J}, were estimated from the observed changes in dye fluorescence (Section 2:2:4). The responses shown are from a single experiment representative of 2 others.

- 173 -



Figure 24: $\underline{\text{AMP}}_{i}$ and Adenosine-induced elevation of cytosolic free calcium, $\underline{\text{CCa}}^2 \textbf{J}_{i}$, in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with Adenosine (upper trace) or AMP (lower trace) at the concentrations indicated. Above represents typical responses from one of four similar experiments.

a tanàna ara-

人名法马卡伦 化化化学 化基苯化合物 网络拉斯林 网络小小小小小小小小小小小小小小小小

3.1.4 Thrombin-induced metabolism of inositol phospholipids

The effect of Thrombin on inositol phospholipid metabolism in rat platelets was monitored in cells prelabelled with $\begin{bmatrix} 3^2 P \\ P \end{bmatrix} P_i$. The basal levels of each of the inositol phospholipid species routinely monitored are as described in Section 3.1.1.

3.1.4.1 Thrombin-induced platelet aggregation

Platelets prepared under radiolabelling conditions were still functional with respect to aggregation induced by Thrombin (5-50 nM) as demonstrated in Figure 25. The aggregation response to Thrombin is concentration-dependent and irreversible and is preceded by a shape change lasting 10-15s. Unlike ADPinduced platelet aggregation, Thrombin-induced platelet aggregation was evident in the absence of added Ca²⁺ ions.

3.1.4.2 Effect of Thrombin on platelet inositol phospholipid metabolism: Time course

In rat platelets Thrombin (50 nM, 1 U/ml) induced a rapid increase in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation (Figure 26). The peak response, which represented a 770% increase above basal levels in this experiment (range 770-890% in other experiments) occurred within 30-60s of agonist addition. Compared to ADP-induced formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA these results suggest that Thrombin is far more efficacious at stimulating the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA. In the absence of agonist, the levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA did not change significantly throughout the course of the experiment. After the initial peak, levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA returned slowly towards basal levels. However, unlike ADP-induced formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA, levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA remain elevated - 459% above





Plasma free suspensions of platelets (0.45 ml) prepared as described in Methods were exposed to Thrombin at the concentrations indicated and aggregation was monitored photometrically with continuous stirring.



Figure 26: <u>Time-course of Thrombin-induced changes in [32p] PtdA in rat</u> <u>platelets</u>. 0.5 ml samples of [32p]P₁ prelabelled platelets prepared as described (Section 2:2:1:2) were exposed to Thrombin. Results are expressed in terms of fold stimulation of the basal levels (= 1). The levels of C³²p] PtdA were measured at the times indicated after agonist addition. The experiment shown is typical of three similar experiments. basal in this particular experiment by 4 min. This effect may be reflective of the more efficacious nature of Thrombin i.e., being able to elicit and maintain the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA over a longer period of time compared to ADP. Throughout the course of the experiment no significant changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns were observed (data not shown). The lack of changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns vere observed (data not shown). The lack of changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns relative to changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA is unusual in that, compared to ADP-induced changes in inositol phospholipid metabolism, Thrombin is much more efficacious and one may have expected Thrombin to have an effect on the $\begin{bmatrix} 3^2 P \end{bmatrix}$ -labelled pool of PtdIns.

3.1.4.3 Effect of Thrombin on platelet inositol phospholipid metabolism: Concentration-response relationship

Having established the time course of Thrombin-induced inositol phospholipid metabolism, the concentration-response relationship for Thrombin-induced $\begin{bmatrix} 3^2p \end{bmatrix}$ PtdA formation at 30s was investigated, Figure 27. The threshold concentration for stimulation of $\begin{bmatrix} 3^2p \end{bmatrix}$ PtdA formation was <15 nM and maximal effects were observed at 150 \cap M. The calculated EC₅₀ is 31 \pm 0.25 nM (mean \pm SEM, n = 2-3). In the experiment shown, the maximal stimulation of $\begin{bmatrix} 3^2p \end{bmatrix}$ PtdA formation was approximately 13 fold of control (the levels ranged between 7 and 13 fold in different experiments). When similar experiments are performed using either horse (Lapetina <u>et al</u>., 1981), rabbit (Lloyd and Mustard, 1974) or human platelets (Broekman <u>et al</u>., 1980; Holmsen <u>et al</u>., 1981) similar results are observed. The present findings confirm and extend those of Koutouzov <u>et al</u>. (1985) and Kito et al (1985) who showed a comparable increase in





0.5 ml samples of $\begin{pmatrix} 32 \\ p \end{pmatrix}_{P}$ prelabelled platelets prepared as described (Section 2:2:1:2) were exposed to ADP at the concentrations indicated. Results measured in terms of fold stimulation - basal levels equal to 1. $\begin{pmatrix} 32 \\ p \end{pmatrix}$ PtdA levels were measured at 30s after agonist addition. The results are means [±] SEM of triplicate determinations. The experiment shown is typical of 3 similar experiments. $[3^{2}P]$ PtdA formation in rat platelets following Thrombin addition. Like the studies of Kito <u>et al</u>. (1975), no changes in levels of $[3^{2}P]$ PtdIns, following exposure of platelets to Thrombin were observed.

3.1.4.4 Inhibition of Thrombin-induced changes in [32p] Phosphatidate

Thrombin-induced platelet activation exhibits properties of both a receptor operated and enzyme-catalysed reaction (see Section 3) and methods designed to inhibit its response in platelets are complicated by this fact. However, there are compounds known to antagonise (although not at the receptor level) the response to Thrombin in platelets. Hirudin, an anticoagulant substance derived from leech salivary glands (Markwardt, 1960) is known as a high affinity inhibitor of Thrombin and can inhibit both the stimulation of platelets by Thrombin and the binding of Thrombin to platelets (Detwiler and Feinman, 1973; Ganguly and Sonnichsen, 1976; Tam and Detwiler, 1978). In human platelets, Holmsen et al. (1981) demonstrated that in the presence of Thrombin, excess (approximately 20 fold) Hirudin, added 10s before Thrombin, can inhibit acid hydrolase secretion, Arachidonic acid metabolism and phosphatidic acid formation. The effects of excess Hirudin on Thrombin-induced changes in 3201 PtdA formation in rat platelets were examined. Figure 28 depicts the effects of excess Hirudin added 10s before a submaximal concentration of Thrombin. Hirudin (8 U/ml) inhibited Thrombin-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation by 75 ± 7.5% (mean ± range, n = 2).



Figure 28: Effect of Hirudin on Thrombin-induced (32p)PtdA formation in rat platelet

lan an an airte a geire an an an an bha a

en el la companya de la companya de

0.5 ml samples of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ P prelabelled platelets prepared as described (Section 2:2:1:2) were exposed to Thrombin (20 nM) in the absence or presence of Hirudin as indicated. Hirudin was added 10s prior to Thrombin and the levels of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdA measured 30s thereafter. Results are expressed as fold stimulation of the basal level (= 1). The experiment shown is typical of 2 similar experiments. 3.1.4.4.1 Effect of PGE

The effects of PGE_1 on Thrombin-induced formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA was examined. Figure 29 depicts the results obtained against 50 nM Thrombin, PGE_1 (10 μ M) inhibited the agonist response by 26 \pm 2.6% (mean \pm range, n = 2). Thus, as for ADP-induced formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA, PGE_1 can also attenuate Thrombin-induced formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA in rat platelets, presumably as a result of activating a cyclic AMP-dependent process.

3.1.4.5 Effect of Thrombin on polyphosphoinositide metabolism: <u>Time course</u>

The effect of Thrombin on polyphosphoinositide metabolism was examined in an attempt to discern which inositol phospholipids is(are) metabolised following receptor occupancy by agonist. As with studies using ADP as the agonist (Figure 13) it appears that the metabolism of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂ is one of the first events occurring subsequent to agonist-receptor interaction. Figure 30 depicts the time-course of Thrombininduced metabolism of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂.Within 5s of agonist addition there is a significant loss in $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdIns(4,5)P₂ (21% in experiment shown, 22 \pm 1.4, mean \pm range, n = 3). The maximal loss of $[3^2P]$ PtdIns(4,5)P₂, 34% in experiment shown $(33 \pm 1.4\%, \text{ mean} \pm \text{ range}, n = 3)$ occurs at 15s. When compared to ADP-induced metabolism of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂ (Figure 13), Thrombin evokes, by far, a greater decrease in $\begin{pmatrix} 3^2 P \end{pmatrix}$ PtdIns(4,5)P₂ - again reflective of the more efficacious nature of Thrombin in rat platelets. Throughout the course of the experiment no significant changes in either $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4)P or $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns



Figure 29: Effect of PGE, on Thrombin-induced [32p] PtdA formation in rat platelet

0.5 ml samples of $(32p)P_1$ prelabelled platelets prepared as described (Section 2:2:1:2) were exposed to Thrombin (50 nM) in the absence or presence of PGE₁ as indicated. PGE₁ was added 60s prior to Thrombin and the levels of (232p) PtdA measured 30s thereafter. Results are expressed as fold stimulation of the basal level (= 1) and are typical of 2 similar experiments.





后,你们就在我们不会在我这样的。"这时,这些情况是一次的,我们还

were observed (data not shown). After 15s there is a slow return towards basal levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂. Similar results are obtained using human platelets (Rendu <u>et al.</u>, 1983; Agranoff <u>et al.</u>, 1983) suggesting that metabolism of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns (4,5)P₂ is one of the earliest events occurring after agonistreceptor interaction in a variety of different cell types including rat platelets.

3.1.5 <u>Thrombin-induced changes in cytosolic free calcium $\left[Ca^{2+}\right]_{i}$ </u> 3.1.5.1 <u>Thrombin-induced platelet aggregation</u>

Platelet aggregation was examined, under Quin 2 loading conditions, in order to demonstrate maintenance of platelet functional responsiveness during Thrombin-induced changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$. Figure 31 demonstrates the aggregatory response obtained with Thrombin (5-50 nM). When compared to unloaded or vehicle treated platelets, Thrombin-induced platelet aggregation was markedly enhanced in Quin 2 loaded platelets. Shape change was only evident at the lowest Thrombin concentrations tested, possibly being masked at the higher concentrations of Thrombin due to the potentiated aggregatory responses. However, the results serve to illustrate that Quin 2 loaded platelets, stimulated with Thrombin, exhibit a concentrationdependent response.

3.1.5.2 Effect of Thrombin on rat platelet cytosolic free calcium [Ca²⁺];

The changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$ in rat platelets subsequent to Thrombin exposure were investigated, in order to elucidate whether or not they were contemporaneous with known platelet



Figure 31: Aggregation of washed rat platelets in response to Thrombin.

Plasma free suspensions of platelets (0.45 ml) preincubated with either Quin 2 AM (extreme right) vehicle (DMSO) (middle) or unloaded (extreme left) were exposed to Thrombin at the concentrations indicated and aggregation was monitored photometrically with continuous stirring.

Live Formal, ni

and a frequencial for the second and a second and a second and a second and the second and the second and the s

functional responses and hence may be involved in platelet Figure 32a depicts the changes in $\left[\operatorname{Ca}^{2+}\right]_{:}$ in rat activation. platelets following addition of Thrombin (0.5 - 15 nM). A rapid and concentration-dependent elevation in $\left[\operatorname{Ca}^{2+1}\right]_{i}$ is observed. Maximal elevation in $\left[Ca^{2+}\right]_{i}$ occurs within 10 sec and as with ADP-induced changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$, with increasing agonist concentration, the rate and decline of response is greater. These results indicate that changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$ occur within the same time scale as platelet functional responsiveness. Figure 32b depicts the cumulative concentration-responsiveness for Thrombin-induced elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$. It is apparent that Thrombin is much more potent and efficacious compared to ADP (Figure 15b) at elevating $\left[\operatorname{Ca}^{2+1}\right]_i$. Maximal elevation in $\left[\operatorname{Ca}^{2+}\right]_{i}$ (above basal level of approximately 70 nM) of approximately 4000 nM is observed at Thrombin concentrations of 15-50 nM. The EC₅₀ value is 7 \pm 0.25 nM (mean \pm SEM, n = 3). The extent of Thrombin-induced elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ was greatly reduced in the absence of external Ca²⁺ (i.e. using cells suspended in Ca²⁺-free buffers containing EGTA (2 mM), Figure 33. This indicates that, as with ADP, the majority of the Thrombin-induced elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ derives via influx of external Ca^{2+} .

Similar concentration-dependent elevation in platelet $[Ca^{2+}]_i$ evoked by Thrombin are obtained when human platelets are used (Zavoico and Feinstein, 1984; Brydon <u>et al.</u>, 1984).

3.1.5.3 <u>Inhibition of Thrombin-induced changes in $\left[\operatorname{Ca}^{2+}\right]_i$ </u> The inhibition of Thrombin-induced changes in $\left[\operatorname{Ca}^{2+}\right]_i$ was investigated using Hirudin and PGE₁.



Figure 32a: Thrombin-induced elevation of cytosolic free calcium (ca^{2+}) Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with Thrombin at the indicated concentrations. Changes in (ca^{2+}) , were estimated from the observed changes in dye fluorescence (Section 2:2:4). Above responses are from a single experiment representative of 4 others.





Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated with Thrombin at the concentrations indicated. Results are cumulative observations from several experiments (means $\stackrel{-}{-}$ SEM) and are expressed as the increase in $(Ca^{2+})_i$ above basal.



Figure 33: Thrombin-induced elevation of cytosolic free calcium in the presence and absence of extracellular free calcium in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were incubated at the concentrations indicated, either in the presence or absence of extracellular free calcium. Changes in $[Ca^{2+}]_{i}$ were estimated from the observed changes in dye fluorescence. Above represents typical responses from a representative experiment.

and the second second

3.1.5.3.1 Effect of Hirudin

When added to platelets before Thrombin, Hirudin inhibited the resultant elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ (Figure 34). In the example shown, 1.25 (µ/ml Hirudin inhibits the elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ induced by a maximal (15 nM) and submaximal (1.5 nM) concentration of Thrombin by 58 \pm 12% and 100 \pm 0% (mean \pm range, n = 2) respectively. As well as attenuating the response to 15 nM Thrombin, the presence of Hirudin also severely slows its development. These results indicate that in addition to inhibiting Thrombin-induced $\left[3^{2}P\right]$ PtdA formation in rat platelets, Hirudin can also inhibit Thrombin-induced elevation of $\left(\operatorname{Ca}^{2+}\right]_{i}$.

3.1.5.3.2 Effect of PGE

The effects of PGE_1 were examined on Thrombin-induced changes in $[Ca^{2+}]_i$. After prior exposure to 10 µM PGE₁, the response to a maximal (15 nM) and submaximal (1.5 nM) concentration of Thrombin was reduced to 76 \pm 0.9% and 100 \pm 0% (mean \pm range, n = 2-3) respectively (Figure 35). In addition to attenuating the response to Thrombin, PGE₁ also severely slowed the development of the response to 15 nM Thrombin. These results indicate that PGE₁-induced elevation of cyclic AMP can result in the inhibition of agonist-induced elevation of $[Ca^{2+}]_i$ in rat platelets.

3.2 <u>RPM</u>

3.2.1 Agonist-induced metabolism of inositol phospholipids

Metabolism of inositol phospholipids in RPM were monitored in cells prelabelled with either $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ (3 µCi/6 x 10⁶ cells) or $\begin{bmatrix} 3 H \end{bmatrix}$ inositol (10 µCi/10⁶ cells). Within approximately 24 hrs,



Figure 34: Effect of Hirudin on Thrombin-induced elevation of cytosolic free calcium [Ca²⁺], in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were exposed to Thrombin, at the concentrations indicated, in the presence or absence of Hirudin (1.25 μ/ml) (H) as indicated. Hirudin was added prior to Thrombin and changes in fluorescence monitored thereafter were converted to changes in $[Ca^{2+}]_i$. Traces shown are from a single experiment representative of 2 others.



Figure 35: Effect of PGE on Thrombin-induced elevation of cytosolic free calcium CCa²⁺, in rat platelets.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:1:3) were exposed to Thrombin, at the concentrations indicated, in the presence or absence of PGE₁ as indicated. PGE₁ (10 μ M) was added 60s prior to Thrombin and changes in fluorescence monitored thereafter and converted to yield changes in CCa^{2+} . The above responses are from a single experiment, typical of 3 others.

the second programs and harden to be a factor of

- 193 -
incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ into PtdA was approaching steady-state conditions, incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix} P_i$ into PtdIns was still increasing at this time point (Figure 5 in Section 2). Due to the relatively high turnover of the 4- and 5-phosphate moieties in PtdIns (4)P and PtdIns(4,5)P₂ (Section 2.2.9.1.1), incorporation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ P_i into these lipids is assumed to approach steady state conditions by 24 hrs. Thus, a preincubation period of 24 hrs was chosen for all experiments utilizing the radiolabel $\begin{bmatrix} 3^2 P \end{bmatrix} P_{I}$. Under the conditions of radiolabelling the basal levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ prelabelled inositol phospholipids routinely measured, were 63 306 ± 4383 dpm; 88512 ± 6678 dpm; 46 755 ± 2721 dpm and 384 ± 48 dpm (mean ± SD) for $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂; $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns (4)P; $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns and $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA respectively. Time course studies following the rate of incorporation of $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol into the inositol phospholipids in RPM were not performed. Instead a period of preincubation with $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol lasting three days was chosen. This duration of preincubation compares favourably with those used with other cultured cells (see Drummond et al., 1984; Berridge et al., 1984; Rebecchi et al., 1983). After the three day incubation period, incorporation of $\begin{bmatrix} 3_H \end{bmatrix}$ inositol into the inositol phospholipids was 370 176 ± 33 416 dpm, 49 940 ± 19 708 dpm and 10 092 ± 3460 dpm (mean ± SEM) for $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂; $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4)P and $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns respectively. These values are indicative that any inositol phosphates formed, consequent upon agonist addition, could be derived from the parent compounds. Basal levels of the inositol phosphates monitored were routinely 352 ± 10 dpm; 352 ± 10 dpm and 2332 ± 20 dpm (mean ± SD) for $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Ins(1,4,5)P₃; $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Ins $(1,4)P_2$ and $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins 1P respectively.

3.2.1.1 Functional Responsiveness in RPM

In any biological study, the demonstration of an agonistinduced functional response helps to validate biochemical responsiveness (e.g. inositol phospholipid metabolism and changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$ induced by the same agonist. Possible parameters which could be monitored in RPM include agonist-induced release of 5HT or adenine nucleotides or morphological changes. Although, in the present studies, I could demonstrate both $\begin{bmatrix} 3_H \end{bmatrix}$ 5HT $(10024 \pm 1384 \text{ dpm}/4 \times 10^5 \text{ cells})$ and $[^{3}\text{H}]$ Adenine (11 192 \pm 10^{5} \text{ cells}) $304 \text{ dpm}/4 \times 10^5 \text{ cells}$) to be associated with the RPM in experiment: designed to monitor agonist-induced release of radiolabel marker from RPM, no radioactivity was found in the high speed supernatent fraction in such studies (results not shown). This would suggest that in RPM, incubated with either $\begin{bmatrix} 3 \\ H \end{bmatrix}$ 5HT or $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Adenine, that ADP and Thrombin failed to elicit a functional response. Studies using mature megakaryocytes have shown both release of 5HT and ATP from storage granules (Fedorko, 1977; Miller, 1983) suggesting that in these immature cells the transduction processes - transferring information to the inside of the cell - may not be fully functional or developed. Studies designed to monitor morphological changes in RPM subsequent to agonist addition were beyond the scope of the present study.

Although no index of functional responsiveness in RPM could be demonstrated, to establish whether or not similar mechanisms possibly underlying cellular reactivity in rat platelets exist in RPM, the effects of agonists on inositol phospholipid metabolism and changes in $\left[\operatorname{Ca}^{2+}\right]_i$ were investigated.

The following sections outline the effect of ADP on inositol phospholipid metabolism in RPM. The effects of Thrombin on inositol phospholipid metabolism are discussed in Section 3.2.4.

3.2.1.2 Effect of ADP on RPM inositol phospholipid metabolism: <u>Time-course</u>

The time course of agonist-induced metabolism of inositol phospholipids was monitored initially. Figure 36 depicts the results obtained from ADP-induced $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA formation in RPM. In the absence of agonist the levels of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA did not change significantly throughout the duration of the experiment. ADP (3 μ M) induced a rapid increase in $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA formation representing a 36% increase above basal levels in the example shown (36 ± 0%, mean ± SEM, n = 2) within 60s of agonist addition. Thereafter levels in $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA decline slightly reaching a plateau level of approximately 31% above basal levels in this particular experiment. No changes in $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdIns levels were detected throughout the duration of the experiment (results not shown).

3.2.1.3 Effect of ADP on RPM inositol phospholipid metabolism: Concentration-response relationship

Having established the time-course of ADP-induced $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA formation the next parameter to be investigated was the concentration-dependence of this response. Figure 37 depicts the results obtained for ADP-induced formation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA monitored 60s after agonist addition to cells. ADP was tested over the concentration range 1-1000 µM. At 1 µM ADP no significant formation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA was observed; a 1.3 fold stimulation occurred at 3 µM (range 1.23 - 1.3 fold in different experiments) which apparently was maximal. The estimated EC₅₀

- 196 -







Figure 37: Concentration-response relationship for ADP-induced stimulation of [32P] PtdA formation in RPM.

0.4 ml samples of $\begin{bmatrix} 3^2 p \end{bmatrix}$ P, prelabelled RPM prepared as described (Section 2:2:2:3) were exposed to ADP at the concentrations indicated for 60s. Results are expressed in terms of fold stimulation of the basal levels (equal to 1). The results are means $\stackrel{+}{\rightarrow}$ SEM of triplicate determinations. The experiment shown is typical of 3 similar experiments. value was 1.6 \pm 0.07 μ M (mean \pm SEM, n = 2-3). These results indicate that, like rat platelets, RPM are capable of metabolising inositol phospholipids, resulting in the formation of PtdA, subsequent to agonist-receptor interaction.

3.2.1.4 Inhibition of ADP-induced changes in [32P] Phosphatidate

As in studies with rat platelets, availability of known antagonists of the effects of ADP in human platelets were restricted. However, PGE₁ has been shown to elevate levels of cyclic AMP in RPM (Section 2.2.9) and studies were carried out using this compound to discern whether or not if a qualitatively similar cyclic AMP-dependent mechanism that acts to inhibit $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation in rat platelets also exists in RPM. Figure 38 depicts the effect of PGE₁ on submaximal stimulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA by ADP. Prior exposure to PGE₁ (10 µM) inhibited ADP-induced formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA by 36 \pm 5% (mean \pm range, n = 2). These results indicate that, as in rat platelets, cyclic AMP-dependent mechanisms exist in RPM to inhibit agonist induced metabolism of inositol phospholipids.

3.2.1.5 Effect of ADP on polyphosphanositide metabolism: Time course

Studies in rat platelets (Section 3.1.1.5.1 and 3.1.4.5) demonstrates that one of the earliest events initiated by receptor occupation by agonist is the metabolism of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns $(4,5)P_2$ resulting in an apparent reduction in the level of this inositol phospholipid. Similar studies were performed using $\begin{bmatrix} 3^2 P \end{bmatrix} P_1$ prelabelled RPM in an attempt to discern whether or not a qualitatively similar rapid metabolism of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns $(4,5)P_2$



Figure 38: Effect of PGE₁ on ADP-induced $\begin{bmatrix} 32_P \end{bmatrix}$ PtdA formation in RPM.

0.4 ml samples of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ P prelabelled RPM prepared as described (Section 2:2:2:3) were exposed to ADP (2 MM) in the presence or absence of PGE as indicated. PGE was added 60s prior to ADP and the levels of $\begin{bmatrix} 32 \\ P \end{bmatrix}$ PtdA measured 60s thereafter. Results are expressed in terms of fold stimulation of the basal (equal to 1). The experiment shown is typical of 2 similar experiments. is evident. The results are summarised in Table 3.

In contrast to the situation in rat platelets (Section 3.1.1.5.1) ADP (3 μ M) does not appear to induce the metabolism of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂ in RPM. No significant changes in the levels of either PtdIns(4,5)P2, PtdIns(4)P or PtdIns are observed at any time during the experiment; however significant stimulation of "PtdA" formation at 10, 30, 60 and 300s suggests that the inositol phospholipid cycle has been perturbed. This lack of effect on metabolism of polyphosphoinositides may be a reflection of the low efficacy exhibited by ADP in RPM. Reference to Sections 3.2.1.2 and 3.2.1.3 will demonstrate that in this cell the maximal stimulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation, induced by ADP, ranges from between 30-40% above basal levels. Such a weak effect may stimulate the metabolism of only a small 'pool' of PtdIns(4,5)P2 which may be rapidly replenished. Additionally, the errors inherent in using this techniqu'e mean that small changes in levels of any of the inositol phospholipids monitored will be difficult to detect. However, an interesting point to note is that significant amounts of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA are formed at very early time points during these experiments. This at variance with those results obtained while monitoring agonist-induced changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation (Section 3.2.1.2). Different, and more thorough, phospholipid extraction procedures compared to those utilized to obtain results in Section 3.2.1.2 are utilized in the present experiments. It is quite possible that any slight formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA will be more readily detected using the present extraction procedure compared to any other method of extraction.

Table 3: Time-course of ADP-induced changes in the polyphosphoinositides in RPM.

Phospholipid	E.	ADP in	<u>d (s)</u> 60	300	
	,	10			
Ptd Ins 4, 5P2	1.04 <u>+</u> 0.05	0.99 <u>+</u> 0.18	1.02 <u>+</u> 0.04	1.06 ± 0.06	0.97 <u>+</u> 0.02
Ptd Ins 4P	0.98 <u>+</u> 0.05	0.95 <u>+</u> 0.02	0.98 <u>+</u> 0.04	1.03 <u>+</u> 0.05	1.03 ± 0.08
Ptd Ins	1.05 <u>+</u> 0.02	1.01 <u>+</u> 0.06	0.89 <u>+</u> 0.01	0.94 <u>+</u> 0.09	1.02 <u>+</u> 0.03
Ptd A	1.20 <u>+</u> 0.44	1.25 <u>+</u> 0.005	1.42 <u>+</u> 0.37	1.31 <u>+</u> 0.27	1.18 <u>+</u> 0.07

0.5 ml samples of $\begin{bmatrix} 3^2 P \end{bmatrix}$ P, prelabelled RPM prepared as indicated (Section 2.2.2.3) were exposed to ADP for the times indicated. The results shown are mean values $\stackrel{+}{=}$ SD of triplicate determinations calculated as fold stimulation of basal (= 1). The experiment shown is typical of 2 similar experiments.

化过度超越感情的 建树脂 法保证法 化乙基乙基乙基乙基乙基乙基乙基乙基

。 1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:1993年:19

•

These results confirm the ability of ADP to induce the formation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA in RPM. However, unlike rat platelets, no demonstrable ADP-induced metabolism of PtdIns $(4,5)P_2$ could be produced. This may simply imply the lack of sensitivity of this technique in detecting extremely small changes in the levels of inositol phospholipids or indicate that one of the initial events occurring subsequent to agonist-receptor interaction in RPM does not include metabolism of PtdIns $(4,5)P_2$. If the latter were true, then the formation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA, previously assumed to be solely derived from inositol phospholipid metabolism, may be a coincidental event occurring in RPM upon agonist exposure.

To resolve this dilemma I thus attempted to monitor the effects of ADP on RPM inositol phosphate formation. Since these compounds are the water soluble products of inositol phospholipid metabolism, an accumulation of these compounds in particular $Ins(1,4,5)P_3$, would be suggestive that the metabolism of PtdIns $(4,5P)_2$ is one of the first events occurring following agonist-receptor interaction.

3.2.1.6 Effect of ADP on inositol phosphate formation in RPM

As outlined in Section 2.2.9.1, the metabolism of inositol phospholipids can be monitored either by measuring the disappearance of radioactive labelled inositol phospholipids e.g. disappearance of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdIns $(4,5)P_2$; by measuring the accumulation of radioactive labelled intermediaries in the inositol phospholipid metabolic cycle e.g. $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA or by measuring the accumulation of the water soluble products of inositol phospholipid metabolism e.g. the inositol phosphates. Unlike rat

- 203 -

platelets, RPM are capable of incorporating $\begin{bmatrix} 3\\ H \end{bmatrix}$ Inositol into their phospholipids (Section 2.2.9.1.2) and hence the metabolism of inositol phospholipids can be monitored in RPM by following the accumulation of inositol phosphates subsequent to agonist addition. Figure 39 depicts the time-course of ADP (3 μ M)induced accumulation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ -inositol phosphates in RPM from a typical experiment. The following values represent, the increase in the individual inositol phosphate species, expressed as a percentage above basal levels, for the example shown followed by the range in brackets.

Accumulation of $[^{3}H]$ Ins 1P was observed after a lag period of approximately 60s, at which point levels of $[^{3}H]$ Ins 1P had accumulated to 20% above basal (20 \pm 0.7%, mean \pm range = 2). Thereafter $[^{3}H]$ Ins 1P continued to accumulate, until by 5 min, levels were 50% above basal (35 \pm 20%, mean \pm range, n = 2).

Rapid accumulation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4)P_2$ was observed. Within 5 sec of agonist addition levels of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4)P_2$ were 40% above basal (66 ± 36%, mean ± range, n = 2). Thereafter $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4)P_2$ continued to accumulate, until by 60s levels were 220% above basal (191 ± 40%, mean ± range, n = 2).

 $\begin{bmatrix} 3_{\text{H}} \end{bmatrix}$ Ins $(1,4,5)P_3$ accumulated more rapidly than $\stackrel{3_{\text{H}}}{}$ Ins(1,4)For $\begin{bmatrix} 3_{\text{H}} \end{bmatrix}$ Ins 1P. Within 5s, $\begin{bmatrix} 3_{\text{H}} \end{bmatrix}$ Ins $(1,4,5)P_3$ levels were 120% above basal (79 $\stackrel{\pm}{}$ 57%, mean $\stackrel{\pm}{}$ range, n = 2) and the peak response of 190% above basal (144 $\stackrel{\pm}{}$ 65%, mean $\stackrel{\pm}{}$ range, n = 2) occurred within 30s. Thereafter levels in this triphosphate declined, until by 5 min, levels were reduced to 100% above basal (58 $\stackrel{\pm}{}$ 59%, mean $\stackrel{\pm}{}$ range, n = 2).

The fact that $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4,5)P_3$ and $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4)P_2$ do not continue to accumulate suggests that both are hydrolysed,

- 204 -





0.5 ml samples of $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol prelabelled RPM prepared as described (Section 2:2:2:4) were exposed to ADP (3 µm). The levels of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins. (1,4,5)P₂ (0): $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins (1,4)P₂ (D) and $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins IP (\triangle) were measured at the times indicated after agonist addition. The experiment shown is typical of 2 similar experiments. by specific phosphomonesterases, to yield [3H] Ins $(1,4)P_2$ and subsequently [3H] Ins 1P and [3H] inositol in the case of [3H] Ins $(1,4,5)P_3$, and [3H] Ins 1P and [3H] inositol in the case of [3H] Ins $(1,4)P_2$. This probably explains the lag period observed before significant accumulation of [3H] Ins 1P is observed - since, theoretically, in addition to being formed by the action of phospholipase C on PtdIns, [3H] Ins 1P could also be formed by the sequential hydrolyses of [3H] Ins $(1,4,5)P_3$ to [3H] Ins $(1,4)P_2$ and thence to [3H] Ins 1P. The other water soluble metabolites of inositol phospholipid metabolism, namely inositol and glycerophosphoinositol, were not routinely monitored as preliminary experiments demonstrated no significant changes in levels of [3H] glycerophosphoinositol and only a late increase in levels of [3H] inositol.

These results argue that one of the earliest events following agonist-receptor interaction in RPM is the metabolism of PtdIns(4,5)P₂ with consequent formation of $Ins(1,4,5)P_3$ assuming that PtdIns(4,5)P₂ is the sole source of $Ins(1,4,5)P_3$ (see Section 1.3.1.5). That agonist-induced accumulation of $\begin{bmatrix} ^3H \end{bmatrix}$ Ins(1,4,5)P₃ could be demonstrated, whereas the metabolism of $\begin{bmatrix} ^{32}P \end{bmatrix}$ PtdIns(4,5)P₂ could not (Section 3.2.1.5) indicates that monitoring the accumulation of the water soluble metabolites of inositol phospholipid metabolism, as well as being more direct, may be a more sensitive assay by which to demonstrate agonist-induced metabolism of inositol phospholipids.

3.2.2 <u>ADP-induced changes in cytosolic free calcium $\left[Ca^{2+} \right]_{i}$ </u> 3.2.2.1. <u>Effect of ADP on RPM cytosolic free calcium</u>

Having established that, as in rat platelets, ADP elicits inositol phospholipid metabolism in RPM, I next investigated the effect of ADP on cytosolic free calcium in RPM. Basal levels of $\left[Ca^{2+}\right]_{i}$ are approximately 150 nM. ADP (0.03 - 30 μ M) elicits a rapid and concentration-dependent elevation in $\left[\operatorname{Ca}^{2+}\right]_{i}$ (Figure 40a). As with rat platelets, as the agonist concentration increases, the rate of rise and decline in response also increases. Figure 40b depicts the cumulative concentration-response curve for ADP-induced elevation in $\left[\operatorname{Ca}^{2+}\right]_i$ in RPM. The maximal extent of the elevation in $\left[Ca^{2+}\right]_i$ above basal levels (approximately 600 nM) occurs at approximately 3 μ M ADP and the calculated EC₅₀ value is $0.22 \stackrel{+}{=} 0.07 \mu M$ (mean $\stackrel{+}{=} SEM$, n = 3). When ADP-induced changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$ were monitored in calcium-free buffers in the presence of EGTA (2 mM) a small, but reproducible response was observed in the absence of extracellular free calcium (Figure 41). This response amounts to approximately 30-40% degree of Ca²⁺ mobilization and is in contrast with the situation observed in rat platelets where, in the absence of extracellular calcium, the agonist-induced response is completely abolished.

3.2.2.2 Inhibition of ADP-induced changes in $\left[Ca^{2+j} \right]_{i}$

The effects of PGE_1 on agonist-induced changes in $[Ca^{2+}]_i$ in RPM, were investigated. Figure 42 depicts the effects of prior exposure to PGE_1 (10 μ M; 60s) on the elevation of $[Ca^{2+}]_i$ in RPM induced by maximal (3 μ M) and submaximal (0.3 μ M) concentrations of ADP. PGE_1 inhibited the response induced



Figure 40a: ADP-induced elevation of cytosolic free calcium (Ca^{2+}) , in RPM.

Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were incubated with ADP at the concentrations indicated. Changes in $[Ca^{2+}]$, were estimated from the observed changes in dye fluorescence (Section 2:2:4). Above represents typical responses from a single experiment representative of 2 others.



Figure 40b: <u>Concentration-response relationship for ADP-induced elevation</u> in cytosolic free calcium (Ca²⁺), in RPM.

Aliquots (2 ml) of Quin 2 labelled RPM prepared as described (Section 2:2:2:5) were incubated with ADP at the concentrations indicated. Results (mean values $\frac{1}{2}$ SEM) are cumulative observations from several experiments and are expressed as the increase in $\left[Ca^{2+1}\right]_{i}$ (nM) above basal levels.



Figure 41: ADP-induced elevation of cytosolic free calcium in the presence and absence of extracellular free calcium in RPM.

Aliquots (2 ml) of Quin 2-labelled platelets prepared as described (Section 2:2:2:5) were incubated with ADP at the concentrations indicated either in the presence or absence of extracellular free calcium. Changes in [Ca²⁺], were estimated from the observed changes in dye fluorescence (Section 2:2:4). Responses shown are from a single experiment typical of 3 others.



Figure 42: Effect of PGE on ADP-induced elevation of cytosolic free calcium [Ca²⁺], in RPM.

Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were exposed to ADP, at the concentrations indicated, in the presence or absence of PGE₁ as indicated. PGE₁ (10 μ M) was added 60s prior to ADP and changes in fluorescence monitored thereafter and converted to yield changes in [Ca²⁺]. Above are typical responses from a single experiment representative of 2 others.

- 211 -

by 3 μ M ADP by 40 ± 4%, and that evoked by 0.3 μ M ADP by 45 ± 4% (mean values ± range, n = 2-3). Both the rate and extent of ADP-induced elevation of $[Ca^{2+}]_i$ were impaired by PGE₁. These results indicate that, like rat platelets, RPM possess a cyclic AMP-dependent mechanism which acts to oppose evoked increments in $[Ca^{2+}]_i$.

3.2.3 Characterisation of receptors recognizing ADP in RPM

Having established that ADP elicits inositol phospholipid metabolism and elevation of $(Ca^{2+}]_i$ in RPM, as in rat platelets, I attempted to characterise the nature of the receptor(s) recognizing ADP in RPM by comparing the rank order of potency of a series of purines on their ability to induce the formation of $[3^2P]$ PtdA and elevate $[Ca^{2+}]_i$.

3.2.3.1 Effect of ATP, ADP, AMP and Adenosine on RPM inositol phospholipid metabolism: Concentration-response relationship

Figure 43 depicts the effects of ATP, ADP, AMP and Adenosine on the accumulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA in RPM. Both ADP (1-30 μ M) and ATP (1 μ M - 1 mM) but not AMP or Adenosine, when tested at a single concentration of 100 μ M, elicited, in a concentrationdependent manner, the formation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA. The threshold concentration for stimulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation was <3 μ M using ADP and <100 μ M using ATP. For the examples shown, the maximal stimulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation using ADP as the agonist is 1.3 fold of control (range 1.23 - 1.30) and 1.2 fold of control using ATP as the agonist (range 1.2 - 1.4). The estimated EC₅₀ values for ADP and ATP are respectively 1.60 \pm 0.05 μ M and 22 \pm 5 μ M (mean \pm SEM, n = 2-3). In RPM it appears



Figure 43: Concentration-response relationship for agonist-induced stimulation of [32P] PtdA formation in RPM.

0.4 ml samples of $\begin{bmatrix} 3^2 P \end{bmatrix}$ Pi prelabelled RPM prepared as described (Section 2:2:2:3) were exposed to ADP (\triangle); ATP (D); AMP (\Box) or Adenosine (\blacksquare) at the concentrations indicated. Results are measured in terms of fold stimulation, basal levels equal to 1. $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA levels were measured 30s after agonist addition. The results are means $\stackrel{+}{=}$ SD of triplicate determinations. The data shown for ATP and ADP is from separate experiments representative of 2-3 similar experiments. that ADP is more potent than ATP at eliciting the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA.

As in rat platelets, the possibility existed that the effects observed with ATP may result from the action of metabolites produced by hydrolysis of ATP rather than a direct effect of Consequently, the effects of the non-hydrolysable ATP ATP. analogue, ATP $(S, on [3^2P])$ PtdA formation was monitored (Figure Both ATP and ATP S (0.1 μ M - 1 mM) elicit, in a concentration 44). dependent manner, the formation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA. Maximal fold stimulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation induced by ATP δ S is 1.26(1.24 - 1.26 in different experiments). Calculated EC₅₀ values are $17 \pm 5.7 \mu M$ and $17.1 \pm 5.6 \mu M$ (mean \pm SEM, n = 3) for ATP and ATPYS respectively. No statistical difference exists between these two values or in the extent of maximal response elicited by either agonist. Thus ATP and ATPYS exert similar effects indicating that responses elicited by ATP in RPM most probably can be attributed to a direct effect of this agonist and not to a hydrolysis product that also acts upon the receptor recognizing purines.

3.2.3.2 Effect of ATP, ADP, AMP and Adenosine on RPM cytosolic free calcium, $\left[Ca^{2+1}\right]_{i}$

Figure 45 depicts the effects of ATP, ADP, AMP and Adenosine on changes in $[Ca^{2+}]_i$. Both ADP (0.03 - 30 µM) and ATP (0.03 -100 µM) but not AMP or Adenosine (1 - 100µM) elicit, in a concentration dependent manner, the elevation of $[Ca^{2+}]_i$. The maximal elevation in $[Ca^{2+}]_i$ was approximately 600 nM using ADP as the agonist and approximately 50 nM using ATP as the agonist. The calculated EC₅₀ values are 0.22 $\stackrel{+}{=}$ 0.07 µM and

- 214 -



Figure 44: Concentration-response relationship for ATP- and ATP S induced formatior of [32P] PtdA in RPM.

0.5 ml samples of $\begin{bmatrix} 3^2 P \end{bmatrix}$ P, prelabelled RPM prepared as described (Section 2:2:2:3) were exposed to ATP (open symbols) or ATP)S (Closed symbols) at the concentrations indicated. Results are measured in terms of fold stimulation - basal levels equal to 1. $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA levels were measured 30s after agonist addition. The results are means \pm SEM of triplicate determinations. The experiment shown is typical of 3 similar experiments.

11



Figure 45: Concentration-response relationship for agonist-induced changes in cytosolic free calcium (Ca⁻¹); in RPM.

Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were incubated with ADP (Δ); ATP (0); AMP (\Box) or Adenosine (Ξ) at the concentrations indicated. Results measured in terms of increase in $\begin{bmatrix} Ca^{2+1} \end{bmatrix}_{i}$ above basal levels (means - SEM). The data shown are the cumulative results from 3-4 experiments.

- 216 -

3.3 \pm 1.55 μ M (mean \pm SEM, n = 3-4) for ADP and ATP respectively. ATP-induced changes in $[Ca^{2+}]_i$ are rapid in onset (Figure 46) and comparable in rate to those evoked by ADP (Figure 40a). This could imply that ATP exerts a direct effect on RPM $[Ca^{2+}]_i$ without prior metabolism to ADP. To substantiate this point, the effects of ATP and ATPYS on RPM $[Ca^{2+}]_i$ were then compared (Figure 47). The estimated EC₅₀ values for agonist-induced elevation in $[Ca^{2+}]_i$ were 1.45 \pm 0.09 μ M for ATP and 0.26 \pm 0.02 μ M for ATPYS (mean \pm SEM, n = 2).

Figure 48 illustrates the effects of ATP and ATP/S on changes in $\left[C_a^{2+}\right]_i$ in RPM; no major difference in the nature of the evoked response is evident. Although ATP/S appears to be more potent than ATP, there is but a marginal difference in the maximal response (increase in $\left[Ca^{2+}\right]$ i) elicited by the two agonists. Thus, ATP-induced elevation of $\left[Ca^{2+}\right]_i$ in RPM results primarily from a direct action of ATP. Lastly, Figure 49 illustrates the total lack of effect of AMP and Adenosine on $\left[Ca^{2+}\right]_i$ in RPM.

In conclusion, from the data presented in this section it appears that RPM possess a receptor, at which ADP is the most potent agonist, which is coupled to inositol phospholipid metabolism and elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$.

3.2.4 Thrombin-induced metabolism of inositol phospholipids

The effect of Thrombin on inositol phospholipid metabolism in RPM was monitored in cells prelabelled with either $\begin{pmatrix} 3^2 P \end{pmatrix} P_i$ or $\begin{pmatrix} 3 H \end{pmatrix}$ Inositol. The basal levels of each of the inositol phospholipid species was routinely monitored, using either $\begin{pmatrix} 3^2 P \end{pmatrix} P_i$ or $\begin{pmatrix} 3 H \end{pmatrix}$ Inositol radiolabels, in addition to basal levels of



Figure 46: ATP-induced elevation of cytosolic free calcium (Ca²⁺), in RPM.

Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were incubated with ATP at the concentrations indicated. Changes in $[Ca^{2+}]$, were estimated from the observed changes in dye fluorescence (Section 2:2:4). Traces shown are from a single experiment representative of 2 others.

- 218 -

Fluorescence

RFI





Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were incubated with ATP (open symbols) or ATPYS (closed symbols) at the concentrations indicated. Results are measured in terms of increase in $[Ca^{2+}]$, above basal levels (mean - SEM). The experiment shown is typical of 2 others.

1999、1999年19月1日(1999年)(1999年)(1999年)(1999年)(1999年)(1999年)(1999年))(1999年) 1999年(1999年)(1999年)(1999年)(1999年)(1999年)(1999年)(1999年)(1999年)(1999年))(1999年)

- 219 -





Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were incubated with ATP (upper trace) or ATP(S (lower trace) at the concentrations indicated. Changes in $[Ca^{2+}]$, were estimated from the observed changes in dye fluorescence (Section 2:2:4). Those shown are typical responses from a single experiment representative of at least 2-3 others.

- 220 -



Figure 49: AMP- and Adenosine-induced elevation of cytosolic free calcium Cca²

Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were incubated with Adenosine (upper trace) or AMP (lower trace) at the concentrations indicated. Above represents typical responses of a single experiment, representative of 2-3 others.

and the set of the second second second second second

 $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol phosphates are as described in Section 3.2.1.

3.2.4.1 Effect of Thrombin on RPM inositol phospholipid metabolism: Time-course

Using a relatively high concentration of agonist, the time-course of Thrombin-induced metabolism of inositol phospholipids in RPM was examined. The time-course of Thrombin (200 nM)induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation is shown in Figure 50. The peak response, representing a 24% increase above basal levels in the experiment shown (24 \pm 0%, mean \pm SEM, n = 2) was observed at 60s. Thereafter levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA declined, reaching basal levels by approximately 5 min. No changes in $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns were observed either in the presence or absence of agonist. These results suggest that, like ADP, Thrombin is capable of eliciting the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA in RPM.

3.2.4.2 Effect of Thrombin on RPM inositol phospholipid metabolism Concentration-response relationship

Using a duration of agonist exposure of 60s, the concentration-response relationship for Thrombin-induced stimulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA formation was next investigated (Figure 51). It is clear, that in RPM, Thrombin is a very weak stimulant of inositol phospholipid metabolism eliciting an increase in $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA of approximately 12% above basal in the experiment shown. Due to the errors inherent in the system it is difficult to detect small changes in $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA of <10%. The observed changes with Thrombin fall close to this limit and thus difficulties arose in attempting to obtain consistent and measurable responses to Thrombin in this cell. From the available data, an estimate of the EC₅₀ value is



Figure 50: Time-course of Thrombin-induced changes in $\begin{bmatrix} 32p \end{bmatrix}$ PtdA in RPM. 0.4 ml samples of $\begin{bmatrix} 32p \end{bmatrix}$ P, prelabelled RPM prepared as described (Section 2:2:2:3) were exposed to Thrombin (200 nM) and the levels of $\begin{bmatrix} 32p \end{bmatrix}$ PtdA were measured at the times indicated after agonist addition. Results are expressed as fold stimulation of the basal level (equal to 1). The experiment shown is typical of 2 similar experiments.



Figure 51: Concentration-response relationship for Thrombin-induced stimulation of [32P] PtdA formation in RPM.

0.4 ml samples of $\begin{bmatrix} 3^2 P \end{bmatrix} P_1$ pre-labelled RPM prepared as described (Section 2:2:2:3) were exposed to Thrombin at the concentrations indicated and $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA levels were measured 60s after agonist addition. Results are expressed in terms of fold stimulation (basal level = 1). The results are means $\stackrel{+}{=}$ SEM of triplicate determinations. The experiment shown is typical of 3 similar experiments.

网络马马斯勒特斯马斯 化合同分子 网络金属铁铁

an shine she and the set of the set of the

< 150 nM. Due to the high cost of Thrombin and the weak responses evoked by this agonist, studies designed to monitor the inhibition of Thrombin-induced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation were not performed.

3.2.4.3 Effect of Thrombin on polyphosphoinosotide metabolism: Time-course

The effect of Thrombin on the metabolism of the polyphosphoinositides was next examined using $\begin{bmatrix} 3^2 P \end{bmatrix} P_1$ prelabelled RPM. The results are summarised in Table 4.

No demonstrable effect on levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂; $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdIns(4)P or $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdIns was observed subsequent to thrombin addition. However, the appearance of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA (significant at 5s) suggests, but does not necessarily prove, that metabolism of inositol phospholipids had occurred. (Incidentally these results, like those obtained with ADP (Section 3.2.1.5) are at variance with those obtained when monitoring $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation using a different extraction procedure (Section 3.2.4.1) again demonstrating that the present extraction procedure may be more effective). As outlined in Section 2.2.9.1.1, 1,2-diacylglycerol (the precursor of PtdA) can be derived from triacylglycerol. It is quite possible that production of PtdA is concurrent, but coincidental, with However, if one assumes that inositol agonist addition. phospholipids are the sole source of PtdA formation in RPM, then one possible explanation to describe the effect of Thrombin on the metabolism of polyphosphoinositides in RPM may be that, like ADP, Thrombin - due to its weak effect - is able to metabolise only a small 'pool' of $PtdIns(4,5)P_2$, one which is

able 4:	lime-course of	Thrombin-induced	changes	in the	e polyphosphoinositides
	<u>in RPM</u> .				

Phospholipid	Thrombin incubation period(s)					
	5	10	30	60	30 0	
Ptd Ins 4,5P ₂	1.04 <u>+</u> 0.08	1.04 <u>+</u> 0.08	1.03 <u>+</u> 0.07	1.01 <u>+</u> 0.08	1.08 <u>+</u> 0.05	
Ptd Ins 4P	1.01 <u>+</u> 0.02	0.99 <u>+</u> 0.03	0.99 <u>+</u> 0.01	1.02 ± 0.01	1.05 <u>+</u> 0.01	
Ptd Ins	0.95 <u>+</u> 0.02	0.87 <u>+</u> 0.08	0.86 <u>+</u> 0.06	0.95 <u>+</u> 0.04	0.91 <u>+</u> 0.10	
Ptd A	1.23 <u>+</u> 0.12	1.26 <u>+</u> 0.007	1.39 <u>+</u> 0.10	1.28 <u>+</u> 0.10	1.26 <u>+</u> 0.05	

Legend to Table 4

0.5 ml samples of $\begin{bmatrix} 3^2 P \end{bmatrix} P_1$ prelabelled RPM prepared as indicated (Section 2:2:2:3) were exposed to Thrombin for the times indicated. The results shown are mean values $\stackrel{-}{=}$ SD of triplicate determinations calculated as fold stimulation of basal (= 1). The experiment shown is typical of 2 similar experiments.

"你们的你还要找了你就是你们,你就要完美的时候,你们不能是你的你,要要们的?"。

rapidly replenished. Alternatively, metabolism of $PtdIns(4,5)P_2$ may play no part in the receptor-operated mechanisms for cellular reactivity in RPM, in which case, as already stated, formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA may in fact be a coincidental event.

To resolve this situation, of whether or not $PtdIns(4,5)P_2$ is metabolised in RPM following agonist addition, I attempted to monitor the accumulation of inositol phosphates in RPM. An early accumulation of $Ins(1,4,5)P_3$, the water soluble metabolite $PtdIns(4,5)P_2$, would be indicative of the metabolism of $PtdIns(4,5)P_2$ being in fact one of the earliest events occurring after agonist-receptor interaction in RPM.

3.2.4.4 Effect of Thrombin on inositol phosphate formation in RPM

The effect of Thrombin on inositol phosphate accumulation in RPM prelabelled with $\begin{bmatrix} 3\\ H \end{bmatrix}$ Inositol was examined. Thrombin (250 nM) induced the formation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4,5)P_3$; $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4)P_2$ and $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins 1P; the time-course of such a formation, from a typical experiment, is depicted in Figure 52. The following values represent the increase in appearance of the individual inositol phosphates expressed as a percentage above basal levels, for the example shown, followed by the range in brackets.

Thrombin induced a significant increase in the level of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins 1P only at the longest time point studied (5 min). At this point, $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins 1P had accumulated to 60% above basal (60 \pm 6%, mean \pm range, n = 2).

A rapid accumulation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4)P_2$ was observed. By 5s levels had accumulated to 21% above basal (21 ± 2%, mean ± range, n = 2). Thereafter levels of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins $(1,4)P_2$ continued to increase and were still increasing by 60s.



Figure 52: Time-course of Thrombin-induced formation of [3_H] inositol polyphosphates in RPM.

0.5 ml aliquots of $\begin{bmatrix} 3\\ H \end{bmatrix}$ inositol prelabelled RPM prepared as described (Section 2:2:2:4) were exposed to Thrombin (250 nM). The levels of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4,5)P₃ (0); $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4)P₄ (D) and $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins 1P (\triangle) were measured at the times indicated after agonist addition. The experiment shown is typical of 2 similar experiments.

 Thrombin also induced a rapid accumulation of $\begin{bmatrix} 3\\H \end{bmatrix}$ Ins(1,4,5) P₃ and by 5s levels were 6% above basal (6 ± 1%, mean ± range, n = 2). Peak response representing a 100% increase above basal (100 ± 10%, mean ± range, n = 2) was observed at 30s. Thereafter levels declined slightly reaching a plateau value of 74% above basal (74 ± 4%, mean ± range, n = 2) by 30-60s.

The continued presence of the diphosphate, in the example shown, is unlike the situation observed with ADP (Figure 39) where by 60s levels of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4)P₂ began to decrease. However, like ADP-induced accumulation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4)P₂, levels of this bisphosphate incurred the greatest degree of accumulation compared to either [3H] Ins $(1,4,5)P_3$ or [3H] Ins 1P. The lag period before any accumulation of $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Ins 1P occurs may be partly explained by the prolonged accumulation of $(3_{\rm H}]$ Ins(1,4)P₂, since, in addition to being formed from PtdIns by the action of phospholipase C, $\begin{bmatrix} 3 \\ H \end{bmatrix}$ Ins 1P can also be produced by the sequential hydrolysis of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4,5)P₂ to $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4)P₂ and hence to $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins 1P. However these results demonstrate that, like ADP, Thrombin is capable of inducing the rapid formation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4,5)P₃ and hence the metabolism of $PtdIns(4,5)P_2$. When compared to ADP however, Thrombin appears less efficacious at stimulating the formation of inositol phosphates, since at the peak response for each of the phosphates measured, greater accumulation was observed when ADP was used as the agonist. The ability of Thrombin to induce the accumulation of inositol phosphates in RPM, in the absence of a well defined effect on the metabolism of PtdIns(4,5)P, and accumulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA (Sections 3.2.4.2 and 3.2.4.3)
indicates that monitoring the production of inositol phosphates is a more sensitive technique by which to monitor agonistinduced metabolism of inositol phospholipids in RPM.

3.2.5 <u>Thrombin-induced changes in cytosolic free calcium $[Ca^{2+}]_{i}$ </u> 3.2.5.1 <u>Effect of Thrombin on RPM cytosolic free calcium Ca^{2+}_{i} </u>

Having investigated the effects of Thrombin on inositol phospholipid metabolism in RPM, I next investigated its effects on changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$. Thrombin is a poor stimulath of elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ in RPM (Figure 53). The maximum elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ observed (using 500 nM Thrombin) was approximately 200 nM above basal. Due to the high concentrations of thrombin required to construct full concentration-response curves - and the prohibitively high cost of such experiments - only a partial investigation into the effects of Thrombin on $(Ca^{2+})_i$ in RPM was viable. From the available data an estimated EC 50 value for Thrombin-induced elevation in Ca^{2+} , is 150 ± 100 nM (n = 2-3) and like experiments using ADP as the agonist approximately 30-40% change in $\left[\operatorname{Ca}^{2+}\right]_{i}$ is due to mobilization of intracellular calcium (results not shown). Thus Thrombin, while inducing the elevation of cytosolic free calcium, appears to be less efficacious at evoking the increase in $\left[\operatorname{Ca}^{2+}\right]_{i}$ in RPM, as was the case with inositol phospholipid metabolism.



Figure 53: Thrombin-induced elevation of cytosolic free calcium [ca²⁺], in RPM.

the strength of the second strength and the

lander medilik in her given date en significant i Simmer

the advantations growwatched a defension to prove that

Aliquots (2 ml) of Quin 2-labelled RPM prepared as described (Section 2:2:2:5) were incubated with Thrombin at the concentrations indicated. Changes in [Ca²⁺], were estimated from the observed changes in the dye fluorescence (Section 2:2:4). The responses shown above are from a single experiment representative of 2-3 others.

DISCUSSION

An early but historically very intuitive hypothesis by Michell (1975), concerned the role of agonist-induced phosphatidylinositol metabolism and resynthesis (the so-called "PI cycle") as a common receptor-mediated event for Ca²⁺ mobilization in many tissues. This concept was later modified (Michell et al., 1981) to include the possibility that metabolism of phosphatidylinositol-(4,5)-bisphosphate, rather than phosphatidylinositol, was the primary receptor-linked event. The major characteristics of this "PI-response" include the rapid agonist-induced metabolism of inositol phospholipids involving only those specific receptors known to act predominantly through calcium mobilization and occurring independently of extracellular calcium levels. Πn addition, a close coupling between inositol phospholipid metabolism receptor occupation, mobilization of calcium and the physiological response is often observed (Berridge, 1980; Michell and Kirk, 1981; Michell et al., 1981; Gill, 1985). More definitive evidence associating metabolism of inositol phospholipids and subsequent changes in cytoplasmic calcium levels has been reported by Berridge and colleagues (Berridge and Fain, 1979; Fain and Berridge, 1979). In blowfly salivary glands they demonstrated that in the continuous presence of agonist, the ability to mobilize Ca²⁺ gradually decreased. Upon incubation with inositol (and presumably subsequent inositol phospholipid metabolism) the ability of agonist to induce Ca²⁺ mobilization In the present study as a preliminary to elucidating returned. the mechanism(s) underlying stimulus-response coupling in both rat platelets and RPM, I decided to investigate the nature of

inositol phospholipid metabolism and calcium mobilization in both cell types. Such an investigation was designed to elucidate firstly whether or not common transduction processes exist in each cell type; secondly, whether such processes were capable of being manipulated pharmacologically; and thirdly revealing whether or not the megakaryocyte could be a potential site for novel antithrombolic therapy.

4.1 Pharmacology

4.1.1 Metabolism of inositol phospholipids

From the results (Section 3) it is clear that in both rat platelets and RPM, both Thrombin and ADP receptors are coupled to the metabolism of inositol phospholipids: monitored in rat platelets, either as the accumulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA or disappearance of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdIns(4,5)P₂ and in the RPM as the accumulation of $\begin{bmatrix} 3^2P \end{bmatrix}$ PtdA or $\begin{bmatrix} 3^2P \end{bmatrix}$

The time-courses of agonist-induced stimulation of $\binom{3^2p}{p}$ PtdA formation differed slightly between the two agonists and cell types. In rat platelets, ADP (Figure 10) attained its maximum stimulation within 10s, whereas using Thrombin (Figure 26) maximum stimulation was attained within 30s. The slower time to peak using Thrombin as the agonist may be explained by the fact that Thrombin evokes a response of greater magnitude than ADP. In RPM, both ADP (Figure 36) and Thrombin (Figure 50) attained maximal stimulation within 60s. The longer time to peak in RPM compared to rat platelets may reflect a difference in the balance of enzymes required for the metabolism of inositol phospholipids in each cell type. Additionally, it is possible

- 233 -

that the transduction processes in RPM are not as fully developed compared to rat platelets. The observation that stimulated levels of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA sometimes remain elevated above basal levels probably indicates some unique feature of that particular agonist in that particular cell type. When disappearance of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdIns(4,5)P₂ is monitored, in rat platelets, a significant decrease is observed within 5s of agonist addition using either ADP (Figure 13) or Thrombin (Figure 30) confirming that, like so many other systems, in rat platelets, metabolism of $PtdIns(4,5)P_{2}$ is one of the first events occuring subsequent to agonist-receptor interaction. No significant change in levels of inositol phospholipids, including PtdIns(4,5)P2, is observed in RPM when ADP (Table 3) or Thrombin (Table 4) is These results may suggest that metabolism of $PtdIns(4,5)P_{2}$ used. is not one of the first events to occur soon after agonistreceptor interaction in RPM or that agonist-induced metabolism. of $PtdIns(4,5)P_2$ in RPM triggers the metabolism of a small pool of $PtdIns(4,5)P_2$: one which is rapidly replenished such that no net decrease in levels of $[3^2P]$ PtdIns(4,5)P₂ occurs. The significant accumulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA using ADP and Thrombin in these studies (Tables 3 and 4) suggests that metabolism of inositol phospholipids did occur and favours the latter suggestion. When accumulation of $[3_H]$ inositol phosphates were monitored in RPM, both ADP (Figure 39) and Thrombin (Figure 52) elicited significant accumulation of $\begin{bmatrix} 3\\ H \end{bmatrix}$ Ins(1,4,5)P₃ within 5s of agonist addition. Assuming that $PtdIns(4,5)P_2$ is the sole source of Ins(1,4,5)P3 formation, this data would suggest that in RPM, exposed to either ADP or thrombin, metabolism of $PtdIns(4,5)P_2$

- 234 -

is in fact one of the first events initiated as a consequence of receptor occupancy. The fact that appearance of $Ins(1,4,5)P_3$ was observed, but not disappearance of $PtdIns(4,5)P_2$, substantiates the claim made previously - that ADP - and Thrombin-receptor occupation triggers the metabolism of a small pool of $PtdIns(4,5)P_2$ that is rapidly replenished - and suggests that, in RPM, monitoring accumulation of inositol phosphates may prove a more sensitive assay of inositol phospholipid metabolism rather than measuring the disappearance of $PtdIns(4,5)P_2$.

The construction of concentration-response curves allows a comparison of the potency of agonist examined, as well as indicating their efficacy. In rat platelets, the calculated EC_{50s} indicate a rank order of potency of Thrombin >>ADP (Figures 27 and 11). In addition, the maximum stimulation of PtdA formation evoked by Thrombin is far greater compared to that evoked by ADP. Hence, in rat platelets, it appears that Thrombin is by far more potent and efficacious compared to ADP. Similar studies performed in RPM suggest that ADP is more efficacious compared to Thrombin (Figures 37 and 51). However, in RPM, the extent of PtdA formation evoked by each agonist is low and approximates the limits of sensitivity of the phospholipid assay. Consequently, interpretation of the data frequently is difficult.

Nevertheless, these results suggest that, like so many other cell types, the rapid metabolism of inositol phospholipids, in particular $PtdIns(4,5)P_2$, is one of the first events occurring soon after agonist addition to either rat platelets or RPM (see Berridge, 1984). In rat platelets, at least, such rapid metabolism is contemporaneous with known platelet functional responses.

4.1.2 Changes in cytosolic free calcium [Ca²⁺]

Inherent in the hypothesis of the inosital phospholipid signal pathway being involved in cellular activation, is that the metabolism of inosital phospholipids somehow is coupled to changes in cytosolic free calcium. Changes in $\left[\operatorname{Ca}^{2+}\right]_i$ in rat platelets and RPM were monitored using the fluorescent quinoline dye Quin 2. As in phospholipid studies, both ADP and Thrombin when added to samples of rat platelets or RPM, elicited in a concentration-dependent manner, the elevation of Ca^{2+}_i (see Figures 15a,b; 32a,b; 40a,b; 53).

In rat platelets (Figures 32b and 15b) the calculated EC_{50} values indicate a rank order of potency of Thrombin >>ADP with respect to elevation of $[Ca^{2+}]_i$. Additionally, maximal elevation in $[Ca^{2+}]_i$ above basal reflects a value of approximately 1200 nM when ADP is used as the agonist and approximately 4000 nM when Thrombin is used. Hence in rat platelets, Thrombin is more potent and efficacious compared to ADP. When similar studies are performed in RPM (Figures 40b and 53), it appears that ADP is more efficacious compared to Thrombin. As with the analysis of Thrombin-induced metabolism of inositol phospholipids in RPM, large and prohibitively expensive amounts of Thrombin were necessary to perform these studies. Hence, as in the phospholipid studies, no EC_{50} values can be quoted for Thrombin-induced effects in RPM. Nevertheless, it appears that in both rat platelets and RPM, subsequent to the addition of ADP or Thrombin, a rapid and concentration-dependent elevation in $[Ca^{2+}]$ is observed. In rat platelets, at least, such rapid responses are contemporaneous with platelet functional responses (aggregation).

One prominent feature of the evoked Ca²⁺-signal, in both cell types, is the rapid decline in fluorescence (and hence in cytoplasmic free calcium concentration) towards basal values once the maximal elevation of $(Ca^{2+})_{i}$ has been attained. The decline is generally more rapid at higher agonist concentrations. Although this probably represents a mechanism by which the cell can limit its response after agonist addition, the exact mechanism(s) underlying this effect is unknown. Nevertheless, recent evidence suggests that the activation of protein kinase C by 1,2-diacylglycerol produced as a result of inositol phospholipid metabolism - besides acting synergistically with Ca²⁺ ions (Section 1.3.1.8.3) can actually exert a negative feedback role in agonist-induced Ca²⁺-mobilization (Drummond, 1985; Drummond and MacIntyre, 1985; MacIntyre et al., 1985a; MacIntyre et al., 1985b). In human platelets, MacIntyre et al. (1985a) demonstrated that prior addition of active phorbol esters e.g. PMA, inhibited the platelet activating factor (PAF)-induced elevation in cytoplasmic free calcium. More importantly, addition of PMA at the peak of the response, accelerated the decline in Ca^{2+} signal generated by PAF. These results would tend to suggest a role for activated protein kinase C in the regulation of calcium homeostasis after agonist addition. Similar effects are obtained in GH3 cells (Drummond, 1985). Indeed Drummond and MacIntyre (1985) have proposed that activated protein kinase C

may serve as a bidirectional regulator of cellular function both stimulating and inhibiting cellular activation. Other compounds such as membrane permeant derivatives of cyclic AMP and cyclic GMP, when added at the peak of the Ca²⁺-signal, are also known to rapidly reduce the Ca²⁺-signal in human platelets (MacIntyre <u>et al</u>., 1985a; MacIntyre <u>et al</u>., 1985b). However, no changes in levels of either cyclic AMP or cyclic GMP are detected following agonist addition (MacIntyre <u>et al</u>., 1985a; MacIntyre <u>et al</u>., 1985b) and hence presumably have little or no physiological role in calcium homeostasis.

Markedly different responses in agonist-induced changes in $\left[\operatorname{Ca}^{2+}\right]$; are observed in rat platelets and RPM when extracellular Ca²⁺ is removed from the suspension medium (Figures 16, 33 and 41). In rat platelets, the complete abolition of the response to ADP and Thrombin is observed when extracellular Ca²⁺ is removed. The simplest interpretation of this is that the change in $\left[Ca^{2+}\right]_{i}$, observed after agonist addition, in rat platelets, is entirely due to influx of extracellular Ca²⁺. An alternative explanation for this apparent dependency on extracellular calcium however could be that (extracellular) calcium is required to fill a trigger pool of calcium which is released into the cytosol following agonist addition. Nevertheless, the results observed in rat platelets, in the absence of extracellular calcium, are similar to those observed in human platelets where approximately 80-90% of the observed increase in $[Ca^{2+}]_{i}$ is due to influx, the remaining 10-20% derived via mobilization of Ca²⁺ from (intra)-cellular compartments (MacIntyre et al., 1985c). In RPM a small, but reproducible, response is observed

- 238 -

in the absence of extracellular Ca^{2+} , indicating a small degree of Ca^{2+} -mobilization is involved in agonist-induced elevation of $\left[\operatorname{Ca}^{2+}\right]_i$. The reason for the differences in requirement for extracellular Ca^{2+} for agonist-induced response between the two cell types is unknown. That the elevation of $\left[\operatorname{Ca}^{2+}\right]_i$ induced by ADP and Thrombin in rat platelets is lost in the absence of extracellular Ca^{2+} may suggest an obligatory requirement for an influx of calcium in the activation (e.g. metabolism of inositol phospholipids and ancillary sequelae) of rat platelets by these agonists. However, it has been demonstrated (Simon <u>et al</u>., 1984) that metabolism of inositol phospholipids can occur at basal levels of $\left[\operatorname{Ca}^{2+}\right]_i$.

The observed agonist-induced changes in $[Ca^{2+}]_i$ and inositol phospholipid metabolism in rat platelets and RPM exhibit many similarities with responses evoked by a variety of agonists in different cell types (see Berridge, 1984; MacIntyre <u>et al</u>., 1985c) and suggests that the receptor-initiated transduction processes in both rat platelets and RPM involve the metabolism of inositol phospholipids and changes in $[Ca^{2+}]_i$.

The major defect in the use of RPM as a surrogate to probe receptor-mediated events (ergo inositol phospholipid metabolism and Ca^{2+} flux) in megakaryocytes is the lack of a functional response to which the observed biochemical effects could be correlated. Although this is a disadvantage, it should be noted that a similar scenario pertains in rat brain slices, where, although now copious studies have been presented on the ability of various agonists to perturb the metabolism of inositol phospholipids (Jacobson <u>et al.</u>, 1985; Batty <u>et al.</u>, 1985) no functional correlate can be demonstrated in these preparations. The possibility of measuring heparin neutralising activity in RPM, as an index of functional responsiveness, was considered, however due to lack of time no studies were performed.

4.1.3 Specificity of agonist effects

One means of ensuring that stimulation of inositol phospholipid metabolism is a consequence of the specific interaction between an agonist and its specific receptors, and not just a non-specific effect occurring at the plasma membrane, would be to use selective receptor antagonists and demonstrate selective inhibition of agonist-induced phospholipid metabolism in the presence of these compounds. In the present study ADP and Thrombin were used to probe the transduction mechanisms in both rat platelets and RPM. Specific antagonists of the actions of ADP on human platelets e.g. 2 alkyl derivatives of AMP and ATP (Cusack et al., 1985) are known. However, such compounds were not available in substantial quantities to allow a study of their suitability as antagonists at the ADP receptor on rat platelets or RPM. Another method by which one can inactivate or inhibit platelet reactivity is to increase cytosolic levels of cyclic AMP (see Section 1.3.1.2). This can be performed by using agents such as PGE1. Although levels of cytoplasmic cytoplasmic cyclic AMP were not measured in the present study, earlier studies have demonstrated PGE1-induced elevation of cyclic AMP in rat platelets (Michel et al., 1976). The effects of PGE1 on inositol phospholipid metabolism and changes in cytosolic free calcium in rat platelets induced by ADP and Thrombin are demonstrated in Figures 12, 17, 29 and 35.

,

- 240 -

It is clear from these results that an increase in cytoplasmic levels of cyclic AMP is capable of inhibiting agonist-induced responses in rat platelets. Similar effects of PGE, on ADPinduced metabolism of inositol phospholipids and changes in $[Ca^{2+}]_{i}$ were observed in RPM (Figures 38 and 42). Again, levels of cyclic AMP in RPM subsequent to addition of PGE, were not measured in the present study. However, preliminary experiments carried out in this department did demonstrate PGE,-induced elevation of cyclic AMP in RPM (MacIntyre et al., 1983a and Section 2.2.9). In studies where Thrombin was used as the agonist, the effects of Hirudin were examined. Hirudin is known as a high affinity inhibitor of Thrombin and can inhibit both the stimulation of platelets by Thrombin and the binding of Thrombin to platelets (Detwiler and Feinman, 1973; Ganguly and Sonnichsen, 1976; Tam and Detwiler, 1978). Figures 28 and 34 demonstrate that, in rat platelets, Hirudin is capable of reducing both the metabolism of inositol phospholipids and changes in $\left[\operatorname{Ca}^{2+1}_{i}\right]_{i}$ induced by Thrombin.

Whilst these results demonstrate the reversal or inhibition of ADP- or Thrombin-induced metabolism of inositol phospholipids and changes in $[Ca^{2+}]_i$ in either cell type, conclusive proof of these effects being a receptor-mediated event await the development of specific receptor antagonists.

4.2 Classification of ADP receptor in rat platelets and RPM

Having established that the receptors recognizing ADP in both rat platelets and RPM, when activated, are coupled to the inositol phospholipid signal pathway (ergo metabolism

of inositol phospholipids and sequelae of events including elevation of cytoplasmic free calcium), I next attempted to characterise the nature of this "ADP" receptor in both cell types. This study involved a comparison of the rank order of potency of a range of purines e.g. ATP, ADP, AMP and Adenosine) on metabolism of inositol phospholipids (monitored as an accumulation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA) and elevation of cytoplasmic free From the results obtained it appears that the same calcium. rank order of potency (ADP > ATP >> AMP = Adenosine = 0) is obtained regardless of the nature of experiment or the tissue used. This suggests that a similar type of receptor exists on both cell types. This receptor has been characterised initially as an "ADP" receptor, since ADP is a full agonist in RPM at eliciting formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA and elevating cytoplasmic free calcium (Figures 43 and 45) whereas in rat platelets ADP is more potent compared to ATP, the latter acting as a partial agonist with respect to eliciting the formation of [32P] PtdA and elevating cytoplasmic free calcium (Figures 18 and 20). These results would tend to suggest that perhaps a separate and distinct receptor exists on rat platelets and RPM for ADP. As the potency ratios for ADPand ATP-induced inositol phospholipid metabolism:elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ differ, in rat platelets, (Table 5) the possibility exists that both effects are not mediated by a single receptor and thus heterogeneity of purinoreceptors must be invoked. In RPM it appears that only one "ADP" receptor exists, since the above mentioned ratio is the same irrespective of whether ADP or ATP is used as the agonist (Table 6). In view of the

- 242 -

<u>Table 5</u>: Comparison of potency (EC₅₀) values for agonistinduced $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA formation and elevation of $\begin{bmatrix} Ca^{2+1} \end{bmatrix}_{i}$ in rat platelets.

 EC_{50} (μ M)

	Agonist	ATP	ADP
Response			
Formation of [32P] PtdA		3.8 ± 1.0	0.9 ± 0.2
Elevation of [Ca ²⁺] _i		23.5 ± 54.	0.7 ± 0.2

Aliquots of $\begin{bmatrix} 3^2 P \end{bmatrix}$ P pre-labelled or Quin 2 labelled platelets prepared as described (Sections 2.2.1.2 and 2.2.1.3) were exposed to either ATP or ADP and the concentration-response relationship for both agonist induced formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA and elevation in $\begin{bmatrix} Ca^{2+} \end{bmatrix}$ was examined. From these curves the estimated EC values were calculated. The results represent the EC values (mean ± SEM, n = 2-6) obtained, using either ADP or ATP as the agonist, in such experiments.

Significance of differences between ATP c_dADP included PHDA formation and mobilization of Cazt in rat-plateles

(32P)Ptall formshon P=001 Gazt mobilization P200005

<u>Table 6</u>: Comparison of potency (EC₅₀) values for agonistinduced $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA formation and elevation of $\begin{bmatrix} Ca^2 + \end{bmatrix}$ in RPM.

$$EC_{50}$$
 (μ M)

	Agonist	ATP	ADP
Response			
Formation of [3 ² P] PtdA		22 ± 5.00	1.57 ± 0.05
Elevation of [Ca ²⁺]		3•3 ± 1•55	0.22 ± 0.07

Aliquots of $\begin{bmatrix} 3^2 p \end{bmatrix}$ P, prelabelled or Quin 2 labelled RPM prepared as described (Sections 2.2.2.3 and 2.2.2.5) were exposed to either ATP or ADP and the concentration-response relationship for both agonist-induced formation of $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA and elevation in $\begin{bmatrix} Ca^{2+1} \end{bmatrix}$, was examined. From these curves the EC₅₀ values for both agonists were estimated. The results represent the EC₅₀ values (mean $\stackrel{+}{-}$ SEM, n = 2-4) obtained, using either ADP or ATP as the agonist, in such experiments.

Significance of differences between ATP and ADP induced PtdA formation and Mobilizehougt Caz+ in RPM

[32P] PHORA formation P2005 (a2+mobilization PC00520025

possibility that the response observed with ATP may be secondary to its conversion to ADP, or due to the presence of a contaminant in the ATP preparation used during the studies, I decided to investigate the nature of the response obtained with ATP in greater detail. Were the responses observed with ATP were due to a contaminant, one may expect a "constant" level of contamination and hence any differences observed between ATP and ADP may be expected to be constant. Clearly from Figures 18, 20, 43 and 45 this is not the case and argues against contamination explaining the results obtained with ATP. Prior conversion of ATP to the di- and monophosphate compounds or Adenosine is a distinct possibility. However, studies investigating the nature of the ATP response in rat platelets were performed using washed platelets and, in studies with RPM, a balnanced salt solution was used to resuspend the cells, hence reducing the contribution of possible plasma/serum nucleotidase activity on nucleotides. The possibility of ectonucleotidases residing on the surface of these cells, capable of metabolising ATP remained a possibility. Hence, experiments were performed using the non-hydrolysable ATP analog ATPYS in an attempt to establish whether the observed response was due to ATP per se or to the metabolites of ATP. Figures 19, 22, 44 and 47 in the Results Section demonstrate that ATPYS, without exception, is capable of inducing the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA and elevating cytoplasmic free calcium in both rat platelets and RPM. Tn most experiments similar EC_{50} values for both ATP and ADP were These results argue for a direct action of ATP on reported. receptors recognizing purines in rat platelets and RPM. Additional evidence includes i) the observation that regardless

- 245 -

of whether ADP, ATP or ATP'S was used, no lag period before the onset of agonist-induced response (at least with respect to changes in cytosolic free calcium) was observed, Figures 15a, 23, 40a and 4**3**; ii) reference to Figures 24 and 49 demonstrate a total lack of effect of AMP and Adenosine at eliciting elevation of cytoplasmic free calcium. If the response to ATP was due to its metabolites (ergo ADP, AMP and Adenosine) then one may expect to observe a greater response than is actually observed with AMP or Adenosine. Due to the lack of effect of exogenous AMP and Adenosine, if the response to ATP is in fact due to a metabolite, then the primary candidate would be ADP. Of course, the possibility that part of the response elicited by ATP being due to ATP metabolites as observed in guinea pig atria and ileum by Moody et al. (1984) - cannot be ruled out and awaits further study. Nevertheles several workers have demonstrated the ability of ATP to elicit the formation of $\begin{bmatrix} 3^2 P \end{bmatrix}$ PtdA and elevate cytoplasmic free calcium in a variety of tissues including rat hepatocytes (Charest et al., 1985; Haynes et al., 1985); arterial smooth muscle (Watts and Borovey, 1983) and Ehrlich ascites tumor cells (Dubyak, 1986) underscoring the ability of ATP to elicit direct effects independently of any subsequent conversion.

Studies by Agarwal <u>et al</u>. (1980) suggested that ATP responses (aggregation) in rat platelets occurred as a result of its prior conversion to ADP. Clearly these results are at variance with those presented here and may be explained by methodological differences in experimental techniques employed in the respective studies. In conclusion, while attempting to classify the nature of the receptor(s) recognizing purines in both rat platelets and RPM, it appears that at least two of the four criteria used

- 246 -

to characterise any given receptor has been satisfied (see Section 3.1.3), namely a demonstration of the structure activity relationship, for a range of purines, has been made for the 'ADP' receptor and the nature of the transduction processes coupled to these receptors has been alluded to. Further investigations involving radioligand binding studies and the use of specific "ADP" receptor antagonists will aid in the further characterisation of the receptor(s) present on both cell types. Nevertheless, the available data suggests that there exists receptor(s) on both rat platelets and RPM which recognize both ADP and ATP. The data is inconsistent with other possible mechanisms, such as the membrane permeabilising effect of ATP known to occur at high $(>10 \mu M)$ concentrations (G omperts, 1983; Heppel et al., 1985). Of interest is the possibility that the receptor(s) recognizing purines on rat platelets may be different from those receptors recognizing purines on human platelets, since at these receptors, ADP is an agonist whereas ATP is an antagonist (Macfarlane and Mills, 1974). A recent review by Gordon (1986) has proposed the designation of P_{2T} for the receptor recognizing purines on human platelets (and megakaryocytes). Whether an additional subtype of receptor is now required to accommodate the present data awaits further study.

4.3 Correlation with metabolism of inositol phospholipids and elevation of $\left[\operatorname{Ca}^{2+1}\right]_{i}$

The hypothesis presented by Michell (1975) stated that the metabolism of phosphatidylinositol was somehow coupled to the increase in cytoplasmic free calcium. Although this hypothesis has been updated to incorporate new data, the nature of the link between the two events is still unknown. However, two pieces of evidence (albeit circumstantial) obtained from the present study suggest a correlation between the metabolism of inositol phospholipids and changes in cytosolic free calcium in both rat platelets and RPM. Firstly, the rank order of activity at eliciting elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ is the same as that inducing the metabolism of inositol phospholipids (measured as accumulation of PtdA) whether ATP, ADP, AMP, Adenosine or Thrombin is used as the agonist. Secondly, a contemporaneous relationship exists between changes in PtdA and elevation of $\left[\operatorname{Ca}^{2+}\right]_{i}$ in each cell type. Perhaps the most definitive evidence coupling these two events is the marked stiochiometry exhibited between receptor occupation, metabolism of inositol phospholipids, changes in $\left[\operatorname{Ca}^{2+}\right]_{i}$ and the Ca^{2+} -dependent physiological response - a close correlation between the concentrationresponse curves for these events is suggestive that the same receptor is responsible for mediating all the above mentioned responses. However, a general feature of receptors which are coupled to the inositol phospholipid signalling system is that the agonist concentrations required to maximally activate inositol phospholipid metabolism (and $Ins(1,3,5)P_3$ accumulation) are invariably higher than those required for maximal stimulation of Ca²⁺-mobilization and Ca²⁺-dependent cellular functions

(Kirk <u>et al</u>., 1981; Charest <u>et al</u>., 1985; Lynch <u>et al</u>., 1985; Creba et al., 1983). Using the formation of PtdA as an index of inositol phospholipid metabolism, such a discrepancy between agonist concentration inducing maximal elevation of $\left[Ca^{2+}\right]_{i}$ and formation of PtdA is observed for Thrombin in rat platelets and ATP and ADP in RPM (Figures 54a, b). In rat platelets no such difference exists when ADP is used as the agonist. Such a phenomenon, whereby a discrepancy does exist in the agonist concentration-range over which maximum PtdA formation and elevation in $[Ca^{2+}]_{i}$, occurs, may be explained by the concept of receptor reserve (Michell et al., 1981) whereby occupancy of a few receptors evokes maximal elevation of $\left[\operatorname{Ca}^{2+1}\right]_{i}$, but only a small degree of inositol phospholipid metabolism. However, formation of PtdA occurs distal to receptor occupancy and is preceded by metabolism of $PtdIns(4,5)P_2$ and accumulation of $Ins(1,4,5)P_3$. Perhaps a more valid correlation would be to compare the concentration-response curves for accumulation of ' $Ins(1,4,5)P_3$ and elevation of $[Ca^{2+}]_i$. When this is performed in rat hepatocytes, using a range of agonists (Lynch et al., 1985), a similar discrepancy between the concentration of agonist required to elicit maximal response for both effects is observed. Given these results, and the existence of spare receptors (receptor reserve) the present studies probably do reflect a receptor reserve for Thrombin, but not ADP, in rat platelets and for both ADP and ATP in RPM. However, further studies, most notably radioligand binding analysis would be required to substantiate this hypothesis in rat platelets and RPM. When ATP is used as the agonist in rat platelets, the

- 250 -





Aliquote of $\begin{bmatrix} 3^2 p \end{bmatrix} P$, prelabelled or Quin 2 labelled platelets prepared as described (Sections 2.2.1.2 and 2.2.1.3) were exposed to agonist at the concentrations indicated. $\begin{bmatrix} 3^2 p \end{bmatrix}$ PtdA levels (0) were measured 30s after agonist addition and elevation of $\begin{bmatrix} ca^{2+1} \end{bmatrix}$. (D) was calculated from the observed changes in dye fluorescence.¹ Results are the cumulative data from 2-5 experiments. For ease of comparison, data are expressed as percentage of maximum response. - 251 -



<u>Figure 54b</u>: Concentration-response relationship for agonist induced stimulation of PtdA formation and elevation of $[Ca^{2+}]_{,.}$

Aliquots of $\begin{bmatrix} 3^2p \end{bmatrix}$ P prelabelled or Quin 2 labelled RPM prepared as described (Sections¹2.2.2.3 and 2.2.2.5) were exposed to agonist at the concentrations indicated. $\begin{bmatrix} 3^2p \end{bmatrix}$ PtdA levels (0) were measured 60s after agonist addition and elevation of $\begin{bmatrix} Ca^{2+} \end{bmatrix}$. (**D**) was calculated from the observed changes in dys fluorescence.¹ Results are the cumulative data from 2-5 experiments. For ease of comparison, data are expressed as percentage of maximum response. concentration-response curve for elevation of $(Ca^{2+})_i$ appears to lie to the right of that for PtdA formation. This is a different phenomenon compared to that noted above and could suggest that transduction processes other than inositol phospholipid metabolism are involved in Ca²⁺-signalling. However, the exact mechanism(s) underlying its manifestation are not known.

The major unanswered question in cells utilizing the inositol phospholipid signal pathway still remains as to how metabolism of inositol phospholipids is coupled to elevation of $[Ca^{2+}]_{\cdot}$. Sources of Ca²⁺ used during cellular activation can be either intracellular or extracellular in origin. The missing link between mobilization of intracellular Ca²⁺ and metabolism of inositol phospholipids appears to be the water soluble metabolite of $PtdIns(4,5)P_2$ metabolism, $Ins(1,4,5)P_3$. Studies have shown its rapid formation subsequent to agonist addition, in a variety of different cell types including human platelets (Rittenhouse-Sasson, 1985); hepatocytes (Thomas et al., 1984) and GH, clonal pituitary tumor cells (Drummond <u>et</u> <u>al</u>., 1985; see Berridge, 1984). It appears to bind to specific sites (receptors) (probably the endoplasmic reticulum or dense tubular system in platelets (Spat et al., 1986). Moreover, it is capable of eliciting the release of Ca²⁺ from intracellular, non-mitochondria stores in various cell types (Streb <u>et al</u>., 1983; Burgess <u>et al</u>., 1984; Joseph et al., 1984; Gershengorn et al., 1984; O'Rourke et al., 1985; Authi and Crawford, 1985). Studies to examine whether or not $Ins(1,4,5)P_3$ is formed sufficiently rapidly in rat platelets following receptor occupancy by agonist to be utilized to mobilize intracellular calcium were hampered

and effectively precluded by the fact that these platelets do not appear to incorporate inositol to any great extent. However, due to the fact that RPM are grown and maintained in culture, it is feasible to label these cells with $\begin{bmatrix} 3 \\ H \end{bmatrix}$ inositol over a period of days allowing it to incorporate into the inositol phospholipid pool. The results presented here demonstrate that upon exposure to ADP (Figure 39) or Thrombin (Figure 52) RPM are capable of forming significant amounts of inositol triphosphate within 5s of agonist addition. This rapid formation of inositol trisphosphate is consistent with an action as a second messenger to mobilize intracellular calcium. The exact isomeric form of inositol trisphosphate accumulated in RPM was not determined. Irvine et al. (1985) demonstrated in rat parotid glands, that the 1,4,5-isomer was the predominant species accumulating at early time points. Thereafter the 1,3,4-isomer of inositol trisphosphate accumulated. Given that only the 1,4,5-isomer has definitively been demonstrated to induce release of Ca²⁺ from intracellular stores (Streb et al., 1983; Irvine et al., 1984b; Burgess et al., 1985; Burgess et al., 1984a) the nature of the isomer of inositol trisphosphate produced, in any system, is of utmost importance if one is to correlate its appearance with Ca²⁺-mobilization from internal stores. A possible additional source of intracellular calcium other than that mobilized from the endoplasmic reticulum/dense tubular system - could be that associated with $PtdIns(4,5)P_2$. This inositol phospholipid is known to bind Ca^{2+} and Mg^{2+} with high affinity (Dawson and Hauser, 1970) and it is possible that upon its metabolism it may release this bound Ca²⁺ (Billah and

- 253 -

Lapetina, 1982). However, under normal conditions, the concentration of Mg^{2+} ion is thought to exceed that of the Ca²⁺ ion by at least 1000 fold (see Downes and Michell, 1982) suggesting that the Mg^{2+} ion will bind preferentially to PtdIns $(4,5)P_2$, and that $PtdIns(4,5)P_2$ may not be a physiologically important source of Ca^{2+} .

While there appears little doubt as to how metabolism of inositol phospholipids is coupled to the mobilization of intracellular calcium, the method by which metabolism of inositol phospholipids is coupled to Ca²⁺-influx, in those cells predominantly using this source for cellular reactivity, is unknown. Extracellular calcium may enter cells through, at least, two types of calcium-channel, either voltage-operated or receptor-operated. Human platelets lack the former type of channel (MacIntyre and Rink, 1982; Doyle and Rtlegg, 1985) and there is no reason to assume that rat platelets, possess them. Considering the receptor-operated Ca²⁺ channels, the mechanism(s) governing its opening are unclear, however several hypotheses have been proposed (Berridge, 1980). For instance agonistreceptor interaction itself may perturb the plasma membrane opening a calcium channel: it is envisaged in this scenario that metabolism of inositol phospholipids could regulate the length of time the channel remains open. An alternative mechanism involves the proposed ionophoric properties of PtdA and its derivative lyso PtdA, both formed (rather indirectly) as a result of agonist-induced metabolism of inositol phospholipids. Both PtdA and LysoPtdA have been demonstrated to act as calcium ionophores (Putney et al., 1980; Lapetina et al., 1981a).

- 254 -

However, the inability of PtdA to translocate Ca^{2+} across membranes of purified phosphatidylcholine vesicles (Holmes and Yoss, 1983) suggested that it may not represent an important mechanism for regulating Ca^{2+} -flux. Recently, Watson <u>et al</u>. (1985) have proposed that, rather than acting as an intracellular messenger in mediating the mobilization of Ca^{2+} , LysoPtdA may act directly through cell surface, high affinity receptor sites, coupled directly or indirectly to mechanisms regulating Ca^{2+} influx. Other workers have proposed the existence of specific calcium channels in platelets (see Section 1.3.1.3) whether these channels are also regulated by the metabolism of inositol phospholipids is unknown.

Clearly there are many possible mechanisms of increasing levels of $\left[\operatorname{Ca}^{2+}\right]_{i}$ in cells which act predominantly through the inositol phospholipid signal pathway. One recent report has added a further dimension to the role of inositol phospholipid metabolism in Ca²⁺-flux. Putney (1986) proposes, that besides mobilizing Ca²⁺ from intracellular stores, Ins(1,4,5)P₃ may also act at the plasma membrane to increase the permeability to Ca^{2+} and thus allow an influx of Ca^{2+} into the cell. His proposal, based on a "capacitance model" allows for both the mobilization of Ca²⁺ from internal stores and Ca²⁺ influx to be governed by a single messenger, namely $Ins(1,4,5)P_3$. Recently several workers have demonstrated the existence of a higher phosphorylated derivative of inositol phosphates, designated Inositol(1,3,4,5)P4 (inositol tetrakisphosphate) (Batty et al., 1985; Hansen et al., 1986; Heslop et al., 1986; Michell, 1986;

- 255 -

Nahorski and Batty, 1986). Whilst a biological/physiological role has yet to be attributed to Inositol(1,3,4,5)P₄, Hansen et al. (1986) have suggested that it may also serve to integrate plasma membrane calcium fluxes with intracellular calcium mobilization.

4.4 <u>Alternatives to inositol phospholipid metabolism in rat</u> platelet/RPM activation

Having proposed that the inositol phospholipid signal pathway operates in both rat platelets and RPM, it is important to consider other possible mechanisms of activation in these two cell types. Hirata and Axelrod (1980) proposed that the methylation of membrane phospholipids plays an important role in receptor-mediated signal transduction in many cells. Platelets contain the necessary enzymes required for the methylation of phosphatidylethanolamine to phosphatidylcholine (Shattel et al., In the present study, no significant changes in either 1981). $\begin{bmatrix} 3^2 P \end{bmatrix}$ Phosphatidylethanolamine or $\begin{bmatrix} 3^2 P \end{bmatrix}$ phosphatidylcholine were recorded subsequent to agonist addition in either rat platelets or RPM (results not shown). Additionally, other workers have demonstrated that inhibitors of these enzymes (e.g. methyltransferases and 5-adenosylhomocysteine hydrolase) do not affect platelet activation (Hotchkiss et al., 1981; Shattel et al., 1982). It would thus appear that phospholipid methylation, while of possible importance in signal transduction in other cell types, is of minor importance in rat platelets Although \mathbf{d}_{ogma} suggests that inositol phospholipid and RPM. metabolism and Ca²⁺ flux are intimately involved in platelet activation, it is still possible that other transduction

mechanisms are involved in the sequence of events that link agonist-receptor interaction to cellular response (Huang and Detwiler, 1986). Clearly further studies are necessary to evaluate the importance of these additional processes.

CONCLUDING REMARKS

The present study was undertaken to investigate the nature of the transduction processes in both rat platelets and RPM cells. From my results it appears that the inositol phospholipid signalling pathway (ergo agonist-induced metabolism of inositol phospholipids and the associated sequelae of events, including elevation of $\left[Ca^{2+}\right]_{i}$) operates in both cell types. Upon agonist addition it appears that ADP, ATP and Thrombin all are capable of eliciting inositol phospholipid metabolism and elevation of $\left[\operatorname{Ca}^{2+}\right]$; in both rat platelets and RPM, although in RPM Thrombin is very much less efficacious compared to the other agonists. One possible explanation could be that, in RPM, receptors for Thrombin are poorly coupled to the transduction process. Additionally, given that the transit time from immature megakaryocytes to megakaryocytes capable of releasing platelets can be as short as 43 hours (Ebbe and Stohlman, 1965) the full expression of various receptors, destined for platelets, may occur at different periods in megakaryocyte development. In which case one may suggest that receptors for Thrombin are expressed later in development compared to receptors for purines. Future studies, involving more differentiated stages of RPM development may resolve this proposition.

- 257 -

Various workers, using mature megakaryocytes, have demonstrated, often to a striking degree, a similarity in the response obtained in these cells to known platelet agonists and those obtained in platelets using similar compounds (see Section 1.2.3). Whilst the present studies have confirmed and extended this similarity between the two types of cell we must consider other possible effects of the agonists used in the present study whilst investigating the pharmacology of the RPM. Such considerations extend to the possibility that, besides being agonists in platelets, the compounds used in the present study (e.g. Thrombin and purines) could possibly act as agents of cellular proliferation and growth in these immature cells. The two major ionic events that contribute to the onset of proliferation are changes in the level of calcium and the activation of a neutral Na^+/H^+ exchange carrier (Durham and Walton, 1982; Moolenaar et al., 1983). The methods by which metabolism of inositol phospholipids are coupled to elevation of $\left[\operatorname{Ca}^{2+}\right]$; have already been alluded to. Phorbol esters, acting presumably through protein kinase C (See Section 1.3.1.8.3), have been demonstrated to increase cellular pH in Swiss 3T3 cells (Burns and Rozengurt, 1983) suggesting that 1,2-diacylglycerol, produced from inositol phospholipid metabolism, may be responsible for activating the Na⁺/H⁺ carrier system involved in regulating cellular pH.Changes in inositol phospholipid metabolism have been described upon activation of a variety of cultured cells (Ristow et al., 1980; Sawyer and Cohen, 1981; Habenicht et al., 1981) by a variety of mitogens. Thus, whilst I have demonstrated the agonist-induced perturbation of the

- 258 -

inositol phospholipid signal pathway in RPM, the consequence of such effects, in RPM, may be quite different to those in rat platelets and may actually involve the promotion of growth and proliferation of these cells. Unfortunately, microscopic examination of the RPM before and after agonist addition were beyond the scope of the present study. However, this is one particular area that would be of interest to investigate further.

An attempt was made to characterise the nature of the receptor recognizing ADP in both rat platelets and RPM. It appears that the results obtained from this study do not fulfil entirely, any of the established criteria for either a P_1 - or P_2 - purinoreceptor or their subtypes (see Section 3). This suggests that a new subtype of purinoreceptor may be required to designate the receptor recognizing ADP on each cell type. Clearly further studies, including an analysis of the ability/ inability of ATP to elicit the elevation of cyclic AMP and/or production of icosanoids (see Section 3) will aid in the further characterisation of these receptors.

The products resulting from the action of phospholipase C on inositol phospholipids have been demonstrated to an integral role in cellular reactivity in many cell types, including platelets, subsequent upon agonist addition. It is conceivable that selective inhibition of this enzyme may represent a means by which to inhibit cellular (platelet) reactivity. Clearly such agents, if they were to exert similar effects in the progenitor cell, would represent a novel addition to anti-

- 259 -

thrombotic therapy.

Starling.

BIBLIOGRAPHY

- ADUNYAH, S.E. and DEAN, W.L. (1985). Biochem. Biophys. Res. Communc. <u>128</u>, 1274.
- AGARWAL, K.O., HASKEL, E.J. and PARKS, R.E. (1980). Biochem. Pharmac., <u>29</u>, 1799.
- AGRANOFF, B.W., MURTHY, P. and SEGUIN, E.B. (1983). J. Biol. Chem., <u>258</u>, 2076.
- AHAROY, D., SMITH, J.B. and SILVER, M.J. (1982). Biochim. Biophys. Acta., <u>718</u>, 193.
- AHLQUIST, R.P. (1948). Am. J. Physiol., <u>153</u>, 586.
- AKKERMAN, J.W.N. (1979). Thromb. Haemos., 39, 712.
- AKKERMAN, J.W.N., GORTER, G., SOONS, H. and HOLMSEN, H. (1983). Biochim. Biophys. Acta., <u>760</u>, 34.
- AKTORIES, K. and JAKOBS, K.H. (1984). Eur. J. Biochem., <u>145</u>, 333.
- ALLEN, D. and MICHELL, R.H. (1978). In: Cyclitols and Phosphoinositides, pp.325. Eds. WELLS, W. and EISENBERG, F. Academic Press.
- ANDERSON, W.B., ESTIVOL, A., TAPROVAARA, H. and GOPALAKRISHNA, R. (1985). Adv. Cyclic Nucl. Prot. Phos. Res., <u>19</u>, 287.
- ANTONIADES, H.N., SCHER, C.D. and STILES, C.D. (1979). Proc. Nat. Acad. Sci. U.S.A., <u>76</u>, 1809.
- APITZ-CASTRO, R. and MURCIANO, A. (1978). Biochim. Biophys. Acta., <u>544</u>, 529.
- APITZ-CASTRO, R., RAMIREZ, E., MAINGON, R., De MURCIANO, A. and FIBBI, A., (1976). Biochim. Biophys. Acta., <u>455</u>, 371.
- ARSLAN, P., DI VIRGILIO, F., BELTRAME, M., TSEIN, R.Y. and POZZAN, T. (1985). J. Biol. Chem. <u>260</u>, 2718.
- ASHBY, B., KOWALSKA, M.A., WERNICK, E., RIGMAIDEN, M., DANIEL, J.L. and SMITH, J.B. (1985). J. Cyclic Nucleo Prot. Phos., <u>10</u>, 473.

AUB, D.L. and PUTNEY, J.W. (1984). Life Sci., 34, 1347.

AUTHI, K.S. and CRAWFORD, N. (1985). Biochem. J. 230, 247.

- AUTHI, K.S., EVENDEN, B.J. and CRAWFORD, N. (1986). Biochem. J. <u>233</u>, 707.
- BATTY, I.R., NAHORSKI, S.R. and IRVINE, R.F. (1985). Biochem. J. 232, 211.
- BEGUINOT, L., HANOVER, J.A., ITO, S., RICHERT, N.D., WILLINGHAM, M.C. and PASTAN, I. (1985). Proc. Natl. Acad. Sci. U.S.A. <u>82</u>, 2774.
- BEHNKE, 0. (1969). J. Ultras. Res., <u>26</u>, 111.
- BEHNKE, 0. (1968). J. Ultras. Res., <u>24</u>, 412.
- BELL, R.L. and MAJERUS, P.W. (1980). J. Biol. Chem. 255, 1790.
- BELL, R.L., KENNERLY, D.A., STANDFORD, N. and MAJERUS, P.W. (1979). Proc. Natl. Acad. Sci. U.S.A. <u>76</u>, 3238.
- BENNETT, J.S. and VILAIRE, G. (1979). J. Clin. Invest. <u>64</u>, 1393.
- BENNETT, J.P., COCKCROFT, S. and GOMPERTS, B.D. (1981). J. Physiol., <u>317</u>, 335.
- BENNETT, J.S., COLMAN, R.F. and COLMAN, R.W. (1978). J. Biol. Chem., <u>253</u>, 7346.
- BERRIDGE, M.J. (1980). Trend Pharmacol. Sci., 1, 419.
- BERRIDGE, M.J. (1983). Biochem. J. 212, 849.
- BERRIDGE, M.J. (1985). Biochem. and Biophys. Res. Communc. <u>129</u>, 958.
- BERRIDGE, M.J. (1981). Molec. Cell Endocr. 24, 115.
- BERRIDGE, M.J. (1984). Biochem. J. <u>220</u>, 345.
- BERRIDGE, M.J. and FAIN, M.J. (1979). Biochem. J., <u>178</u>, 59.
- BERRIDGE, M.J. and IRVINE, R.F. (1984). Nature, <u>312</u>, 315.
- BERRIDGE, M.J., DAWSON, R.M.C., DOWNES, P. and HESLOP, J.P. (1983). Biochem. J. <u>212</u>, 473.
- BERRIDGE, M.J., HANLEY, M.R. and DOWNES, C.P. (1982). Biochem. J., <u>206</u>, 587.
- BERRIDGE, M.J., HESLOP, J.P., IRVINE, R.F. and BROWN, K.D. (1984). Biochem. J. <u>222</u>, 195.
- BERRIDGE, M.J., BUCHAN, P.B. and HESLOP, J.P. (1984a). Molec. Cell Endocr., <u>36</u>, 37.

- BILLAH, M.M. and LAPETINA, E.G. (1982). J. Biol. Chem. <u>257</u>, 12705.
- BILLS, T.K., SMITH, J.B. and SILVER, M.J. (1977a). Biochim. Biophys. Acta., <u>488</u>, 370.
- BILLS, T.K., SMITH, J.B. and SILVER, M.J. (1977b). J. Clin. Invest., <u>60</u>, 1.
- BIRNBAUMER, L. and RODBELL, M. (1976). J. Biol. Chem., <u>244</u>, 3477.
- BIRNBAUMER, L., POHL, S.L. and KAUMANN, A.J. (1974). Adv. Cyl. Nuc. Res., <u>4</u>, 239.
- BIRNBAUMER, L., CODINA, J., MATTERA, R., CERIONE, R.A., HILDEBRANDT, J.D., SUNYER, T., ROJAS, F.J., CARON, M.G., LEFKOWITZ, R.J. and IYENGAR, R. (1985). Recent Progr. H. Res. <u>41</u>, 41.
- BIZZOZERO, G. (1881). Virchows. Arch. Anat. Physiol. 90, 261.
- BLOW, D.M. (1975). Enzymes. 3rd Ed. 3, 185.
- BORN, G.V.R. (1962). Nature, <u>194</u>, 927.
- BORN, G.V.R. (1970). J. Physiol., 209, 487.
- BOWYER, D.E. and KING, J.P. (1977). J. Chromatogr., <u>143</u>, 473.
- BRANDT, D.R., ASANO, T., PEDERSEN, S.E. and ROSS, E.M. (1983). Biochemistry, <u>22</u>, 4357.
- BRANDT, S.T., DOUGHERTY, R.W., LAPETINA, E.G. and NIEDEL, J.E. (1985). Proc. Natl. Acad. Sci. U.S.A., <u>82</u>, 3277.
- BRASS, L.F. (1984a). J. Biol. Chem. 259, 12563.
- BRASS, L.F. (1984b). J. Biol. Chem., <u>259</u>, 12571.
- BRASS, L.F. (1984c). J. Biol. Chem., <u>260</u>, 2231.
- BRASS, L.F. and JOSEPH, S.K. (1985). J. Biol. Chem. 260, 15172.
- BRETON-GORIOUS, J. and REYES, F. (1976). Int. Rev. Cytol., <u>46</u>, 252.
- BRETON-GORIOUS, J. and VAINCHENKER, W. (1986). SEM Haematol. 23, 43.
- BROEKMAN, M.J. (1984). Biochem. Biophys. Res. Communc. <u>120</u>, 226.

- BROEKMAN, M.J., WARD, J.W. and MARCUS, A.J. (1980). J. Clin. Invest., <u>66</u>, 275.
- BROWN, J.E., RUBIN, L.J., GHALAYINI, A.J., TARVER, A.P., IRVINE, R.F., BERRIDGE, M.J. and WOLLHEIM, C.B. (1984a). Nature <u>309</u>, 562.
- BROWN, J.E., RUBIN, L.J., GHALAYINI, A.J., TARVER, A.P., IRVINE, R.F., BERRIDGE, M.J. and ANDERSON, R.E. (1984). Nature, <u>311</u>, 160.
- BRYDON, L.J., DRUMMOND, A.H., KIRKPATRICK, K.A., MACINTYRE, D.E.M., POLLOCK, W.K. and SHAW, A.M. (1984). Br. J. Pharm., <u>81</u>, 187P.
- BUNTING, S., MONCADA, S. and VANE, J.R. (1983). Br. Med. Bull. <u>39</u>, 271.
- BURGESS, G.M., IRVINE, R.F., BERRIDGE, M.J., McKINNEY, J.S. and PUTNEY, J.W. (1984a). Biochem. J. <u>224</u>, 741.
- BURGESS, G.M., GODFREY, P.P., McKINNEY, J.S., BERRIDGE, M.J., IRVINE, R.F. and PUTNEY, J.W. (1984b). Nature, <u>309</u>, 63.
- BURGESS, G.M., McKINNEY, J.S., IRVINE, R.F. and PUTNEY, J.W. (1985). Biochem. J. <u>232</u>, 237.
- BURNS, C.P. and ROZENGURT, E. (1983). Biochem. Biophys. Res. Communc., <u>116</u>, 931.
- BURNSTOCK, G. and BUCKLEY, N.J. (1985). In: "Methods in Pharmacology" 6, 193. Eds. PATON, D.M. Plenum Press Pub. Corp.
- BURNSTOCK, G. (1978). In: "Cell Membrane Receptors for Drugs and Hormones: A Multidisciplinary Approach" Eds. STRAUB, R.W. and BOLIS, L. Raven Press, N.Y.
- BURNSTOCK, G. (1981). In: "Purinergic Receptors", Receptors and Recognition Series B. Vol. 12. Ed. BURNSTOCK, G. Chapman and Hall.
- BUSHFIELD, M., GIBSON, I.F., MURDOCH, F.A. and MacINTYRE, D.E. (1986). Br. J. Pharm. in press.
- CAREY, F., MENASHI, S., AUTHI, K.S., HACK, N., LAGARDE, M. and CRAWFORD, N. (1985). Adv. Exp. Med. and Biol. <u>192</u>, 185.
- CARLSSON, E. (1979). Acta Pharmac. Toxicol., 44, Suppl. II, 17.
- CASSEL, D. and SELINGER, Z. (1976). Biochim. Biophys. Acta., <u>452</u>, 538.

- CASTAGNA, M., BRASSEUR, R., PAVOINE, C., BREDOUX, R., MELY, B., RUYSSCHAERT, J.M., LEVY-TOLEANDO, S. (1985). Adv. Exp. Med. Biol., <u>192</u>, 249.
- CASTAGNA, M., TAKAI, Y., KAIBUCHI, K., SANO, K., KIKKAWA, U. and NISHIZUKA, Y. (1982). J. Biol. Chem. <u>257</u>, 7847.
- CASTRO-MALASPINA, H., RABELLINO, E.M., YEN, A., NACHMAN, R.L. and MOORE, M.A. (1981). Blood, <u>57</u>, 781.
- CASWELL, A.H. (1972). J. Membr. Biol., 7, 345.
- CASWELL, A.H. and HUTCHISON, J.D. (1971a). Biochem. Biophys. Res. Communc. <u>42</u>, 43.
- CASWELL, A.H. and HUTCHISON, J.D. (1971b). Biochem. Biophys. Res. Communc., <u>43</u>, 625.
- CHAP, H., SIMON, M.F., FAUVEL, T., PLANTAVID, M., MAUCO, G. and DOUSTE-BUAZY, L. (1985). Nouv. Rev. fr Haematol. 27, 229.
- CHAREST, R., BLACKMORE, P.F. and EXTON, J.H. (1985a). J. Biol. Chem., <u>260</u>, 15789.
- CHAREST, R., BLACKMORE, P.F., BERTHON, C. and EXTON, J.H. (1983) J. Biol. Chem. <u>258</u>, 8769.
- CHAREST, P., PRPIC, V., EXTON, J.H. and BLACKMORE, P.F. (1985). Biochem. J., <u>227</u>, 79.
- CHARO, I.F., FEINMAN, R.D. and DETWILER, T.C. (1976). Biochem. Biophys. Res. Communc. <u>72</u>, 1462.
- CHERNOFF, A., LEVINE, R.F. and GOODMAN, DE W.S. (1980). J. Clin. Invest. <u>65</u>, 926.
- CHEUNG, W.Y. (1980). Science 207, 19.
- CHIANG, T.M., DIXIT, S.N. and KANG, A.H. (1976). J. Lab. Clin. Med. <u>88</u>, 215.
- CHIOU, C.Y. and MALAGOOI, M.H. (1975). Br. J. Pharmac., 53, 279.
- CHIU, H.C., SCHICK, P.K. and COLMAN, R.W. (1985). J. Clin. Invest., <u>75</u>, 339.
- CHOQUETTE, D., HAKIN, G., FILOTEO, A.G., PLISHKER, G.A., BOSTWICK, J.R. and PENNISTON, J.T. (1984). Biochem. Biophys. Res. Communc., <u>125</u>, 908.
- CICORIA, A.D. and HEMPLING, H.G. (1980). J. Cell Physiol. <u>105</u>, 105.
COCKROFT, S. (1981). Trend Pharm. Sci., 2, 340.

COCKCROFT, S. and GOMPERTS, B.D. (1985). Nature, 314, 534.

- COCKCROFT, S., BENNETT, J.P. and GOMPERTS, B.D. (1980a). FEBS Letts., <u>110</u>, 115.
- COCKCROFT, S., BENNETT, J.P. and GOMPERTS, B.D. (1980b). Nature, <u>288</u>, 275.
- CODINA, J., HILDEBRANDT, J.D., SEKURA, R.D., BIRNBAUMER, M., BRYAN, J., MANDLARK, C.R., IYENGAR, R. and BIRNBAUMER, L. (1984). J. Biol. Chem., <u>259</u>, 5871.
- COHEN, I. (1979). In: 'Methods and Achievements in Experimental Pathology' 9, 40. Ed. GABBIANI, G. S. Karger, Basel.
- COHEN, I., GERRARD, J.M., BERGMAN, R.N. and WHITE, J.G. (1979). In: 'Protides of the Biological Fluids' pp.555. Ed. PEETERS, H. Pergamon Press, Oxford.
- COLEMAN, R.A., HUMPHREY, P.A. and LUMLEY, P. (1984). Trends Pharm. Sci. <u>59</u>, 303.
- CONNOLLY, T.M., BROSS, T.E. and MAJERUS, P.W. (1985). J. Biol. Chem. <u>260</u>, 787B.
- CONNOLLY, T.M. and MAJERUS, P.W. (1986). Fed. Proc. 45, 1872.
- COOPER, D.M.F. and RODBELL, M. (1979). Nature, <u>282</u>, 517.
- COQUIL, J.F., FRANKS, D.J., WELLS, J.N., DEPUIS, M. and HAMET, P. (1980). Biochim. Biophys. Acta., <u>631</u>, 148.
- COUTURIER, A., BAZGAR, S. and CASTAGNA, M. (1984). Biochem. Biophys. Res. Communc., <u>121</u>, 448.
- CRAIG, I.H., BELL, F.P., GOLDSMITH, C.H. and SCHWARTZ, C.J. (1973). Atherosclerosis, <u>18</u>, 277.
- CRADDOCK, C.B., ADAMS, W.S., PERRY, S. and LAWRENCE, J.S. (1955). J. Lab. Clin. Med., <u>45</u>, 906.
- CRAMER, H. and SCHULTZ, J. (1977). In: 'Cyclic 3'-5'-Nucleotides: Mechanisms of Action'. Eds. CRAMER, H. and SCHULTZ, J. J. Wiley and Sons, London.
- CRAVEN, P.A. and DERUBERTIS, F.R. (1980). Metabolism, 29, 842.
- CRAWFORD, N. (1985). Adv. Exp. Med. Biol. 192, 1.
- CREBA, J.A., DOWNES, C.P., HAWKINS, P.T., BREWSTER, G., MICHELL, R.H. and KIRK, C.P. (1983). Biochem. J. <u>212</u>, 733.

- CRONKITE, E.P. (1957). In: 'Homeostatic Mechanisms', pp.96 Brookhaven Symposia in Biology, 10.
- CROSET, M. and LARGARDE, M. (1983). Biochem. Biophys. Res. Communic. <u>112</u>, 878.
- CUSACK, N.J. and HOURANI, S.M.O. (1982). Br. J. Pharm. 75, 397.
- CUSACK, N.J., HOURANI, S.M.O. and WELFORD, L.A. (1985). Adv. Exp. Med. Biol., <u>192</u>, 29.
- CUTLER, L., RODAN, G. and FEINSTEIN, M.B. (1978). Biochim. Biophys. Acta, <u>542</u>, 353.
- CUTLER, L.S., CHRISTIAN, C.P. and FEINSTEIN, M.B. (1985). Biochem. Biophys. Acta, <u>845</u>, 403.
- DABROWSKI, R. and HARTSHORNE, D.J. (1978). Biochem. Biophys. Res. Communc. <u>85</u>, 1352.
- DANIEL, J.L., HOLMSEN, H. and ADELSTEIN, R.S. (1977). Thromb. Haemos. <u>38</u>, 984.
- DANIEL, J.L., MOLISH, I.R. and HOLMSEN, H. (1981). J. Biol. Chem., 256, 7510.
- DANIEL, J.L., MOLISH, I.R., RIGMAIDEN, M. and STEWART, G. (1984). J. Biol. Chem. <u>259</u>, 9826.
- DAVEY, M.G. and LUSCHER, E.F. (1967). Nature, 216, 857.
- DAWSON, A.P., COMERFORD, J.G. and FULTON, D.V. (1986). Biochem. J. <u>234</u>, 311.
- DAWSON, R.M.C. and HAUSER, H. (1970). In: 'Calcium and Cellular Function' pp.17. Ed. CUTHBERT, A.W. Macmillan, London.
- DAY, H.J. and HOLMSEN, H. (1971). Semin. Haematol. 4, 3.
- DE CHAFFOY DE COURCELLES, D., ROEVENS, P. and VAN BELLE, H. (1984b). FEBS Letts. <u>173</u>, 389.
- DE CHAFFOY DE COURCELLES, D., ROEVENS, P. and VAN BELLE, H. (1984a). Biochem. Biophys. Res. Communc. <u>123</u>, 589.
- DE CHAFFOY DE COURCELLES, D., ROEVENS, P. and VAN BELLE, H. (1986). FEBS LETTS., <u>195</u>, 115.
- DEAN, W.L. and SULLIVAN, D.M. (1982). J. Biol. Chem. 257, 14390.
- DEMERS, L.M., BUDIN, R.E. and SHAIKH, B.S. (1980). Proc. Soc. Exp. Biol. Med., <u>163</u>, 24.
- DETWILER, T.C. and McGOWAN, E.B. (1985). Adv. Exp. Med. Biol., 192, 15.

DETWILER, T.C. (1981). Ann. NY Acad. Sci., 370, 67.

DETWILER, T.C. and FEINMAN, R.D. (1973). Biochem., <u>12</u>, 282.

- DETWILER, T.C. and HUANG, E.M.C. (1980). IN: "The Regulation of Coagulation". Proceedings of the International Workshop on Regulation of Coagulation. Eds. MANN, K.G. and TAYLOR, F.B. Elsevier, N. Holland, Amsterdam.
- DETWILER, T.C., CHARO, I.F. and FEINMAN, R.D. (1978). Thromb. Haemos. <u>40</u>, 207.
- DIAMANT, B. and KRUGER, P.G. (1967). Acta Physiol. Scand. 71, 291.
- DOUGLAS, W.W. and RUBIN, R.P. (1961). J. Physiol. <u>159</u>, 40.
- DOWNES, C.P. and MICHELL, R.H. (1982). Cell Calcium, 3, 467.
- DOWNES, P., HAWKINS, P.T. and MICHELL, R.H. (1982). Biochem. Soc. Trans. <u>10</u>, 250.
- DOYLE, V.M. and RUEGG, U.I. (1985). Biochem. Biophys. Res. Communc. <u>127</u>, 161.
- DRUMMOND, A.H. (1985). Nature, <u>316</u>, 752.
- DRUMMOND, A.H. and MACINTYRE, D.E. (1985). Trends Pharm. Sci. <u>6</u>, 233.
- DRUMMOND, A.H., BUSHFIELD, M. and McPHEE, C.H. (1984). Mol. Pharmacol. 25, 201.
- DRUMMOND, A.H., KNOX, R.J. and McPHEE, C.H. (1985). Biochem. Soc. Trans. <u>13</u>, 58.
- DRUST, D.S. and MARTIN, T.F.J. (1985). Biochem. Biophys. Res. Communc., <u>128</u>, 531.
- DUBYAK, G.R. (1986). Arch. Biochem. Biophys. 245, 84.
- DUNWIDDIE, T.V. and FREDHOLM, B.B. (1982). Proc. Scand. Brit. Pharmac. Soc. p.8.
- DURELL, J., GARLAND, J.T. and FRIEDEL, R.O. (1969). Science, <u>165</u>, 862.
- DURFLER, F.J., MAHAN, L.C., KOACHMAN, A.M. and INSEL, P.A. (1982). J. Biol. Chem. <u>257</u>, 11901.
- DURHAM, A.C.H. and WALTON, J.M. (1982). Biosci. Rep., 2, 15.
- EBBE, S. (1981). In: "Megakaryocyte Biology and Precursors: In Vitro Cloning and Cellular Properties, pp.1. Eds. EVATT, LEVINE and WILLIAMS. Elsevier, N. Holland, Amsterdam.

- EGAWA, K., SACKTOR, B. and TAKENAWA, T. (1981). Biochem. J. <u>194</u>, 129.
- EHRLICH, H.P. and GORDON, J.L. (1976). In: "Platelets in Biology and Pathology" Vol. 1, pp.353.
- ELDOR, A., AVITZOUR, M., OR, R., HANNA, R. and PENCHAS, S. (1982). Brit. Med. J. <u>285</u>, 397.
- EVANS, T., HELPER, J.R., MASTERS, S.B. and BROWN, J.H. (1985). Biochem. J., 232, 751.
- FAIN, J.N. and BERRIDGE, M.J. (1979). Biochem. J., <u>178</u>, 45.
- FAIN, J.N. and BERRIDGE, M.J. (1979a). Biochem. J., <u>180</u>, 655.
- FARESES, R.V. (1984). Mol. and Cell Endocrinol., 35, 1.
- FEDORKO, M.E. (1977). Lab. Invest., <u>36</u>, 310.
- FEIN, A., PAYNE, R., CARSON, D.W., BERRIDGE, M.J. and IRVINE, R.F. (1984). Nature, <u>311</u>, 157.
- FEINMAN, R.D. and DETWILER, T.C. (1974). Nature, <u>249</u>, 172.
- FEINMAN, R.D. and DETWILER, T.C. (1974). Nature, 249, 172.
- FEINSTEIN, M.B. (1982). Haemostasis Thromb., 6, 25.
- FEINSTEIN, M.B. (1980). Biochem. Biophys. Res. Communc. 93, 593.
- FEINSTEIN, M.B. (1978). In: Calcium in Drug Action. pp.197. Ed. WEISS, G. Plenum, N. York.
- FEINSTEIN, M.B. and HADJIAN, R.A. (1981). Mol. Pharm., 21, 422.
- FEINSTEIN, M.B., EGAN, J.J. and OPAS, E.E. (1983b). J. Biol. Chem. <u>258</u>, 1260.
- FEINSTEIN, M.B., EGAN, J.J., SHAAFI, R.I. and WHITE, J. (1983a). Biochem. Biophys. Res. Communc., <u>113</u>, 598.
- FEINSTEIN, M.B., HALENDA, S.P., ZAVOICO, G.B. and CHESTER, D. (1985). In: "Prostaglandins, Leucotrienes and Lipoxins. Ed. BAILEY, J.M. Plenum, N. York.
- FISHER, G.J., BAIKSHIAN, S. and BALDASSARE, J.J. (1985). Biochem. and Biophys. Res. Communc., <u>129</u>, 958.
- FLORIO, V.A. and STERNWEISS, P.C. (1985). J. Biol. Chem., <u>260</u>, 3477.

- FOX, J.E.B., LYNHAM, J.A. and HASLAM, R.J. (1978). Fed. Proc., <u>37</u>, 799.
- FRIEDMAN, F. and DETWILER, T.C. (1975). Biochemistry, <u>14</u>, 1315.
- GAARDER, A., JONSEN, J., LALAND, S., HELLEM, A. and OWREN, P. (1961). Nature, <u>192</u>, 531.
- GANGULY, P. and SONNICHSEN, W.J. (1976). Br. J. Haematol., <u>34</u>, 291.
- GERRARD, J.AM., PETERSON, D.A. and WHITE, J.G. (1981). In: "Platelets in Biology and Pathology" Vol. 2, pp.407. Ed. GORDON, J. Elsevier, N. Holland Biomedical Press, Amsterdam.
- GERRARD, J.L., WHITE, J.G. and PETERSON, D.A. (1978). Thromb. Haemos., <u>40</u>, 224.
- GERRARD, J.M. and WHITE, J.G. (1978a). In: "Progress in Haemostasis and Thrombosis" Vol. 4, pp.87. Ed. SPAET, T.H. Grune and Stratton, N. York.
- GERRARD, J.M., PETERSEN, D.A. and WHITE, J.G. (1982). In: "Platelets in Biology and Pathology" Vol. 2, pp.407. Ed. GORDON, J.L.
- GERRARD, J.M., SCHOLLMEYER, J.V., PHILLIPS, D.R. and WHITE, J.B. (1979). Am. J. Pathol., <u>94</u>, 509.
- GERRARD, J.M., WHITE, J.G., RAO, G.H.R. and TOWNSEND, D. (1976). Am. J. Pathol. <u>82</u>, 283.
- GERSHENGORN, M.C., GERAS, E., PURRELLO, V.S. and REBECCHI, M.J. (1984). J. Biol. Chem., <u>259</u>, 10675.
- GEWIRTZ, A.M. (1986). SEM Haematol., 23, 27.
- GILL, D.L. (1985). Adv. Cyclic Nucleot. and Prot. Phosp. Res., 19, 307.
- GILMAN, A.G. (1984). Cell, <u>36</u>, 577.
- GOMPERTS, B.D. (1983). Nature, 306, 64.
- GOODHARDT, M., FERRY, W., GEYNET, P. and HANOUNE, J. (1982). J. Biol. Chem. <u>257</u>, 11577.
- GORDON, J.L. (1986). Biochem. J., <u>233</u>, 309.
- GORDON, J.L. and MILNER, A.J. (1976). In: "Platelets in Biology and Pathology" Vol. 1, pp.3. Ed. GORDON, J.L.

ŧ

- GORMAN, R.R., BUNDY, G.L., PETERSON, D.C., SUN, F.F., MILLER, O.V. and FITZPATRICK, F.A. (1977). Proc. Natl. Acad. Sci. U.S.A., <u>74</u>, 4007.
- GRAFF, G., STEPHENSON, J.H., GLASS, D.B., HADDOX, M.K. and GOLDBERG, N. (1978). J. Biol. Chem. <u>253</u>, 7662.
- GRAFF, G., NAHAS, N., NIKOLOPOULOU, M. and NATARAJAN, V. and SCHMID, H.H. (1984). Arch. Biochem. Biophys, <u>228</u>, 299.
- GRETTE, K. (1962). Acta Physiol. Scand. <u>56</u>, Suppl. 195.
- GRUETTER, D.Y. and IGNARRO, L.J. (1979). Prostaglandins, 18, 541.
- GRYNKIEWICZ, G., POENIE, M. and TSEIN, R.Y. (1985). J. Biol. Chem., <u>260</u>, 3440.
- HABENICHT, A.J., GLOMSET, J.A., KING, W.C., NIST, C., MOTCHELL, C.D. and ROSS, R. (1981). J. Biol. Chem. 256, 12329.
- HAGEN, I. (1975). Biochim. Biophys. Acta. 392, 242.
- HAHN, F.E. and OLSSON, R.A. (1975). The Pharmacologist, 19, 382.
- HALENDA, S.P. and FEINSTEIN, M.B. (1984). Biochem. Biophys. Res. Communc., <u>124</u>, 507.
- HALLAM, T.J. and RINK, T.J. (1985). J. Physiol., 368, 131.
- HALLAM, T.J. and RINK, T.J. (1985a). FEBS Letts., <u>186</u>, 175.
- HALLAM, T.J., SANCHEZ, A. and RINK, R.J. (1983). J. Physiol., <u>343</u>, 97-98P.
- HAMBERG, M. and SAMUELSON, B. (1974). Proc. Natl. Acad. Sci. U.S.A., 71, 3400.
- HAMBERG, M., SVENSSON, J., WAKABAYASHI, T. and SAMUELSSON, B.= (1974). Proc. Natl. Acad. Sci. U.S.A., <u>71</u>, 345.
- HAMBERG, M. and SAMUELSSON, B. (1973). Proc. Natl. Acad. Sci. U.S.A., <u>70</u>, 899.
- HAMBERG, M., SVENSSON, J. and SAMUELSSON, B. (1975). Proc. Natl. Acad. Sci. U.S.A., <u>72</u>, 2994.
- HAMMARSTROM, S. and FALARDEAU, P. (1977). Proc. Natl. Acad. Sci. U.S.A., <u>74</u>, 3691.
- HANSEN, C.A., MAH, S. and WILLIAMSON, J.R. (1986). J. Biol. Chem. <u>261</u>, 8100.

- HARDISTY, R.M., HUTTON, R.A., MONTGOMERY, D., RICKARD, S. and TREBLOCK, H. (1970). Br. J. Haematol., 19, 307.
- HARKER, L.A., ROSS, R. and GLOMSET, J.A. (1978). Thromb. Haemos. <u>39</u>, 313.
- HARKER, L.A. and FINCH, C.A. (1969). J. Clin. Invest., <u>48</u>, 963.
- HARMON, J.T. and JAMIESON, G.A. (1985). Biochemistry, 24, 58.
- HASLAM, R.J., (1964). Nature, 202, 765.
- HASLAM, R.J. and DAVIDSON, M.M.C. (1984a). Biochem. J., 222, 351.
- HASLAM, R.J. and DAVIDSON, M.M.L. (1985). Nature, 313, 821.
- HASLAM, R.J. and DAVIDSON, M.M.L. (1984). J. Receptor Res., <u>4</u>, 605.
- HASLAM, R.J., DAVIDSON, M.M., DAVIES, T., LYNHAM, J.A. and McLENAGHAN, M.D. (1978). Adv. Cyclic Nucleo Res., <u>9</u>, 533.
- HASLAM, R.J., DAVIDSON, M.M.C. and DESJARDINE, J.V. (1978a). Biochem. J., <u>176</u>, 83.
- HASLAM, R.J. and LYNHAM, J.A. (1977). Biochem. Biophys. Res. Communc. <u>77</u>, 714.
- HASLAM, R.J., LYNHAM, J.A. and FOX, J.E.B. (1979). Biochem. J., <u>178</u>, 397.
- HASLAM, R.J., SALAMA, S.E., FOX, E.B., LYNHAM, J.A. and DAVIDSON, M.M. (1980). In: "Platelets: Cellular Response Mechanisms and their Biological Significance". pp.213. Eds. ROTMAN, A., MEYER, F.A., GITLER, C. and SIBERBENG, A. Wiley, Gt. Britain.
- HATHAWAY, D.R. and ALDELSTEIN, R.S. (1979). Proc. Natl. Acad. Sci. U.S.A. <u>76</u>, 1653.

HAWTHORNE, J.N. (1982). Nature, <u>295</u>, 281.

۱

HAWTHORNE, J.N. and PICKARD, M.R. (1979). J. Neurochem., 32, 5.

- HAYNES, R.C., PICKING, R.A. and SISTARE, F.D. (1985). Fed. Proc., <u>44</u>, 483.
- HELDIN, C-H., WESTERMARK, B. and WASTESON, A. (1979). Proc. Natl. Acad. Sci. U.S.A., <u>76</u>, 3722.

HELLEM, A.J. (1960). Scand. J. Clin. Lab. Invest., 12, Suppl. 51,

- HEPPEL, L.A., WEISMAN, G.A. and FRIEDBERG, I. (1985). J. Membr. Biol., <u>86</u>, 189.
- HESKETH, T.R., SMITH, G.A., MOORE, J.P., TAYLOR, M.V. and METCALFE, J.C. (1983). J. Biol. Chem., <u>258</u>, 4876.
- HESLOP, J.P., IRVINE, R.F., TASHIJAN, A. and BERRIDGE, M.J. (1985). J. Exp. Biol. <u>119</u>, 395.
- HIDAKA, H. and ASANO, T. (1976). Biochim. Biophys. Acta, 429, 429.
- HIDEBRANDT, J.D., CODINA, J., ROSENTHAL, W., SUNYER, T., IYENGAR, R. and BIRNBAUMER, L. (1985). Adv. Cyclic Nucleot. Prot. Phos. Res., <u>19</u>, 87.
- HIRATA, F. and AXELROD, J. (1980). Science, 209, 1082.
- HOFFMAN, R., MAZUR, E., BRUNO, E. and FLOYD, V. (1981). N. Eng. J. Med., <u>305</u>, 533.
- HOLCK, M., THORENS, S., MUGGLI, R. and EIGENMANN, R. (1984). J. Cardiovas. Pharmac., 6, 520.
- HOKIN, M.R. and HOKIN, L.E. (1953). J. Biol. Chem., 203, 967.
- HOLMES, R.P. and YOSS, N.L. (1983). Nature, 305, 637.
- HOLMSEN, H. (1975). In: Ciba Foundation Symposium, Biochemistry and Pharmacology of Platelets, pp.175. Elsevier, Amsterdam.
- HOLMSEN, H. (1977). Thromb. Haemos., <u>38</u>, 1030.
- HOLMSEN, H. (1977a). In: "Platelets and Thrombosis". pp.45. Ed. MILLS, D.L.B. and PARETTI. Acadmic Press, London.
- HOLMSEN, H. (1972). Clinics in Haematol. Vol. 1, 235.
- HOLMSEN, H. (1974). In: "Platelets: Production, Function, Transfusion and Storage". p.207. Eds. BALDINI, M.G. and EBBE, S. Grune and Stratton, N. York.
- HOLMSEN, H. and DANGELMAIER, C.A. (1981). J. Biol. Chem., 256, 10449.
- HOLMSEN, H. and DAY, H.J. (1970). J. Lab. Clin. Med., 75, 840.
- HOLMSEN, H. and WEISS, H. (1979). Ann. Rev. Med., 30, 119.
- HOLMSEN, H., DANGELMAIER, C.A. and HOLMSEN HOLM-KJETIL (1981). J. Biol. Chem., 256, 9393.

- HOLMSON, H. (1982). In: "Hemostasis and Thrombosis. Basic Concepts and Clinical Practice". pp.431. Philadelphia, Lippincott.
- HOLMSON, H., DANGELMAIER, C.A. and RONGVED, S. (1984). Biochem. J., <u>222</u>, 157.
- HOTCHKISS, A., JORDAN, J.V., HIRATA, F., SHULMAN, N.R. and AXLEROD, J. (1981). Biochem. Pharmacol., <u>30</u>, 2089.
- HUANG, E.M. and DETWILER, T.C. (1986). In: "Biochemistry of Platelets" pp.1. Eds. PHILIPS, D.R. and SCHMAN, M.
- HOVIG, T. (1963). Throm. Diath. Haemorrh., 9, 248.
- HWANG, D.H. (1980). Prostaglandin Med., 5, 163.
- IEYASU, H., TAKAI, Y., KAIBUCHI, K., SAWAMURA, M. and NISHIZUKA, Y (1982). Biochem. Biophys. Res. Communc., <u>108</u>, 1701.
- IMAI, A., NAKASHIMA, S. and NOZAWA, Y. (1983). Biochem. Biophys. Res. Communc. <u>110</u>, 108.
- IMADKA, T., LYNHAM, J.A. and HASLAM, R.J. (1983). J. Biol. Chem., 258, 11404.
- INAGAKI, M., KAWAMOTO, S. and HIDAKA, H. (1984). J. Biol. Chem., <u>259</u>, 14321.
- INOUE, M., KISHIMOTO, A., TAKAI, Y. and NISHIZUKA, Y. (1977). J. Biol. Chem. <u>252</u>, 7610.
- IRVINE, R.F., DAWSON, R.M.C. and FREINKEL, N. (1983). In: "Contemporary Metabolism" 2. Ed. FRENKEIL, N.
- IRVINE, R.F., LETCHER, A.J. and DAWSON, R.M.C. (1984). Biochem. J., <u>218</u>, 177.
- IRVINE, R.F., ANNGARD, E.E., LETCHER, A.J. and DOWNES, C.P. (1985). Biochem. J. <u>229</u>, 505.
- IRVINE, R.F., LETCHER, A.J., HESLOP, J.P. and BERRIDGE, M.J. (1986). Nature, <u>320</u>, 631.
- IRVINE, R.F., BROWN, K.D. and BERRIDGE, M.J. (1984b). Biochem. J. <u>222</u>, 269.
- IRVINE, R.F., LETCHER, A.J., LANDER, D.J. and DOWNES, C.P. (1984a) Biochem. J. <u>223</u>, 237.

- JACOBSON, M.D., WUSTMAN, M. and DOWNES, C.P. (1973). J. Neurochem., <u>44</u>, 465.
- JACKSON, C.W. (1973). Blood, <u>42</u>, 413.
- JAKOBS, K.H., AKTORIES, K. and SCHULTZ, G. (1984). Adv. Cyclic Nucleotide Res., <u>17</u>, 135.
- JAKOBS, K.H., SAUR, W. and SCHULTZ, G. (1978). FEBS Letts. <u>85</u>, 167.
- JAKOBS, K.H., AKTORIES, K., MINUTH, M. and SCHULTZ, G. (1985). Adv. Cyclic Nucleo. Prot. Phos. Res., <u>19</u>, 137.
- JAKOBS, K.H., MINUTH, M., BAUER, S., GRANDT, R., GREINER, C. and ZUBIN, P. (1986). Basic Res. Cardiol. <u>81</u>, 1.
- JAMES, H.L., GANGULY, P. and JACKSON, C.W. (1977). Thromb. Haemos., <u>38</u>, 939.
- JOHNSON, P.C., WARE, J.A., CLIVEDEN, P.B., SMITH, M., DVORAK, A.M. and SALTZMAN, W. (1985). J. Biol. Chem., <u>260</u>, 2069.
- JOHNSON, S.A. (1971). In: "The Circulating Platelet", pp.284. Ed. JOHNSON, S.A. Academic Press, N. York.
- JOIST, J.H., NIEWIAROWSKI, S., NATH, N and MUSTARD, J.F. (1976). J. Lab. Clin. Med. 87, 659.
- JOLLES, J., ZWIERS, H., DEKKER, A., WIRTZ, K.W.A. and GISPEN, W.H. (1981). Biochem. J., <u>194</u>, 283.
- JONES, R.L., WILSON, N.H. and ARMSTRONG, R.A. (1985). Adv. Exp. Med. Biol., <u>192</u>, 67.
- JOSEPH, S.K. and WILLIAMS, R.J. (1985). FEBS Letts. 180, 150.
- JOSEPH, S.K., THOMAS, A.P., WILLIAMS, R.J., IRVINE, R.F. and WILLIAMSON, J.R. (1984). J. Biol. Chem. <u>259</u>, 3077.
- JY, W. and HAYNES, D.H. (1984). Circ. Res., 55, 595.
- KATBUCHI, K., SANO, K., HOSHIJIMA, M., TAKAI, Y. and NISHIZUKA, Y. (1982). Cell Calcium, <u>3</u>, 323.
- KATBUCHI, K., TAKAI, Y., SAWAMURA, M., HOSHIJIMA, M., FUJIKURA, T. and NISHIZUKA, Y. (1983). J. Biol. Chem., <u>258</u>, 6701.
- KAPLAN, D.R., CHAO, F.C., STILES, C.D., ANTONIADES, H.N. and SCHER, C.D. (1979). Blood, <u>53</u>, 1043.

- KAPLAN, K.L., BROEKMAN, M.J., CHERNOFF, A., LESZNIK, G.P. and DRILLINGS, S. (1979a). Blood, 53, 604.
- KARPATKIN, S. (1978). Blood, 51, 307.
- KASER-GLANZMANN, R., JAKABOVA, M., GEORGE, J.W. and LIECHER, E.F. (1977). Biochim. Biophys. Acta, <u>466</u>, 429.
- KASER-GLANZMANN, R., GERBER, E. and LUSCHER, E.F. (1979). Biochim. Biophys. Acta, 558, 344.
- KASER-GLANZMANN, R., JAKABOVA, M., GEORGE, J.N. and LUSCHER, E.F. (1978). Biochim. Biophys. Acta., <u>512</u>, 1.
- KATADA, T. and UI, M. (1982). Proc. Natl. Acad. Sci. U.S.A., <u>79</u>, 3129.
- KATADA, T., GILMAN, A.G., WATANABE, Y., BAUER, S. and JAKOBS, K.H. (1985). Eur. J. Biochem. <u>151</u>, 431.
- KAUFMAN, R.M., AIRD, R. and CROSBY, W.H. (1965). Blood, 26, 720.
- KAWAHARA, Y., TAKAI, Y., MINAKUUCHI, R., SANO, K. and NISHIZUMA, Y. (1980). Biochem. Biophys. Res. Communc., <u>97</u>, 309.
- KAWAHARA, Y., YAMANISHI, J., TSUNEMITSU, M. and FUKUZAKI, H. (1983). Thromb. Res., <u>30</u>, 477.
- KEIL, B. (1971). Enzymes. 3rd Ed., 3, 249.
- KIKKAWA, U., TAKAI, Y., TANAKA, Y., MEYAKE, R. and NISHIZUKA, Y. (1983). J. Biol. Chem., <u>258</u>, 11442.
- KIRK, C.J., MICHELL, R.H. and HEMS, D.A. (1981a). Biochem. J. <u>194</u>, 155.
- KIRK, C.J., CREBA, J.A., DOWNES, C.P. and MICHELL, R.H. (1981). Biochem. Soc. Trans. <u>9</u>, 377.
- KISHIMOTO, A., TAKAI, Y., MORI, T., KIKKAWA, U. and NISHIZUKA, Y. (1980). J. Biol. Chem. <u>255</u>, 2273.
- KISS, Z. and LUO, Y. (1986). FEBS Letts. 198, 203.
- KITO, M., NARITAL, H., ISHINAGA, M., PARK, H.J. and TAKAMURA, H. (1985). J. Biochem. <u>97</u>, 765.
- KNIGHT, D.E. and SCRUTTON, M.C. (1980). Thromb. Res., 20, 437.
- KNIGHT, D.E., HALLAM, T.J. and SCRUTTON, M.C. (1982). Nature, <u>296</u>, 256.

- KOJIMA, I., LIPPES, H., KOJIMA, K. and RASMUSSEN, H. (1983). Biochem. Biophys. Res. Communc. <u>116</u>, 555.
- KOO, C., LEFKOWITZ, R.J. and SNYDERMAN, R. (1983). J. Clin. Invest., <u>72</u>, 748.
- KOUTOUZOV, S., COTHENET-VERNOUX, L., DAUSSE, J.P. and MARCHE, P. (1985). Biochem. and Biophys. Res. Communc., <u>132</u>, 1051.
- KRAUSE, K.H., SCHLEGEL, W., WOLLHEIM, C.B., ANDERSON, T., WALDVOGEL, F.A. and LEW, P.D. (1985). J. Clin. Invest. <u>76</u>, 1348.
- KRISHNAMURTHI, S., JOSEPH, S.K. and KAKKAR, V.V. (1986). FEBS Letts., <u>196</u>, 365.
- KUO, I.F. and GREENGARD, P. (1969). Proc. Natl. Acad. Sci. U.S.A., <u>64</u>, 1349.
- LAGARDE, M., CROSET, M., GUICHARDANI, M. and DECHAVANNE, M. (1985). Adv. Exp. Med. Biol., <u>192</u>, 327.
- LAGAST, H., POZZAN, T., WALDVOGEL, F.A. and LEW, P.D. (1984). J. Clin. Invest. <u>73</u>, 878.
- LAGES, B. and KRUGER, C. (1983). Thromb. Res., 29, 103.
- LANDS, A.M., ARNOLD, A., MCAULIFF, J.B., LUDUENA, F.P. and BROWN, T.G. (1967). Nature, <u>214</u>, 597.

LANGER, S.Z. (1974). Biochem. Pharm. 23, 1793.

- LANGER, S.Z. and SHEPPERSON, N.B. (1982). Trends Pharm. Sci. 3, 440.
- LAPETINO, E.G. (1982). Trends Pharm. Sci. 3, 115.
- LAPETINA, E.G., BILLAH, M.M. and CAUTRECASAS, P. (1981). J. Biol. Chem. 256, 5037.
- LAPETINA, E.G., SCHMITGES, C.J., CHANDRABOSE, K. and CAUTRECASAS, P. (1977). Biochem. Biophys. Res. Communc. <u>76</u>, 828.
- LAPETINA, E.G., BILLAH, M.M. and CAUTRECASAS, P. (1981a). J. Biol. Chem. <u>256</u>, 11984.
- LAPETINA, E.G., WATSON, S.P. and CUATRECASAS, P. (1984). Proc. Natl. Acad. Sci. U.S.A., <u>81</u>, 7431.
- LEPORE, D.A., HARRIS, R.A. and PENNINGTON, D.B. (1984). Br. J. Haematol., <u>58</u>, 473.

- LEVEN, R.M. and NACHMIAS, V. (1981). In: Megakaryocyte Biology and Precursors: In Vitro Cloning and Cellular Properties, pp.261. Eds. EVATT, LEVINE and WILLIAMS, Elsevier, N. Holland, Amsterdam.
- LEVEN, R.M. and NACHMIAS, V. (1982). J. Cell Biol. 92, 313.
- LEVEN, R.M., MULLIKIN, W.H. and NACHMIAS, V. (1983). J. Cell Biol. <u>96</u>, 1234.
- LEVINE, R.F. (1980). Br. J. Haematol., 45, 487.
- LEVINE, R.F. (1981). In: Megakaryocyte Biology and Precursors: In Vitro Cloning and Cellular Properties, pp.203. Eds. EVATT, LEVINE and WILLIAMS. Elsevier, N. Holland, Amsterdam.
- LEVINE, R.F. (1986). In: "Biochemistry of Platelets" pp.417. Eds. PHILLIPS, D.R. and SCHUMA, M.A. Academic Press.
- LEVINE, R.F., ELDOR, A., HYAM, E., GAMLIEL, H., FUKS, Z. and VLODAVSKY, I. (1985). Blood, <u>66</u>, 570.
- LEVINE, R.F., HAZZARD, K.C. and LAMBERG, J.D. (1982). Blood, <u>60</u>, 1122.
- LIMBIRD, L.E. and LEFKOWITZ, R.J. (1977). J. Biol. Chem. <u>252</u> 799.
- LITOSCH, I., WALLIS, C. and FAIN, J.N. (1985). J. Biol. Chem. <u>260</u>, 5464.
- LLOYD, J.V., NISHIZAWA, E.E. and MUSTARD, J.V. (1973b). Br. J. Haematol., <u>25</u>, 77.
- LLOYD, J.V. and MUSTARD, J.F. (1974). Br. J. Haematol., <u>26</u>, 243.
- LLOYD, J.V., NISHIZAWA, E.E., HALDAR, J. and MUSTARD, J.F. (1972). Br. J. Haematol., <u>23</u>, 571.
- LLOYD, J.V., NISHIZAWA, E.E., JOIST, J.H. and MUSTARD, J.F. (1973). Br. J. Haematol., <u>26</u>, 589.
- LLOYD, J.V., NISHIZAWA, E.E., JOIST, J.H. and MUSTARD, J.F. (1973a). Br. J. Haematol., <u>24</u>, 589.
- LONDOS, C., WOLFF, J. and COOPER, D.M.T. (1983). In: "Regulation and Function of Adenosine E," pp.17. Eds. BERNE, R.M., RALL, T. and RUBIO, R. Martinus Nijhoff, Boston.

- LONDOS, C., COOPER, D.M. and WOLFF, J. (1980). Proc. Natl. Acad. Sci. U.S.A. <u>77</u>, 2551.
- LONG, M.W. and WILLIAMS, N. (1981). In: "Megakaryocyte Biology and Precursors: In Vitro Cloning and Cellular Properties" pp.293. Eds. EVATT, LEVINE and WILLIAMS, Elsevier, N. Holland.

;

- LONG, M.W., WILLIAMS, N. and McDONALD, T.P. (1982). J. Cell Physiol., <u>112</u>, 339.
- LOTE, K. and LOWERY, C. (1979). Thromb. Haemos., <u>41</u>, 407.
- LOWRY, O.H., ROSENBROUGH, N.J., FARR, A.L. and RANDALL, R.J. (1951). J. Biol. Chem., <u>193</u>, 265.
- LUCAS, D.O., BAJJALIEH, S.M., KOWALCHJK, J.A. and MARTIN, T.F.J. (1985). Biochem. Biophys. Res. Communc., <u>132</u>, 721.
- LYNCH, C.J., BLACKMORE, P.F., CHAREST, R. and EXTON, J.H. (1985). Mol. Pharm., <u>28</u>, 93.
- LYONS, R.M., STANFORD, N. and MAJERUS, P.W. (1975). J. Clin. Invest., <u>56</u>, 924.
- MACARA, I.G. (1985). Am. J. Physiol., <u>248</u>, C3-C11.
- MACFARLANE, D.E. and MILLS, D.C.B. (1975). Blood, <u>46</u>, 309.
- MACFARLANE, D.E., MILLS, D.C.B. and SRIVASTAVA, P.C. (1982). Biochemistry, 21, 544.
- MACFARLANE, D.E., SRIVASTAVA, P.C. and MILLS, D.C.B. (1983). J. Clin. Invest., <u>71</u>, 420.
- MACINTYRE, D.E. (1985). In: "Biochemistry of Arachidonic Acid Metabolites". Ed. LANDS, W.E.M., Martinus Nijhoff Pub. Corp., Boston.
- MACINTYRE, D.E. and RINK, T.J. (1982). Thromb. Haemos., 47, 22.
- MACINTYRE, D.E. and SHAW, A.M. (1983). Thromb. Res., 31, 833.
- MACINTYRE, D.E., BUSHFIELD, M. and SHAW, A.M. (1985a). FEBS Letts. <u>188</u>, 383.
- MACINTYRE, D.E., BUSHFIELD, M., MACMILLAN, L.J., MOFFAT, K.J., MURDOCH, F.A., THOMSON, A.G., ROSSI, A.G. and McNICOL, A. (1986). Agents Actions Unpub.
- MACINTYRE, D.E., MACIAG, T., SALZMAN, A.M. and WEINSTEIN, R. (1983a). J. Pharmacol., <u>79</u>, 361P.

- MACINTYRE, D.E., POLLOCK, W.K., SHAW, A., BUSHFIELD, M., MACMILLAN, L.J. and McNICOL, A. (1985c). Adv. Exp. Biol. Med., <u>192</u>, 127.
- MACINTYRE, D.E., SHAW, A.M., BUSHFIELD, M., MACMILLAN, L.J., McNICOL, A. and POLLOCK, W.K. (1985b). Nouv. Rev. fr Hematol., <u>27</u>, 285.
- MACINTYRE, D.E., SHAW, A.M., POLLOCK, W.K., BRYDON, L.J., SALZMAN, E.W., WEINSTEIN, R. and MACIAG, T. (1983b). Thromb. Haemos., <u>50</u>, 281.
- MACMILLAN, L.J.M., MCNICOL, A., POLLOCK, W.K., SHAW, A.M., DRUMMOND, A.H. and MACINTYRE, D.E. (1984). IUPHAR, Proc. 9th International Congress of Pharmacology, 1746P.
- MACPHERSON, G.G. (1972). J. Ultrastruct. Res., 40, 167.

MAGNUSSON, S. (1972). Enzymes. 3rd Ed. 3, 278.

- MAHADEVAPPA, V.G. and HOLUB, B.J. (1982). Biochim. Biophys. Acta <u>713</u>, 73.
- MAHADEVAPPA, V.G. and HOLUB, B.J. (1983). J. Biol. Chem., <u>258</u>, 5337.
- MAHADEVAPPA, V.G. and HOLUB, B.J. (1986). Biochem. Biophys. Res. Communc. <u>134</u>, 1327.
- MALAGODI, M.H. and CHIOU, C.Y. (1974). Eur. J. Pharmac. 27, 25.
- MALAGODI, M.H. and CHIOU, C.Y. (1974a). Pharmacology, <u>12</u>, 20.
- MARCUS, A.J., ULLMAN, H.L. and SAFIER, L.B. (1969). J. Lipid Res., <u>10</u>, 108.
- MARGUERIE, G.A., GINSBERG, M.H. and PLOW, E.F. (1985). Adv. Exp. Med. Biol., <u>192</u>, 41.
- MARKWARDT, F. (1960). Methods in Enzymol. 19, 924.
- MARTIN, B.M., FEINMAN, R.D. and DETWILER, T.C. (1975). Biochem., 14, 1308.

MARTIN, T.F.J. (1983). J. Biol. Chem., 258, 14816.

- MARTIN, J.F., PLUMB, J., KILBEY, R.S. and KISHK, Y.T. (1983a). Br. Med. J., <u>287</u>, 456.
- MARTIN, J.F., SLATER, D.N. and TROWBRIDGE, E.A. (1983b). Lancet, 1, 793.
- MASSINI, P. and LUSCHER, E.F. (1974). Biochim. Biophys. Acta., <u>372</u>, 109.
- MASTRO, A.M. and SMITH, M.C. (1983). J. Cell Physiol., <u>116</u>, 51.
- MAUCO, G., CHAP, H. and DOUSTE-BLAZY, L. (1979). FEBS Letts., 100, 367.
- MAUCO, G., DNAGLEMAIER, R.A. and SMITH, J.B. (1984). Biochem. J., <u>224</u>, 933.
- MAY, W.S., SAHYOUN, N., WOLF, M. and CUATRECASAS, P. (1985). Nature, <u>317</u>, 549.
- McGOWAN, E. and DETWILER, T.C. (1986). J. Biol. Chem., <u>261</u>, 739.
- McGOWAN, E.B. and DETWILER, T.C. (1983). Thromb. Res., 31, 297.
- McGOWAN, E.B., DING, A-H. and DETWILER, T.C. (1983). J. Biol. Chem., <u>258</u>, 11243.
- McGRATH, J.C. (1982). Biochem. Pharm. 31, 467.
- McKEAN, M.L., SMITH, J.B. and SILVER, M.J. (1981). J. Biol. Chem., <u>256</u>, 1522.
- McLAREN, K.M. (1983). J. Clin. Pathol., <u>36</u>, 197.
- McLEAN, J.P. and VELOSO, H. (1967). Life Sci., <u>6</u>, 1983.
- MELLWIG, K.P. and JAKOBS, K.H. (1980). Thromb. Res., <u>18</u>, 7.
- MENASHI, S., AUTHI, S., CAREY, F. and CRAWFORD, N. (1984). Biochem. J., <u>222</u>, 413.
- MENASHI, S., DAVIS, C. and CRAWFORD, N. (1982). FEBS Letts., 140, 298.
- MICHEL, H., CAEN, J.P., BORN, G.V.R., MILLER, R., D'AURIAC, G.A. and MEYER, P. (1976). Br. J. Haematol., <u>33</u>, 27.
- MICHELL, R.H. (1975). Biochim. Biophys. Acta., <u>415</u>, 81.
- MICHELL, R.H. (1982). Nature, 296, 492.
- MICHELL, R.H. (1979). Trends Biochem. Sci., 4, 128.

- 202 -

- MICHELL, R.H. and KIRK, C.P. (1981). Trends Pharmac. Sci., 2, 86.
- MICHELL, R.H. (1986). Nature, 319, 176.
- MICHELL, R.H., JAFFERJI, S.S. and JONES, L.M. (1976). FEBS Letts., <u>69</u>, 1.
- MICHELL, R.H., KIRK, C.J., JONES, L.M., DOWNES, C.P. and CREBA, J.A. (1981). Phil. Trans. R. Soc. Lond. Ser. B., 296, 123.
- MILLER, J.L. (1983). Blood, <u>61</u>, 967.
- MILLER, J.L., STUART, M.J. and WALENGA, R.W. (1982). Biochem. Biophys. Res. Communc., <u>107</u>, 752.
- MILLS, D.C.B. (1974). In: "Platelets and Thrombosis" pp.45. Eds. SHERRY, S. and SCRIABINE, A. Univ. Park Press, Baltimore, Maryland.
- MILLS, D.C.B. (1973). Nature, <u>243</u>, 220.

MILLS, D.C.B., (1970). Symp. Zool. Soc. Lond., <u>27</u>, 99.

- MILLS, D.C.B. and MACFARLANE, D.E. (1974). Thromb. Res., 5, 401.
- MILLS, D.C.B., ROBB, I.A., ROBERTS, G.C.K. (1968). J. Physiol., <u>195</u>, 715.
- MOODY, C.J., MEGGJI, P. and BURNSTOCK, G. (1984). Eur. J. Pharm. <u>97</u>, 47.
- MONCADA, S., RADOMSKI, M. and VARGAS, J.R. (1982). B.J. Pharm., <u>75</u>, 165P.
- MONCADA, S. and VANE, J.R. (1979). Pharm. Rev., 30, 293.
- MONCADA, S. and VANE, J.R. (1979a). N. Engl. J. Med., 300, 1142.
- MONCADA, S. and WHITTLE, B. (1985). Adv. Exp. Med. Biol., <u>192</u>, 337.
- MONCADA, S., GRYGLEWSKI, R., BUNTING, S. and VANE, J.R. (1976). Nature, <u>263</u>, 663.
- MOOLENAAR, W.H., TSEIN, R.Y., VAN DER SAAG, P.T. and DE LAAT, S.W. (1983). Nature, <u>304</u>, 645.

MORGAN, D.A. and BRODSKY, I. (1985). J. Cell Biol., <u>100</u>, 565. MURAYAMA, T. and UI, M. (1985). J. Biol. Chem., <u>260</u>, 7226.

- MURER, E.H. (1969). Science, 166, 623.
- MURER, E.H. (1972). Biochim. Biophys. Acta, <u>261</u>, 435.
- MURER, E.H. (1983). Drugs of the Future, 8, 940.
- MURER, E.H. and SIOJO, E. (1982). Thromb. Haemos., 47; 59.
- MURPHY, E.A., ROWSELL, H.C., DOWNIE, H.C., ROBINSON, B.A. and MUSTARD, J.F. (1962). Canad. Med. Assoc. J., 87, 259.
- MUSTARD, J.F., PACKHAM, M.A., KINLOUGH-RATHBONE, B.L., PERRY, D.W. and REBOECZI, E. (1978). Blood, <u>52</u>, 453.
- NACCHACHE, P.H., MOLSKI, T.F.P., BORGET, P., WHITE, J.R. and SHA'AFI, R.I. (1985). J. Biol. Chem., <u>260</u>, 2125.
- NACHMAN, R., LEVINE, R., and JAFFE, E.A. (1977). J. Clin. Invest., <u>60</u>, 914.
- NACHMIAS, V. (1983). SEM Haematol., 20, 261.
- NAHORSKI, S.F. and BATTY, I. (1986). Trends Pharmacol. Sci., 7, 83.
- NAKAMURA, T. and UI, M. (1985). J. Biol. Chem., 260, 3584.
- NAKASHIMA, S., TOHMATSU, T., HATTORI, H., OKANO, Y. and NOZAWA, Y. (1986). Biochem. Biophys. Res. Communc., <u>135</u>, 1099.
- NAKEFF, A. and DANIELS-McQUEEN, S. (1976). Proc. Soc. Exp. Biol., Med., <u>151</u>, 587.
- NATORI, R. (1954). Jikei Med., 1, 119.
- NESHEIM, M.E., HIBBARD, L.S., TRACY, P.B., BLOOM, H.W., MYRMEL, K.H. and MANN, K.G. (1980). In: "The Regulation of Coagulation" pp.145. Eds. MANN, K.G. and TAYLOR, F.B. Elsevier, N. Holland, N. York.
- NICHOLS, W.L., GERRARD, J.M. and DIDISHEIM, P. (1981). In: "Recent Advances in Blood Coagulation" Vol. 3, pp.1. Ed. POLLER, L.
- NISHIZUKA, Y. (1980). Molec. Biol. Biochem. Biophys., <u>32</u>, 113.
- NISHIZUKA, Y. (1984). Nature, <u>308</u>, 693.
- NORTHUP, J.K., STERNWEIS, P.C., SMIGEL, M.D., SCHLEIFER, L.S., ROSS, E.M. and GILMAN, A.G. (1980). Proc. Natl. Acad. Sci. U.S.A., <u>77</u>, 6516.

- NUGTEREN, D.H. and HAZELHOF, E. (1973). Biochim. Biophys. Acta., <u>326</u>, 448.
- O'ROURKE, F.A., HALENDA, S.P., ZAVOICO, G.B. and FEINSTEIN, M.B. (1985). J. Biol. Chem., <u>260</u>, 956.
- ODELL, T.T. and JACKSON, C.W. (1968). Blood, <u>32</u>, 102.
- OELZ, O., OELZ, R. and KNAPP, H.R. (1977). Prostaglandins, 13,
- ØLLGAARD, E. (1961). Thromb. Diath. Haemorh., 6, 86.
- PACKHAM, M.A. (1983). Thromb. Haemos., 50, 610.
- PACKHAM, M.A. and MUSTARD, J.F. (1986). SEM Haematol., 23, 8.
- **PAULUS**, J.-M. (1975). Blood, <u>46</u>, 321.

225.

- PAULUS, J.-M. (1970). Blood, 35, 298.
- PEARSON, J.D. (1985). In: "Methods in Pharmacology", <u>6</u>, 87. Ed. PATON, D.M. Plenum Pub. Corp.
- PENINGTON, D.G. (1981). In: "Platelets in Biology and Pathology"
 2, pp.19. Ed. GORDON, G.L. Elsevier/N. Holland
 Biomedical Press.
- PENNINGTON, D.G., STREATFIELD, K. and ROXBROUGH, A.E. (1976). Br. J. Haematol., <u>34</u>, 639.
- PFER, D.D., COLMAN, P. and CHESNEY, B.M. (1977). Thromb. Haemos., <u>38</u>, 126.
- PHILIPS, D.R. (1974). Thromb. Diath Haemor., 32, 207.
- PHILIPS, D.R. and POH AGIN, P. (1977). Biochem. Biophys. Res. Communc., <u>75</u>, 940.
- POBINER, B.F., HEWLETT, E.L. and GARRISON, J.C. (1985). J. Biol. Chem., <u>260</u>, 162000.
- POGGIOLI, J., WEISS, S.J., McKINNEY, J.S. and PUTNEY, J.W. (1983). Mol. Pharm., <u>23</u>, 71.
- POLLARD, T.D., FUJIWARA, K., HANDIN, R. and WEISS, G. (1977). Ann. N.Y. Acad. Sci., <u>283</u>, 218.
- POLLOCK, W.K., ARMSTRONG, R.A., BRYDON, L.J., JONES, R.L. and MACINTYRE, D.E. (1984). Biochem. J., <u>219</u>, 833.
- PRENTKI, M., BIDEN, T.J., JANIC, D., IRVINE, R.F., BERRIDGE, M.J. and WOLLHEIM, C.B. (1984). Nature, 309, 562.

- PRENTKI, M., WOLLHEIM, C.B. and LEW, P.D. (1984a). J. Biol. Chem., <u>259</u>, 13777.
- PRESCOTT, S.M. and MAJERUS, P.W. (1983). J. Biol. Chem., <u>258</u>, 764.
- PUTNEY, J.W. (1986). Cell Calcium, 7, 1.
- PUTNEY, J.W., WEISS, S.J., VAN DE WALLE, C.M. and HADDAS, R.A. (1980). Nature, <u>284</u>, 345.
- RABELLINO, E.M., LEVINE, R.B., LEUNG, L.L.K. and NACHMAN, R.L. (1981). J. Exp. Med. <u>154</u>, 88.
- RABELLINO, E.M., NACHMAN, R.L., WILLIAMS, N., WINCHESTER, R.J. and ROSS, G.D. (1979). J. Exp. Med., <u>149</u>, 1273.
- RASSMUSEN, H. (1970). Science, 170, 404.
- RASMUSSEN, H. and BARRET, Q. (1984). Physiol. Rev., <u>64</u>, 938.
- RAZ, A., MINKES, M. and NEEDLEMAN, P. (1977). Biochim. Biophys. Acta., <u>488</u>, 305.
- REBECCHI, M.J. and GERSHENGORN, M.C. (1983). Biochem. J., <u>216</u>, 287.
- REBECCHI, M.J., KOLESNICK, R.W. and GERSHENGORN, M.C. (1983). J. Biol. Chem., <u>258</u>, 227.
- RENDU, F., MARCHE, J., MACLOUF, J., GIRARD, A. and LEVY-TOLEDANO, S. (1983). Biochem. Biophys. Res. Communc., <u>116</u>, 513.
- **REUTER, H.** (1983). Nature, <u>301</u>, 569.
- RINGER, S. (1883). J. Physiol., <u>4</u>, 29.
- RINK, T.J. and HALLAM, T.J. (1984). Trends Biochem. Sci., 9, 215.
- RINK, T.J., SANCHEZ, A. and HALLAM, T.J. (1983). Nature, 305, 317.
- RINK, T.J. and SANCHEZ, A. (1984). Biochem. J., 222, 833.
- RINK, T.J., SMITH, S.W. and TSEIN, R.Y. (1982). FEBS Letts., 148, 21.
- RINK, T.J., SMITH, S.W. and TSEIN, R.Y. (1981). J. Physiol., <u>324</u>, 53.
- RISTOW, R-H., MESSMER, T.O., WALTER, S. and PAUL, O. (1980). J. Cell Physiol., <u>103</u>, 263.

- 286 -

- RITTENHOUSE, S.E. (1983). Proc. Natl. Acad. Sci. U.S.A., <u>80</u>, 5417.
- RITTENHOUSE, S.E. and SASSON, J.P. (1985). J. Biol. Chem., <u>260</u>, 8657.
- RITTENHOUSE-SIMMONS, S.E. (1979). J. Clin. Invest., 63, 580.
- ROBINSON, G.A., BUTCHER, R.W. and SUTHERLAND, E.W. (1971). In: "Cyclic AMP" Eds. ROBINSON, G.A., BUTCHER, R.W. and SUTHERLAND, E.W. Acad. Press., N. York.
- ROSS, A., GAMMET, J., KARIYA, B. and HARKER, L.A. (1974). Proc. Natl. Acad. Sci. U.S.A., <u>71</u>, 1207.
- ROSS, E.M. and GILMAN, A.G. (1980). Annun. Rev. Biochem., <u>49</u>, 533.
- ROSS, R., FAGGIOTT, A., BOWEN-POPE, D. and RAINES, E. (1984). Circulation, <u>70</u>, Suppl. 111, 77.
- ROTH, G.J. and MAJERUS, P.W. (1975). J. Clin. Invest., <u>56</u>, 624.
- RUBIN, R.P., GODFREY, P.P., CHAPMAN, D.A. and PUTNEY, J.W. (1984). Biochem. J., <u>219</u>, 655.
- RUBIN, R.P. (1982). In: "Calcium and Cellular Secretion". Ed. RUBIN, R.P. Plenum, N. York.
- RUDA, E., SCRUTTON, M.C. and TUFFIN, D. (1985). Thromb. Haemos., <u>54</u>, 248.
- RUGGIERO, M. and LAPETINA, E.G. (1985). Biochem. Biophys. Res. Communc., 131, 1198.
- RUPPERT, D. and WEITHMANN (1982). Life Sci., 31, 2037.
- RYO, R., NAKEFF, A., HUANG, S.S., GINSBERG, M. and DEUEL, T.F. (1983). J. Cell Biol., <u>96</u>, 515.
- SAGE, S.O. and RINK, T.J. (1985). FEBS. Letts., 188, 135.
- SAGI-EISENBERG, S., LIEMAN, H. and PECHT, I. (1985). Nature, <u>313</u>, 59.
- SALZMAN, E.W., JOHNSON, P. and WARE, J.A. (1985). Nouv. Rev. Fr Hematol., <u>27</u>, 205.
- SANCHEZ, A., HALLAM, T.J. and RINK, T.J. (1983). FEBS LETTS., <u>164</u>, 43.
- SANDOW, A. (1952). Yale J. Biol. Med., 24, 176.
- SANO, K., TAKAI, Y., YAMANISHI, J. and NISHIZUKA, Y. (1983). J. Biol. Chem., <u>258</u>, 2010.

- SANTIAGO-CALVO, E., MULE, S.J. and HOKIN, L.E. (1963). Biochim. Biophys. Acta, <u>70</u>, 91.
- SAWYER, S.T. and COHEN, S.C. (1981). Biochemistry, 20, 6280.
- SCHACHT, J. (1981). Methods Enzymol., <u>72</u>, 623.
- SCHICK, B.P. and SCHICK, P.K. (1986). SEM Haematol., 23, 68.
- SCHICK, P.K. and WEINSTEIN, M. (1981). J. Lab. Clin. Med., <u>98</u>, 607.
- SCHLEGEL, W., WUARIN, F., ZBAVEN, C., WOLLHEIM, C.B. and ZAHND, G.P. (1985). FEBS. LETTS., <u>189</u>, 27.
- SCHRAMM, M. and ORLY, J. (1976). Proc. Natl. Acad. Sci. U.S.A., 73, 4410.
- SCHRAMM, M. and SELINGER, Z. (1984). Science, <u>225</u>, 1350.
- SEAMON, K.B. and DALY, J.W. (1981). J. Cyclic Nucleot. Res. 7, 201.
- SEYFRED, M.A., FARRELL, L.E. and WELLS, W.W. (1984). J. Biol. Chem., <u>259</u>, 13203.
- SHARKEY, N.A., LEACH, K.L. and BUMBERG, P.M. (19). Proc. Natl. Acad. Sci. U.S.A., <u>81</u>, 607.
- SHATTIL, S., McDONOUGH, M. and BURCH, J.W. (1981). Blood, 57, 537.
- SHATTIL, S., MONTGOMERY, J.A. and CHIANG, P.K. (1982). Blood, <u>59</u>, 906.
- SHAW, J.O. (1981). Prostaglandins, <u>21</u>, 571.
- SHUMAN, N.A., BOTNEY, M. and FENTON, J.W. (1978). J. Clin. Invest., <u>63</u>, 1211.
- SIEGL, A.M., DALY, J.W. and SMITH, J.B. (1982). Mol. Pharmacol., 21, 680.
- SIESS, W. and LAPETINA, E.G. (1983). Biochim. Biophys. Acta., <u>752</u>, 329.
- SIESS, W. (1985). FEBS. LETTS., <u>185</u>, 151.
- SILVER, M.J., SMITH, J.B. and INGERMAN, G.M. (1974). Agents Actions, <u>4</u>, 233.

- SILVERSTEIN, R.L., LEUNG, L.L.K. and NACHMAN, R. (1986). Arteriosclerosis, <u>6</u>, 245.
- SIMON, M-F., WUSTEMAN, M. and DOWNES, P.C. (1985). J. Neurochem., <u>44</u>, 465.
- SIMON, M-F., CHAP, H. and DOUSTE-BLAZY, L. (1984). FEBS. LETTS., <u>170</u>, 43.
- SIMPSON, A.W.M., HALLAM, T.J. and RINK, T.J. (1984). FEBS LETTS., <u>176</u>, 139.
- SINAKOS, Z. and CAEN, J.P. (1967). Thromb. Diath. Haemor. 17, 99.
- SKAER, R.J. (1975). In: "Biochemistry and Pharmacology of Platelets pp.239. Elsevier/N. Holland.
- SMIGEL, M., KATADA, T., NORTHUP, J.K., BOKOCH, G.M., UI, M. and GILMAN, A.G. (1984). Adv. Cyclic Nucleo. Prot. Phos. Res., <u>17</u>, 1.
- SMITH, J.B., INGERMAN, C. and SILVER, M.J. (1976). J. Clin. Invest., <u>58</u>, 1119.
- SMITH, J.B., INGERMAN, C.M., KOCSIS, J.J. and SILVER, M.J. (1973). J. Clin. Invest., <u>52</u>, 965.
- SPAT, A., BRADFORD, P.G., MCKINNEY, J., RUBIN, R.P. and PUTNEY, J.W. (1986). Nature, <u>319</u>, 514.
- SPECTOR, W.G. and WILLOUGHBY, D.A. (1968). In: "The Pharmacology of Inflammation". Ed. SPECTOR, W.G. and WILLOUGHBY, D.A. The England University Press Ltd., London.
- STEINER, M. (1975). Arch. Biochem. Biop., <u>171</u>, 245.
- STOREY, M.A., SHEARS, S.B., KIRK, C.P. and MICHELL, R.H. (1984). Nature, <u>312</u>, 374.
- STREB, H., IRVINE, R.F., BERRIDGE, M.J. and SCHULZ, I. (1983). Nature, 306, 67.
- SUTHERLAND, E.W. and RALL, T.W. (1960). Pharm. Rev., 12, 265.
- SUTHERLAND, E.W. and ROBISON, G.A. (1966). Pharm. Rev., 18, 145.
- SUTHERLAND, E.W., ROBISON, G.A. and BUTCHER, R.W. (1968). Circul., <u>37</u>, 279.
- TAKAI, Y., KISHIMOTO, A., IWASA, Y., KAWAHARA, Y., MORI, T. and NISHIZUKA, Y. (1979). J. Biol. Chem., <u>254</u>, 3692.

- TAKAI, Y., KISHIMOTO, A. and NISHIZUKA, Y. (1982). In: "Calcium and Cellular Function". Vol. 2. pp.386. Ed. CHEUNG, W.Y. Acad. Press, N. York.
- TAKAI, Y., KISHIMOTO, A., KIKKAWA, U., MORI, T. and NISHIZUKA, Y. (1979a). Biochem. Biophys. Res. Communc., <u>91</u>, 1218.
- TALLARIDA, R.J. and JACOBSON, L.S. (1979). In: "The Dose-Response Relationship in Pharmacology". Springer, N. York.
- TAM, S.W. and DETWILER, T.C. (1978). Biochim. Biophys. Acta., 543, 194.
- TATESON, J.E., MONCADA, S. and VANE, J.R. (1977). Prostaglandins, <u>13</u>, 389.
- TAYLOR, C.W. and MERRIT, J.E. (1986). Trends Pharm. Sci., 7, 238.
- TAYLOR, M.V., METCALFE, J.C., HESKETH, T.R., SMITH, G.A. and MOORE, J.P. (1984). Nature, <u>312</u>, 462.
- THOMAS, A.P., ALEXANDER, J. and WILLIAMSON, J.R. (1984). Biol. Chem., <u>259</u>, 5574.
- THOMPSON, N.T. and SCRUTTON, M.C. (1985). Eur. J. Biochem., <u>147</u>, 421.TIFFERT, T., GARCIO-SANCHO, J. and LEW, V.L
- TIFFERT, T., GARCIO-SANCHO, J. and LEW, V.L. (1984). Biochim. Biophys. Acta., <u>773</u>, 145.
- TOLLENSEN, M., FEAGLER, J.R. and MAJERUS, P.W. (1974). Biol. Chem., <u>249</u>, 2646.
- TOUQUI, L., ROTHHUT, B., SHAW, A.M., FRADIN, A., VARGAFTIG, B.B. and RUSSO-MARIE, F. (1986). Nature, <u>321</u>, 177.
- TRANZER, J.B., DA PRADA, M., PLETSCHER, A. (1972). J. Cell Biol., <u>52</u>, 191.
- TROST, T. and SCHWABE, U.S. (1981). Proc. Scand. Brit. Pharmac. Soc. p.8.
- TROST, T. and SCHWABE, U.S. (1981). Mol. Pharmacol., 19, 228.
- TROWBRIDGE, E.A., MARTIN, J.F. and SLATER, D.N. (1982). Thromb. Res., <u>28</u>, 461.
- TSEIN, R.Y., POZZAN, T. and RINK, T.J. (1982a). J. Cell Biol., 94, 325.

- TSEIN, R.Y. (1981). Nature, 290, 527.
- TSEIN, R.Y., POZZAN, T. and RINK, T.J. (1982). Nature 295, 68.
- TSEIN, R.Y., RINK, T.J. and POENIE, M. (1985). Cell Calcium, <u>6</u>, 145.
- TSEIN, R.Y., POZZAN, T. and RINK, T.J. (1984). Trends Biochem. Sci., <u>9</u>, 263.
- TSIEN, R.Y. (1980). Biochemistry, <u>19</u>, 2396.
- TURNER, S.R., TRAINER, J.A. and LYNN, W.S. (1975). Nature, <u>257</u>, 680.
- UI, M. (1984). Trends Pharmac. Sci., 5, 277.
- VAN CALKER, D., MILLER, M., and HAMPRECHT, B. (1979). J. Neurochem., <u>33</u>, 999.
- VAN HAYENTINGEN, S. (1977). Biol. Rev., 52, 509.
- VAN ROOIJEN, I.A.A., ROSSOWSKA, M. and BAZAN, N.G. (1985). Biochem. Biophys. Res. Communc., 126, 150.
- VARSANYI, M., TOLLE- K-G, HEILMAYER, L.M.G., DAWSON, R.M.C. and IRVINE, R.F. (1983). EMBO J. 2, 1543.
- VERGHESE, M.W., SMITH, C.D. and SNYDERMAN, R. (1985). Biochem. Biophys. Res. Communc., <u>127</u>, 450.
- VICENTINI, L.M., DIRIRGILIO, F., AMBROSINI, A., POZZAN, T. and MELDOLESI, J. (1985). Biochem. Biophys. Res. Communc., <u>127</u>, 310.
- VICKERS, J.D., KINLOUGH-RATHBONE, R.L. and MUSTARD, J.F. (1982). Blood, <u>60</u>, 1247.
- VICKERS, J.D., KINLOUGH-RATHBONE, R.L. and MUSTARD, J.F. (1984). Biochem. J., <u>224</u>, 399.
- VOLPE, P., DIVIRGILIO, F., POZZAN, T. and SALVIATI, G. (1986). FEBS LETTS., <u>197</u>, 1.
- VOLPE, P., SALVIATI, G., DIVIRGILIO, F. and POZZAN, T. (1985). Nature, <u>316</u>, 347.
- WACKER, W.F.C., ULMER, D.D. and VALLEE, B.L. (1956). N. Engl. J. Med., <u>255</u>, 499.
- WALENGA, R.W., MILLER, J.L. and STUART, M.J. (1984). Biochem. Biophys. Res. Communc., <u>119</u>, 982.

WALLACE, M.A. and FAIN, J.N. (1985). J. Biol. Chem., <u>260</u>, 9527. WALSH, P.N. (1974). Blood, <u>43</u>, 597.

- WALSH, P.N., BIGGS, R. and BAGNATELLI, I.G. (1974). Br. J. Haematol., <u>26</u>, 405.
- WARE, J.A., JOHNSON, P.C., SMITH, M. and SALZMAN, E.W. (1985). Biochem. Biophys. Res. Commuc. <u>133</u>, 98.
- WATANABE, Y., HORN, F., BAUER, S. and JAKOBS, K.H. (1985). FEBS LETTS., <u>192</u>, 23.
- WATSON, S.P., McCONNELL, R.T. and LAPETINO, E.G. (1984). J. Biol. Chem., <u>259</u>, 13199.
- WATSON, S.P., McCONNELL, R.T. and LAPETINA, E.G. (1985). Biochem. J., <u>232</u>, 61.
- WATSON, S.P., REEP, B., McCONNELL, R.T. and LAPETINA, E.G. (1985a). Biochem. J., <u>226</u>, 831.
- WATSON, S.P., RUGGIERO, M., ABRAHAMS, S.L. and LAPETINA, E.G. (1986). J. Biol. Chem., <u>261</u>, 5368.
- WATTS, R.P. and BOROVAY, M.A. (1983). Fed. Proc., <u>42</u>, 724.
- WEINSTEIN, R., STEMERMAN, M.B., MACINTYRE, D.E., STEINBERG, H.W. and MACIAG, T. (1981). Blood, <u>58</u>, 110.
- WEISS, B., BAFNZIGER, N.L. and ATKINSON, J.P. (1978). Blood, <u>52</u>, 524.
- WEISS, B. and WALLACE, T.L. (1980): In: "Calcium and Cell Function" Vol. 1, Calmodulin. pp.330. Ed. CHEUNG, W.Y. Acad. Press, N. York.
- WEISS, H.J., BAUMGARTNER, H.R., TSCHOPP, T.B., TURITTO, V.I. and CHOHEN, D. (1978a). Blood, <u>51</u>, 267.

WEKSLER, B.B. (1984). Circul., 70, Suppl. III, pp63.

WHITTAKER, M. and IRVINE, R.F. (1984). Nature, <u>312</u>, 636.

WHITE, J.G. (1972). Am. J. Pathol., <u>66</u>, 295.

- WHITE, J.G. (1974). In: "Progress in Haemos. and Thromb." Vol. 2, pp49. Ed. SPAET, T.H. Grune and Stratton, N. York.
- WHITE, J.G. and GERRARD, J.M. (1979). In: "Methods and Achievement: in Experimental Pathology" 9, 1. Ed. GABBIAN, G. S. KARGER, BASEL.

- WHITE, J.G., RAO, G.H.R. and GERRARD, J.M. (1974). Am. J. Pathol., <u>77</u>, 135.
- WHITTLE, B.J.R., HAMID, S., LIDBURY, P. and ROSAM, A.C. (1985). Adv. Exp. Med. Biol., <u>192</u>, 109.
- WHITTLE, B.J.R., MONCADA, S. and VANE, J.R. (1978). Prostaglandins, <u>16</u>, 373.
- WILLIAMS, N. and LEVINE, R.F. (1982). Br. J. Haematol., <u>52</u>, 173.
- WILLIAMS, K., VANDERWEL, M. and HASLAM, R.J. (1983). Adv. Cyclic Nucleot. Prot. Phos. Res., <u>17</u>A, 28.
- WILLIS, A.L., DAVISON, P., RAMWELL, P.W., BROCKLEHURST, N.E. and SMITH, J.B. (1972). In: "Prostaglandins in Cellular Function". pp.227. Eds. RAMWELL, P.N. and PHARISS, B.B. Plenum Press, N. York.
- WILSON, D.B., BROSS, T.E., HOFMAN, S.L. and MAJERUS, P.W. (1984). J. Biol. Chem., 259, 11718.
- WILSON, D.B., CONNOLLY, T.M., BROSS, T.E., MAJERUS, P.W., SHERMAN, W.R., TYLER, A.W., RUBIN, L.J. and BROWN, J.E. (1985a). Biol. Chem. <u>260</u>, 13496.
- WILSON, D.B., NEUFELD, E.J. and MAJERUS, P.W. (1985). J. Biol. Chem., <u>260</u>, 1046.
- WOLF, M., LEVINE, H., MAY, S., CAUTRECASAS, P. and SAHYOUN, N. (1985). Nature, <u>317</u>, 549.
- WONG, P. Y-K. and CHEUNG, W.Y. (1979). Biochem. Biophys. Res. Communc., <u>90</u>, 473.
- WORTHINGTON, R.E. and NAKEFF, A. (1982). Prostaglandins, 23, 841.
- WRIGHT, J.H. (1910). J. Morph. 21, 263.
- YAMAMOTO, H. and VAN BREEMEN, C. (1985). Biochem. Biophys. Res. Communc., <u>130</u>, 270.
- YAMANISHI, J., TAKAI, Y., KAIBUSCHI, K., SANO, K., CASTAGNA, M. and NISHIZUKA, Y. (1983). Biochem. Biophys. Res. Communc., <u>112</u>, 778.

- YAVIN, E. and ZUTRA, A. (1977). Anal. Biochem., <u>80</u>, 430.
- ZAJICEK, J. (1954). Acta Haematol., <u>12</u>, 238.
- ZAVOICO, G.B. and FEINSTEIN, M.B. (1984). Biochem. Biophys. Res. Communc., <u>120</u>, 579.
- ZAVOICO, G.B., HALENDA, S.P., SH'AFI, R.I. and FEINSTEIN, M.B. (1985). Proc. Natl. Acad. Sci. U.S.A., <u>82</u>, 3859.
- ZUCKER, M.B. and GRANT, R.A. (1978). Blood, <u>52</u>, 505.
- ZUCKER, M.B. and BORELLI, J. (1962). Proc. Soc. Exp. Biol. Med., <u>109</u>, 779.
- ZUCKER-FRANKLIN, D. and PETURSSON, S. (1984). J. Cell Biol., <u>99</u>, 390.

BIBLIOGRAPHY ADDENDA

BYE, A., LEWIS, Y. and O'GRADY, J. (1979). Br. J. Clin. Pharmac.,

7, 283.

Bulus, J-M (1975) Blood, 46, 321.

GLASGOW UNIVERSI LIBR/\RY